# Evaluating the Performance of A Sponge-based Moving Bed Bioreactor on Micropollutants Removal

by

Qi Jiang

A thesis submitted in partial fulfillment of the requirements for the degree of Master of Engineering

University of Technology, Sydney Faculty of Engineering and Information Technology

May, 2016

#### **CERTIFICATE OF ORIGINAL AUTHORSHIP**

I certify that the work in this thesis has not previously been submitted for a degree nor has it been submitted as part of requirements for a degree except as fully acknowledged within the text.

I also certify that the thesis has been written by me. Any help that I have received in my research work and the preparation of the thesis itself has been acknowledged. In addition, I certify that all information sources and literature used are indicated in the thesis.

Name: Qi Jiang

Date: 23/05/2016

### Acknowledgements

This thesis would not have been possible without the help and support of the outstanding individuals I met in the course of my research study. Now it is my turn to acknowledge them all.

First of all, I would like to thank my supervisors Dr. Wenshan Guo and Prof. Huu Hao Ngo. The doors to my supervisors' office were always open whenever I ran into trouble or had questions regarding my research or writing. Their guidance helped me in all the time of research and writing of this thesis. I could not have imagined having better advisors and mentors for my research study.

I would like to extend my particular gratitude to Prof. Long Nghiem from the University of Wollongong for his help and valued feedbacks about my experiments and academic writing. Dr. Jinguo Kang is indebted for his help in the GC-MS analysis of micropollutants. Kaushalya Wijekoon and Luong Nguyen are thanked for their assistance with the freeze drying of sludge samples.

My sincere thanks also goes to Faculty of Engineering and Information Technology for providing a fabulous academic atmosphere for doing research. I am particularly thankful to lab manager Mohammed Johir for his kind and generous help throughout my experiments. My appreciation also goes to my other dear colleagues, Yunlong Luo, Lijuan Deng, Thi Phuong Tram Vo, Thi An Hang Nguyen, Van-son Tran and Cheng Chen, for their generous help in various aspects of this research.

Last but not least, I would also like to give my deepest thanks to my parents and my partner for their faith their encouragement in all of my pursuits, as well as for supporting me throughout my life.

iv

## **Table of Contents**

ittle Page		
Certificate of Original Authorship		
Ackowledgement		
Table of Contents		
List of Figures		
List of Tables	xi	
List of Abbreviations	xvi	
List of Symbols	xvii	
Abstract	xix	
Chapter 1 Introduction	1-1	
1.1 Background	1-2	
1.2 Impact of micropollutants and reduction strategies	1-3	
1.3 Objectives of the research	1-6	
1.4 Outline of the thesis	1-6	
Chapter 2 Literature Review	2-1	
2.1 Introduction	2-2	
2.2 Micropollutants contamination	2-4	
2.2.1 Adverse effects of micropollutants	2-4	
2.2.2 Occurrence of micropollutants in aquatic environment	2-7	
2.2.3 Removal of micropollutants in wastewater treatment	2-12	
2.3 Background of the attached growth systems	2-19	
2.3.1 Attached growth systems	2-19	
2.3.2 Media used in attached growth systems	2-26	
2.4 Moving bed biofilm reactors (MBBRs)	2-31	

	2.4.1 Media material	2-32
	2.4.2 Modia geometry and size	2-32
	2.4.2 Modia filling ratio	2-34
	2.4.5 Media mining fatto	2-35
	2.4.5 Miner III to the MDDD	2-36
	2.4.5 Micropollutant removal in MBBR process	2-36
2.5	Attached growth based membrane hybrid system	2-37
	2.5.1 Membrane fouling	2-38
	2.5.2 Enhanced hybrid membrane system with attached growth media	2-39
	2.5.3 Micropollutant removal in attached growth based hybrid membrane	; 2 42
2.6	System	2-42
2.6	Conclusion	2-43
Chap	ter 3 Experimental Investigation	3-1
3.1	Introduction	3-2
3.2	Materials	2.2
		3-2
	3.2.1 Synthetic wastewater	3-2 3-2
	<ul><li>3.2.1 Synthetic wastewater</li><li>3.2.2 Selected micropollutants</li></ul>	3-2 3-2 3-2
	<ul><li>3.2.1 Synthetic wastewater</li><li>3.2.2 Selected micropollutants</li><li>3.2.3 Sponge (polyurethane foam)</li></ul>	3-2 3-2 3-2 3-3
3.3	<ul> <li>3.2.1 Synthetic wastewater</li> <li>3.2.2 Selected micropollutants</li> <li>3.2.3 Sponge (polyurethane foam)</li> <li>Experimental set-up and operation protocol</li> </ul>	3-2 3-2 3-2 3-3 3-3 3-9
3.3	<ul> <li>3.2.1 Synthetic wastewater</li> <li>3.2.2 Selected micropollutants</li> <li>3.2.3 Sponge (polyurethane foam)</li> <li>Experimental set-up and operation protocol</li> <li>3.3.1 MBBR system</li> </ul>	3-2 3-2 3-2 3-3 3-9 3-9
3.3	<ul> <li>3.2.1 Synthetic wastewater</li> <li>3.2.2 Selected micropollutants</li> <li>3.2.3 Sponge (polyurethane foam)</li> <li>Experimental set-up and operation protocol</li> <li>3.3.1 MBBR system</li> <li>3.3.2 MBBR-MBR system</li> </ul>	3-2 3-2 3-3 3-3 3-9 3-9 3-10
3.3	<ul> <li>3.2.1 Synthetic wastewater</li> <li>3.2.2 Selected micropollutants</li> <li>3.2.3 Sponge (polyurethane foam)</li> <li>Experimental set-up and operation protocol</li> <li>3.3.1 MBBR system</li> <li>3.3.2 MBBR-MBR system</li> <li>3.3.3 SSMBR system</li> </ul>	3-2 3-2 3-3 3-3 3-9 3-9 3-10 3-11
3.3	<ul> <li>3.2.1 Synthetic wastewater</li> <li>3.2.2 Selected micropollutants</li> <li>3.2.3 Sponge (polyurethane foam)</li> <li>Experimental set-up and operation protocol</li> <li>3.3.1 MBBR system</li> <li>3.3.2 MBBR-MBR system</li> <li>3.3.3 SSMBR system</li> <li>3.3.4 Operation protocol</li> </ul>	3-2 3-2 3-2 3-3 3-9 3-9 3-10 3-11 3-12
3.3	<ul> <li>3.2.1 Synthetic wastewater</li> <li>3.2.2 Selected micropollutants</li> <li>3.2.3 Sponge (polyurethane foam)</li> <li>Experimental set-up and operation protocol</li> <li>3.3.1 MBBR system</li> <li>3.3.2 MBBR-MBR system</li> <li>3.3.3 SSMBR system</li> <li>3.3.4 Operation protocol</li> <li>Analytical methods</li> </ul>	3-2 3-2 3-2 3-3 3-9 3-9 3-10 3-11 3-12 3-15
3.3 3.4	<ul> <li>3.2.1 Synthetic wastewater</li> <li>3.2.2 Selected micropollutants</li> <li>3.2.3 Sponge (polyurethane foam)</li> <li>Experimental set-up and operation protocol</li> <li>3.3.1 MBBR system</li> <li>3.3.2 MBBR-MBR system</li> <li>3.3.3 SSMBR system</li> <li>3.3.4 Operation protocol</li> <li>Analytical methods</li> <li>3.4.1 Organic matter, nutrients, pH and DO</li> </ul>	3-2 3-2 3-2 3-3 3-9 3-9 3-10 3-11 3-12 3-15 3-15
3.3 3.4	<ul> <li>3.2.1 Synthetic wastewater</li> <li>3.2.2 Selected micropollutants</li> <li>3.2.3 Sponge (polyurethane foam)</li> <li>Experimental set-up and operation protocol</li> <li>3.3.1 MBBR system</li> <li>3.3.2 MBBR-MBR system</li> <li>3.3.3 SSMBR system</li> <li>3.3.4 Operation protocol</li> <li>Analytical methods</li> <li>3.4.1 Organic matter, nutrients, pH and DO</li> <li>3.4.2 MLSS and MLVSS</li> </ul>	3-2 3-2 3-2 3-3 3-9 3-9 3-10 3-11 3-12 3-15 3-15 3-16
3.3 3.4	<ul> <li>3.2.1 Synthetic wastewater</li> <li>3.2.2 Selected micropollutants</li> <li>3.2.3 Sponge (polyurethane foam)</li> <li>Experimental set-up and operation protocol</li> <li>3.3.1 MBBR system</li> <li>3.3.2 MBBR-MBR system</li> <li>3.3.3 SSMBR system</li> <li>3.3.4 Operation protocol</li> <li>Analytical methods</li> <li>3.4.1 Organic matter, nutrients, pH and DO</li> <li>3.4.2 MLSS and MLVSS</li> <li>3.4.3 Micropollutant</li> </ul>	3-2 3-2 3-2 3-3 3-9 3-9 3-10 3-11 3-12 3-15 3-15 3-16 3-17

## Chapter 4 Micropollutant Removal in an Moving Bed Biofilm Reactor (MBBR) at Different HRTs

4-1

4.1	Introduction	4-2
4.2	Organic and nutrient removal	4-2
4.3	Removals of selected micropollutants	4-16
4.4	Comparison between the MBBR and other techniques for micropollutant removal	4-28
4.5	Conclusions	4-30

# Chapter 5 Removal of Micropollutants by an Moving Bed BiofilmReactor-Membrane Bioreacter (MBBR-MBR) Hybrid System atDifferent HRTs5-1

5.1	Introduction	5-2
5.2	Organic matter and nutrient removal	5-5
5.3	Removals of selected micropollutants	5-19
5.4	Membrane fouling analysis	5-32
5.5	Conclusions	5-37

## Chapter 6 Evaluation of Micropollutant removal and fouling reduction in a Moving Bed Biofilm Reactor-Membrane Bioreactor Hybrid System 6-1

6.1 Introduction	6-2
6.2 Organic and nutrient removal	6-3
6.3 Removals of selected micropollutants	6-8
6.4 Membrane fouling analysis	6-10
6.5 Conclusions	6-15
Chapter 7 Conclusions and Recommendations	7-1

7.1	Conclusions	7-2
7.2	Recommendations for future research	7-4

References	<b>R-1</b>
Publications Related to This Research	<b>P-1</b>

## **List of Figures**

- Figure 1.1 Schematic presentation of diffuse and point source entry paths of micropollutants in the environment
- Figure 1.2 The outline of the thesis
- Figure 2.1 Contamination pathways for micropollutants into the aquatic environment (Stuart et al., 2012)
- Figure 2.2 Main removal mechanisms of micropollutants in conventional WWTPs
- Figure 3.1 The attached-growth carriers (sponge cubes) used in this study
- Figure 3.2 On-site photo and schematic diagram of the MBBR
- Figure 3.3 On-site photo and schematic diagram of the MBBR-MBR system
- Figure 3.4 On-site photo of the SSMBR system
- Figure 3.5 Schematic diagram of SPE process for GC-MS analysis of micropollutants
- Figure 3.6 Schematic diagram of solvent extraction process of sludge samples
- Figure 3.7 The analytical instruments used in this study
- Figure 3.8 Procedure used for SPE of analysing micropollutants
- Figure 3.9 Schematic diagram of processes for SMP and EPS extraction
- Figure 3.10 Relative hydrophobicity test (A) and microscopic analysis (B)
- Figure 4.1 TOC removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h)
- Figure 4.2 COD removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h)
- Figure 4.3 NH<sub>4</sub>-N removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h)
- Figure 4.4 TN removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h)
- Figure 4.5 PO<sub>4</sub>-P removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h)
- Figure 4.6 Summarized removal efficiencies in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h)
- Figure 4.7 TOC removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h)

- Figure 4.8 COD removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h)
- Figure 4.9 NH<sub>4</sub>-N removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h)
- Figure 4.10 TN removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h)
- Figure 4.11 PO<sub>4</sub>-P removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h)
- Figure 4.12 Summarized removal efficiencies in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h)
- Figure 4.13 TOC removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h)
- Figure 4.14 COD removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h)
- Figure 4.15 NH<sub>4</sub>-N removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h)
- Figure 4.16 TN removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h)
- Figure 4.17 PO<sub>4</sub>-P removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h)
- Figure 4.18 Summarized removal efficiencies in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h)
- Figure 4.19 TOC removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h)
- Figure 4.20 COD removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h)
- Figure 4.21 NH<sub>4</sub>-N removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h)
- Figure 4.22 TN removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h)
- Figure 4.23 PO<sub>4</sub>-P removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h)

- Figure 4.24 Summarized removal efficiencies in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h)
- Figure 4.25 Micropollutant removals in the MBBR unit at HRT of 24 h, 18 h, 12 h, and 6 h. The error bar of each sample represents the standard deviation over the experiment period (MBBR conditions: aeration rate = 4.5L/min, DO = 5.5-6.5 mg/L; HRT = 24h, 18h, 12h and 6 h)
- Figure 4.26 Relationship of Log D and pharmaceutical compounds removals in the MBBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 24h, 18h, 12h and 6h)
- Figure 4.27 Relationship of Log D and pesticides removals in the MBBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h, 18h, 12h and 6h)
- Figure 4.28 Relationship of Log D and industrial chemicals removals in the MBBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h, 18h, 12h and 6h)
- Figure 4.29 Relationship of Log D and estrogenic hormones removals in the MBBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h, 18h, 12h and 6h)
- Figure 4.30 Fate of the studied micropollutants in the MBBR system (aeration rate: 4L/min; HRT: 24 h)
- Figure 4.31 Fate of the studied micropollutants in the MBBR system (aeration rate: 4L/min; HRT: 18 h)
- Figure 4.32 Fate of the studied micropollutants in the MBBR system (aeration rate: 4L/min; HRT: 12 h)
- Figure 4.33 Fate of the studied micropollutants in the MBBR system (aeration rate: 4L/min; HRT: 6 h)
- Figure 5.1 TOC removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2$ ·h; HRT =

6h; SRT =  $\infty$ )

- Figure 5.2 COD removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.3 NH<sub>4</sub>-N removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.4 TN removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.5 PO<sub>4</sub>-P removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.6 Summarized removal efficiencies in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO = 5.5-6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.7 TOC removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.8 COD removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2$ ·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.9 NH<sub>4</sub>-N removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )

- Figure 5.10 TN removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.11 PO<sub>4</sub>-P removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.12 Summarized removal efficiencies in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO = 5.5-6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.13 TOC removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.14 COD removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.15 NH<sub>4</sub>-N removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.16 TN removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.17 PO<sub>4</sub>-P removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.18 Summarized removal efficiencies in the MBBR-MBR (MBBR

conditions: aeration rate = 4 L/min; DO = 5.5-6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )

- Figure 5.19 TOC removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.20 COD removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2 \cdot \text{h}$ ; HRT = 6 h; SRT =  $\infty$ )
- Figure 5.21 NH<sub>4</sub>-N removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.22 TN removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.23 PO<sub>4</sub>-P removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5-6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.24 Summarized removal efficiencies in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO = 5.5-6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.25 Micropollutant removals in the MBBR-MBR while MBBR unit operating at HRT of 24 h, 18 h, 12 h, and 6 h. The error bar of each sample represents the standard deviation over the experiment period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h, 18h, 12h and 6 h; MBR conditions: aeration rate = 6L/min, DO = 6.4-7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h;

 $SRT = \infty$ )

- Figure 5.26 Relationship of Log D and pharmaceutical removals in the MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ )
- Figure 5.27 Relationship of Log D and pesticides removals in the MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L, HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.28 Relationship of Log D and industrial chemicals removals in the MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L, HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.29 Relationship of Log D and estrogenic hormones removals in the MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L, HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.30 Fate of the studied micropollutants in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.31 Fate of the studied micropollutants in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 18h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )

- Figure 5.32 Fate of the studied micropollutants in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 12h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.33 Fate of the studied micropollutants in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.34 Comparison of TMP profiles in the MBR while MBBR was operating at HRT of 24 h, 18 h, 12 h, and 6 h (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2$ ·h; HRT = 6h; SRT =  $\infty$ )
- Figure 5.35 Comparison of EPS and SMP values in the MBR while MBBR was operating at HRT of 24 h, 18 h, 12 h, and 6 h (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.1 TOC removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ )
- Figure 6.2 COD removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ )
- Figure 6.3 NH<sub>4</sub>-N removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2 \cdot \text{h}$ ; HRT = 6h; SRT =  $\infty$ )
- Figure 6.4 TN removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate = 8.83

 $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ )

- Figure 6.5 PO<sub>4</sub>-P removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2 \cdot \text{h}$ ; HRT = 6h; SRT =  $\infty$ )
- Figure 6.6 TOC removal in the CMBR system (conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.7 COD removal in the CMBR system (conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.8 NH<sub>4</sub>-N removal in the CMBR system (conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.9 TN removal in the CMBR system (conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2 \cdot \text{h}$ ; HRT = 6h; SRT =  $\infty$ )
- Figure 6.10 PO<sub>4</sub>-P removal in the CMBR system (conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.11 Variations of MLSS and MLVSS concentrations in the CMBR
- Figure 6.12 Nematodes (×10) in the CMBR
- Figure 6.13 Micropollutant removals in the CMBR and MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.14 Relationship of Log D and pharmaceutical compounds removals in the CMBR and MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )

- Figure 6.15 Relationship of Log D and pesticides removals in the CMBR and MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.16 Relationship of Log D and industrial chemicals removals in the CMBR and MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.17 Relationship of Log D and estrogenic hormones removals in the CMBR and MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2$ ·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.18 Fate of the studied micropollutants in the hybrid MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.19 Comparison of TMP profiles between the hybrid MBBR-MBR and the CMBR (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2$ ·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.20 Comparison of EPS and SMP values between the hybrid MBBR-MBR and the CMBR (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )
- Figure 6.21 Comparison of fouled membrane and after physical and chemical cleaning

## **List of Tables**

- Table 2.1
   Summary of adverse effects from micropollutants
- Table 2.2
   Concentration of selected micropollutants in aquatic environment
- **Table 2.3**Performance of different attached growth systems
- **Table 2.4**Biomass content on various attached growth media
- **Table 2.5**Data of 8 different plastic media (Stephenson et al., 2013)
- Table 3.1
   Composition and concentration of the stock solution
- **Table 3.2** Physicochemical properties of the selected trace organics
- **Table 4.1**Results for organic matters and nutrients removal by MBBR at HRT of<br/>24 h, 18 h, 12 h, and 6 h
- **Table 4.2**Results for both suspended and attached biomass in MBBR at HRT of24 h, 18 h, 12 h, and 6 h
- Table 4.3Comparison of micropollutant removal efficiency (%) in the MBBRand in other biological treatment technologies
- Table 5.1Results for organic matters and nutrients removal by MBBR-MBR at<br/>HRT of 24 h, 18 h, 12 h, and 6 h
- **Table 5.2**Results for both suspended and attached biomass in MBBR-MBR at<br/>HRT of 24 h, 18 h, 12 h, and 6 h
- **Table 5.3**Membrane resistance for different types of fouling
- **Table 5.4**Sludge characteristics in MBR unit at HRT of 24 h, 18 h, 12 h, and 6 h
- Table 6.1
   Membrane resistances for different types of fouling

## **List of Abbreviations**

CAS	Conventional activate sludge
DO	Dissolved oxygen
EPS	Extracellular polymeric substance
GAC	Granule activated carbon
GC-MS	Gas chromatography-mass spectrometry
HRT	Hydraulic retention time
MBBR	Moving bed biofilm reactor
MBR	Membrane bioreactor
MBBR-MBR	Moving bed biofilm reactor-membrane bioreactor
MF	Microfiltration
MLSS	Mixed liquor suspended solids
MLVSS	Mixed liquor volatile suspended solids
OLR	Organic loading rate
PAC	Powdered activated carbon
РСР	Personal care product
PPCP	Pharmaceutical and personal care product
PU	Polyurethane
PVA	Polyvinyl alcohol
PVDF	Polyvinylidene fluoride
SMP	Soluble microbial products
SND	Simultaneous nitrification-denitrification
SPE	Solid phase extraction
SRT	Sludge retention time
TMP	Transmembrane pressure
TN	Total nitrogen
TOC	Total organic carbon
WWTP	Wastewater treatment plant

## List of Symbols

Cs, a	Concentration of micropollutant on the attached biosolids ( $\mu g/g$ )
Cs, s	Concentration of micropollutant on the suspended biosolids ( $\mu g/g$ )
$C_{w,eff}$	Average effluent concentrations of micropollutants (ng/L)
$C_{w, inf}$	Average influent concentrations of micropollutants (ng/L)
J	Permeation flux $(L/m^2 \cdot h)$
Kow	Octanol-water partition coefficient
Ls	Load of micropollutant removed via sorption (ng)
L <sub>b</sub>	Load of micropollutant removed via biodegradation (ng)
Linf	the influent load of micropollutants over the experimental period (ng)
Leff	the effluent load of micropollutants over the experimental period (ng)
MLSS	Mixed liquor suspended biosolids concentration (g/L)
pKa	Acid dissociation constant
$\Delta P_T$	Transmembrane pressure (kPa)
Q	Flow rate of the MBBR (L/day)
R <sub>c</sub>	Cake resistance formed by cake layer deposited over membrane
	surface (m <sup>-1</sup> )
$R_{\mathrm{f}}$	Fouling resistance caused by pore plugging and/or solute adsorption
	onto the membrane pore and surface $(m^{-1})$
R <sub>m</sub>	Intrinsic membrane resistance caused by membrane itself and
	permanent resistance (m <sup>-1</sup> )
$\Delta SS$	Increased amount of attached biosolids over the study period (g)
Т	Duration of the study period (day)
μ	Viscosity of the permeate $(m^2/s)$

## Abstract

The ubiquitous occurrence of micropollutants and their metabolites in the aquatic environment has posed threats to living organisms to a great extent. However, effective micropollutants removal normally requires longer hydraulic retention time (HRT) when using biological treatment systems. As an ideal and low-cost material for attached-growth microorganisms, polyurethane sponge has exhibited high potential to eliminate micropollutants. In this study, a sponge-based moving bed biofilm reactor (MBBR) was investigated at four different HRTs (24, 18, 12, 6 h), to better understanding of the effect of HRT on micropollutant removal. The MBBR as pretreatment to a membrane bioreactor (MBBR-MBR hybrid system) was also evaluated. Four groups of frequently detected micropollutants in wastewater (total 22 compounds) were selected, namely pharmaceuticals and personal care products (PPCPs), pesticides, hormones and industrial chemicals.

The MBBR alone showed stable and effective removals of TOC (92.6% - 95.8%), COD (93.0% - 96.1%) and NH<sub>4</sub>-H (73.6%-95.6%) at all HRTs while improving PO<sub>4</sub>-P removal at HRT of 18 h. The MBBR showed the highest performance efficiency for removing DOC, COD, NH4-N, PO4-P and TN at HRT of 18 h, which were 96.1±0.4%, 97.4±0.8%, 91.1±1.6%, 49.9±7.2%, and 72.3±6.9%, respectively. This could be explained by the food to microorganisms (F/M) ratio in the MBBR. In addition, higher NH<sub>4</sub>-N removal at HRT of 18 h could be attributed to the increased population of ammonium oxidation bacteria in the MBBR unit. Moreover, the use of phosphate for biomass growth and the phosphorus uptake by phosphate accumulating organisms (PAOs) could contribute to the high removal of PO<sub>4</sub>-P at HRT of 18 h. In terms of micropollutants removal, MBBR achieved comparable removal compared to other biological treatment such as activated sludge processes and membrane bioreactor. Although the micropollutants were subjected to biodegradation and sorption, the results indicated compound-specific variation in removal at all HRTs, ranging from 10.7% (carbamazepine) to 98.4% (ibuprofen). Among the selected micropollutants, most of them were biodegradable excluding carbamazepine, fenoprop and metronidazole. In addition, the micropollutants removal could remain constantly high even at lower HRTs with more consistent removal efficiency over the

experimental period (except for carbamazepine, fenoprop,  $17\alpha$ -ethinylestradiol and 4-tert-octylphenol). Particularly, at HRT of 18 h, the removal of diclofenac was significantly improved by more than 30% and the removals of ketoprofen, gemifibrozil, acetaminiphen, bisphenol A, and pentachlorophenol were also better. Overall, HRT of 18 h was the optimum HRT for biological degradation of the micropollutants in the MBBR.

When using an MBBR as pretreatment to an MBR, the MBBR-MBR hybrid system achieved better removal efficiencies for selected micropollutants, such as metronidazole and carbamazepine. Both metronidazole and carbamazepine are nitrogen bearing compounds, where nitrogen is bound to the cyclic structure. The infinite SRT applied in this study could have facilitated the enhanced removal of the nitrogenous compounds. Even MBR can prevent the washout of slow-growing microorganisms like nitrifiers, the impact of MBR removal was minimal at all HRTs. This may probably due to the low MLSS concentration and the large pore size (0.2 µm; two orders of magnitude larger than the molecular sizes of micropollutants) of the MF membrane used in this study. In addition, a longer HRT (e.g. HRT of 24 h or 18 h) can significantly mitigate membrane fouling when compared with a relatively short HRT (e.g. HRT of 6 h). Especially, the TMP value maintained less than 15 kPa for 60 days (HRT of 18 h) and 68 days (HRT of 24 h). The level of EPS were similar at the beginning of all HRTs, then gradually increased to 15.24 mg/L, 16.43 mg/L, 19.88 mg/L and 22.93 mg/L at the end of operation for MBBR unit under HRT of 24 h, 18 h, 12 h, and 6 h, respectively. The SMP concentration varied for different HRTs but showed minor variation under the same HRT. The SMP concentration was lower at HRT of 24 h, while a significantly higher SMP concentration was observed at HRT of 6 h. As a whole, the MBBR-MBR hybrid system showed improvement in both micropollutants elimination and mitigation in membrane fouling.



University of Technology, Sydney

Faculty of Engineering and Information Technology

# Chapter 1

Introduction

#### 1.1 Background

Under the strong impetus of ever-growing industrialisation, the continued rise in chemical consumption is to achieve high material living standards. The extensive use of chemicals is mainly in medicines, pesticides, personal care products and industrial merchandises. The constant input of these problematic substances leads to their accumulation in the ecosystem, posing great threads to public health and environment. In addition, the mentioned substances are mostly polar and persistent, which affect fundamental biochemical processes in the aquatic environment, even at trace concentrations. We commonly refer them as 'micropollutants'. The main pathway of micropollutant contamination is through wastewater discharge from diverse sources (e.g. municipal wastewater).

Due to the nature of micropollutants, many of these potentially harmful compounds are not biodegradable and do not absorb easily, which makes it difficult or impossible to eliminate them. Moreover, the effluents containing micropollutants from conventional wastewater treatment plants (WWTPs) cannot be well-treated with current treatment systems. In order to improve the water quality and reduce the potential negative ecological effects, development of various treatment processes is essential to ensure the adequate removal of micropollutants.

#### **1.2 Impact of micropollutants and reduction strategies**

A wide range and a large amount of chemicals used on a daily basis at homes, in workplaces or in the urban environment go into the aquatic environment in many ways. Fig. 1.1 illustrates the possible source and pathways for the introduction of micropollutant into the environment. These micropollutants are pharmaceutical and personal care products (PPCPs), steroid hormones, surfactants, industrial chemicals, and pesticides. They mostly end up in the sewer system and being treated in the wastewater treatment plants (WWTPs). However, micropollutants often experience inadequate removal due to the low concentration, diverse physiochemical properties (hydrophobicity, biodegradability, and volatility etc.) and types of treatment. The micropollutants residue in the treated effluent discharging from WWTPs becomes a common route into the aquatic environment.

Besides municipal wastewater, the accumulation of micropollutants could also from hospital discharges, industrial wastewater, stormwater runoff, agricultural runoff, landfill leachate, and livestock waste (Nakada et al., 2008 ;Hillebrand et al., 2012; Kuroda et al., 2012; Tran et al., 2014). For example, a possible route for agricultural pesticides could be migrate from the soil surface through the soil zone, the unsaturated zone and the saturated zone in the well-established way (Stuart et al., 2012). In addition, landfill leachates may contain a large amount of pharmaceuticals, personal care products, as well as a range of industrial compounds (Barnes et al., 2004; Buszka et al., 2009).

Groundwater not only supplies a large portion of public water supply, but also provides water for industry and irrigation, baseflow support to surface water (Clara et al., 2004; Tran et al., 2013; Tran et al. 2015). The interaction between groundwater and surface could consider being another important pathway for various micropollutants. As some treated discharge from industrial premises and sewage treatment is going to surface water directly, certain untreated micropollutants may then infiltrate to groundwater.



Figure 1.1 Schematic presentation of diffuse and point source entry paths of micropollutants in the environment

To combat the current micropollutant-related issues, it is imperative that nontechnological measures (e.g., regulations) and technological methods (e.g., source control and wastewater treatment techniques) are economically and effectively available. Today's wastewater treatment plants (WWTPs) are primarily designed to remove solids, biodegradable substances and nutrients. However, the presence and detection of micropollutants has posed a new challenge for the current wastewater treatment processes to ensure the quality of treated effluent. In this context, attachedgrowth processes are dramatically emerging and have been considered as an attractive choice for upgrading current WWTPs. Compared to conventional activated sludge (CAS) processes, attached-growth processes perform more efficiently, due to their simplicity of operation with low operational and maintenance cost, feasibility of growing microorganisms with relatively low specific growth rates, tolerance to variation in hydraulic and pollutants loading rates, higher biomass concentration and treatment stability, better oxygen transfer and high nitrification rate and greater resistance to toxicity (Albertson, 2000; Guo et al., 2012; Loupasaki and Diamadopoulos, 2012).

The moving bed biofilm reactor (MBBR) is a technology based on the concept of attached growth process. The MBBR delivers a flexible, cost-effective, and easy-tooperate means to address current wastewater issues and the expandability to meet future loads and increasingly stringent discharge requirements. Continued research and experience have demonstrated that MBBRs can offer enhanced biological process and improved treatment efficiency, and are especially well suited to biological nutrient removal (Ødegaard, 2006; Kermani et al., 2008; Di Trapani et al., 2008). While the MBBR has become a proven technology for eliminating conventional contaminants, little is known about the micropollutant removal capacity of this treatment technique. Therefore, the paucity of knowledge has led to some recent efforts to evaluate the effectiveness of MBBR technology for micropollutant elimination (Falås et al., 2012; Falås et al., 2013; Lim et al., 2013; Luo et al., 2014a; Hassani et al., 2014; Dvořák et al., 2014). Encouragingly, these studies have shown successful results for micropollutant removal applications. It has been generally agreed that, by adding attached growth carriers, the MBBR system is able to provide favourable conditions, such as a diverse microbial community and the coexistence of different redox conditions, for the decomposition of a wider spectrum of micropollutants.

Despite the advantages of MBBR, the major concern of its applications is the decrease of sludge settleability when treating high strength wastewater, which may lead to severe operational problems when clarifiers are employed as solid separation systems. To counter this, various hybrid systems have been developed, which involve modifications of the basic MBBR system by adding coagulants (metal salts or cationic polymers) or applying membrane filtration or floatation as the solid separation process (Ødegaard et al., 2006). Among all these modifications, combining membrane technology with MBBR is an established concept with growing popularity, which also results in better membrane performance (Yang et al., 2008; Phattaranawik and Leiknes, 2010; Sun et al., 2012; Duan et al., 2013; Ye et al., 2013). Although there are limited number of studies concerning MBBR-membrane

hybrid systems, some of the results highlighted the potential to mitigate membrane fouling due to the decreased suspended solid environment in the process, regardless system configurations and the choices of biocarriers (Leiknes et al., 2006; Ivanovic et al., 2006; Ivanovic and Leiknes, 2008; Dong et al., 2014). However, some researchers indicated that the MBBR-membrane filtration hybrid system could experience severe membrane fouling when large amounts of submicron colloidal particles were present in the reactor (Leiknes et al., 2006; Sun et al., 2012). In addition, the occasional sludge bulking in the hybrid systems could also contribute on a negative level to the membrane performance.

#### **1.3 Objectives of the research**

The research aims to evaluate the removal efficiency of various organic micropollutants (refer as "micropollutants" in this thesis) in a sponge-based MBBR operating under different HRTs. In addition, a hybrid system combining the MBBR with a submerged membrane bioreactor (MBBR-MBR) was also investigated in terms of micropollutants removal and membrane fouling behaviour. Overall, the objectives of this study are listed as follows:

- To investigate the long-term micropollutants removal in a sponge-based MBBR under different HRTs (HRT of 24h, 18h, 12h and 6h);
- To elucidate the fate of the micropollutants during the treatment in the MBBR
- To evaluate the performance of the MBBR-MBR in terms of micropollutants removal and membrane fouling.

#### **1.4 Outline of the thesis**

This thesis is comprised of six chapters as shown in Figure 1.2. Chapter 1 introduces the background and objectives of this study. Chapter 2 reviews the use of attached growth system regarding micropollutants removal. Chapter 3 describes the materials, experimental set-ups, operational conditions and analytical methods for the MBBR and MBBR-MBR hybrid system. Chapter 4 presents the results of operation of the MBBR system under HRT of 24 h, 18 h, 12 h and 6 h, in terms of micropollutants removal. In Chapter 5, the results of micropollutants removal and membrane fouling were obtained from the MBBR-MBR hybrid system and were analysed to gain the

optimal condition for the MBBR-MBR system and membrane fouling reduction. The final chapter draws the conclusion for all the findings from this study and also gives some recommendations for the future research.



Figure 1.2 The outline of the thesis



University of Technology, Sydney

Faculty of Engineering and Information Technology

## Chapter 2

**Literature Review**
# 2.1 Introduction

During the last decades, the concerns over water scarcity and more stringent discharge limits have driven the demand for more advanced and cost-effective technical solutions to wastewater treatment. In this context, attached-growth processes are dramatically emerging and have been considered as an attractive choice for upgrading current wastewater treatment plants (WWTPs). Compared to conventional activated sludge (AS) processes, attached-growth processes perform more efficiently, due to their simplicity of operation with low operational and maintenance cost, feasibility of growing microorganisms with relatively low specific growth rates, tolerance to variation in hydraulic and pollutants loading rates, higher biomass concentration and treatment stability, better oxygen transfer and high nitrification rate and greater resistance to toxicity (Albertson, 2000; Guo et al., 2012; Loupasaki and Diamadopoulos, 2012).

Attached growth processes are biological treatment processes employing media that supports biomass on its surface and within its porous structure. This highly active attached biofilm enables the attached growth systems to provide more efficient treatment. The attached growth systems include trickling filter (TF), submerged fixed-bed biofilm reactor, granular media biofilter, rotating biological contactor (RBC), fluidized bed reactor, and moving bed biofilm reactor (MBBR). To date, several review papers have been published regarding the use of attached growth reactors in wastewater treatment and the factors that affecting the treatment efficiency. Barwal and Chaudhary (2014) summarized the performance of biocarriers in MBBR, and pointed out that the attached biofilm on media degrades dissolved pollutants in the wastewater stream. Each biofilm media adds productivity via the provision of an active surface area sustaining bacteria within protected cells. It is high-density population of bacteria that achieves high-rate biodegradation with the attached growth system, therefore making the effective surface area of media an important design parameter. Ivanovic and Leiknes (2012) reviewed the status of MBR technology with biofilm implementation for wastewater treatment to identify performance and operational characteristics. They reviewed that the utilization of biofilm media in MBR was to reduce the negative impact of suspended solid,

improve the filterability and lower membrane fouling, enhance nutrient removal and reduce the membrane cake layer formation. Guo et al. (2012) reviewed the roles of various attached growth media used in hybrid membrane processes in enhancement of system performance. They concluded that MBR-coupled attached growth processes have outstanding advantages over the conventional biological treatment process, because attached growth media (e.g. BAC, plastic media and porous sponge) is inclined to improving the performance of MBR by migrating membrane fouling and retaining the membrane permeability. Loupasaki and Diamadopoulos (2012) outlined the design and operational characteristics of attached growth processes in terms of packing media, organic loading rates, temperatures, and achieved removal rates. The review mentioned that the media material selection was crucial in determining what microorganisms would be established and dominated the system. McQuarrie and Boltz (2011) summarized and expended on MBBR process design procedures, provided an overview of some commercially available systems and their components. They mainly focused on the impact of some commercially used plastic media on the MBBR performance as a design parameter. Although part of the above reviews included some information concerning biofilm media and its impact on system performance, no attempt has been made to systematically summarize the effects of media material on various attached-growth systems. Due to the specialty of the attached-growth process, the media is of great importance to maintain a large amount of active biomass and a wide variety of microbial populations, as the high specific surface that provided by attached growth media is good for biofilm attachment and development. Furthermore, the media plays a key role in governing biofilm attachment, because the contaminants in wastewater can be removed by biological degradation incorporated into the attached microbial biomass, adsorbed to the media material or to the microbial biofilm (Guo et al., 2009). Additionally, the characteristics of media determine the structure of the biofilms developed in the reactors, as well as the reactor operation mode and the process effectiveness. Therefore, to better understand the roles of media materials in the attached growth systems, this review was to sum up properties of various biofilm supporting media, as well as to offer a comprehensive overview about their influences on the performance of attached growth systems. The future research needs regarding media materials were also included.

# 2.2 Micropollutants contanmination

With the revolutionized development of technologies and significant advancement of analytical methods, micropollutants have been frequently detected in wastewater recently. Despite the low concentration of micropollutants, they raise considerable ecological issues and are a major concern to the living organism. In addition, most of these micropollutants, such as pharmaceuticals and personal care products (PPCPs) and various industrial chemicals, are not metabolized and have been discharged into sewer system, which pose challenges to the current operation and future upgrade of wastewater treatment plants (WWTPs). Therefore, necessary methods should be taken to protect the aquatic environment from micropollutant contamination.

#### 2.2.1 Adverse effects of micropollutants

The issues of micropollutants have been considered as a detrimental threat to the environment, especially the toxicity on living organisms, due to their biologically active and persistent characteristics. The adverse impacts of micropollutants can be different according to their categories (including PPCPs, steroid hormones, pesticides, and industrial chemicals), doses, and physiochemical properties (Table 2.1).

For PPCPs, pharmaceuticals are basically commercial drugs and medicines that are taken to treat illness, disease, and medical conditions in both humans and animals, while personal care products include a wide variety of compounds, such as perfumes, masks, and make ups etc. These PPCPs are mostly persistent and remain biologically active after disposal. For instance, several antibiotics not only have been reported to significantly inhibit algae and cyanobacteria growth with low concentration, but also potentially favour the development of antibiotics resistant pathogens when they reach certain concentration (Ebert et al., 2011; González-Pleiter et al., 2013; Sandegren, 2014). They also may have unexpected biochemical interactions when mixed with other compounds, and may concentrate in food chain and affect the aquatic organisms. Most of these impacts are chronic sub-lethal, but it also means that constant exposure to very low level of PPCPs can be harmful, especially when the municipal WWTP effluents could cause surface water contamination, raising the

question of drinking water pollution. Therefore, efforts should be made to protect the aquatic environment and human health.

In terms of steroid hormones, they are steroids that act as hormones, including a variety of natural and synthetic compounds. Steroid hormones help control metabolism, inflammation, immune functions, salt and water balance, development of sexual characteristics, and the ability to withstand illness and injury. Natural steroid hormones (estrone, estriol and  $17\beta$ -estradiol) are mostly found in human and mammalian urine, while synthetic hormones (ethinylestradiol) are widely used as ingredient of contraceptive pills. Because steroids are lipid-soluble, they can diffuse fairly freely from the blood through the cell membrane and into the cytoplasm of target cells, hence exposure to them from the environment widely impacts biological communities. The high endocrine disrupting activity of steroid hormones cannot be ignored.

Regarding pesticides, they are substances meant for attracting, seducing, and then destroying any pest. The primary upsides of pesticides include controlling 1) pests and plant disease vectors, 2) human/livestock disease vectors and nuisance organisms and 3) organisms that harm other human activities and structures. Despite the benefit of pesticides, their impacts on the environment and human health still are of great concerns. The United States Environmental Protection Agency pointed out that pesticide exposure can cause a variety of adverse health effects, ranging from simple irritation of the skin and eyes to more severe effects such as affecting the nervous system, mimicking hormones causing reproductive problems, and also causing cancer. In addition, some persistent pesticides accumulate in the contaminated area as a result of their widely usage in the past.

As for industrial chemicals, their toxicity to the aquatic living organisms is of great concern. Bisphenol A (BPA) is an organic synthetic compound that commonly used in manufacturing consumer plastic goods, such as water bottles, sports equipment, CDs, and DVDs. In addition to being present in many products that people use daily, BPA has the ability to bioaccumulate, especially in water bodies, even after wastewater treatment.

Micropollutants	Examples of adverse effects on aquatic organisms	References
PPCPs	• Estrogenic/mutagenic activity and genotoxicity causing by bioaccumulation of diclofenac, ibuprofen and gemfibrozil in fish blood and plasma	Brown et al., 2007
	<ul> <li>Triclosan affects river biofilms and algae community structure at concentration potentially less than 0.5 μg/L</li> </ul>	Franz et al., 2008; Ricart et al., 2010
	• Triclosan and its bi-product can bioaccumulate in algae, snails and fish	Stasinakis and Gatidou, 2010
	• Extensive use of antibiotics could increase the development of antibiotic resistant bacteria in the environment	Dreising, 2011
	• Diclofenac residue accumulation led to extraordinarily high mortality of oriental white-backed vultures in India and Pakistan	Fatta-Kassinos et al., 2011
	•	
Steroid hormones	• Even low level of steroid hormones could cause feminization of aquatic and terrestrial special such as frogs, turtles and mice	Stasinakis and Gatidou, 2010
	<ul> <li>4 ng/L concentration of ethinylestradiol prevented development of secondary sexual characteristics of male fathead minnows</li> </ul>	Länge et al., 2001
	• Estrogen use led to prostate cancer development	Hess-Wilson and Knudsen, 2006
Pesticides	Human sperm DNA damage caused by several organophosphoric pesticides	Salazar-Arredondo, 2008
	• Poisoning events associated with organophosphate insecticides and carbamates are frequently reported	Li et al., 2014
Industrial chemicals	• Interaction with protein and its endocrine disruptor characteristics, Bisphenol A is toxic to aquatic and terrestrial organisms	Rubin, 2011

1 abic 2.1	Summary	of duverse	enteets	monn	meroponu	.unts	
Table 2.1	Summary	of adverse	effects	from	micropollut	ants	

### 2.2.2 Occurrence of micropollutants in aquatic environment

A wide range and a large amount of chemicals used on a daily basis at homes, in workplaces or in the urban environment go into the aquatic environment in many ways. Fig. 2.1 illustrates the possible source and pathways for the introduction of micropollutant into the environment. These micropollutants are pharmaceutical and personal care products (PPCPs), steroid hormones, surfactants, industrial chemicals, and pesticides. They mostly end up in the sewer system and being treated in the wastewater treatment plants (WWTPs) (Stuart et al., 2012). However, micropollutants often experience inadequate removal due to the low concentration, diverse physiochemical properties (hydrophobicity, biodegradability, and volatility etc.) and types of treatment. The micropollutants residue in the treated effluent discharging from WWTPs becomes a common route into the aquatic environment.

Besides municipal wastewater, the accumulation of micropollutants could also from hospital discharges, industrial wastewater, stormwater runoff, agricultural runoff, landfill leachate, and livestock waste (Nakada et al., 2008 ;Hillebrand et al., 2012; Kuroda et al., 2012; Tran et al., 2014). For example, a possible route for agricultural pesticides could be migrate from the soil surface through the soil zone, the unsaturated zone and the saturated zone in the well-established way (Stuart et al., 2012). In addition, landfill leachates may contain a large amount of pharmaceuticals, personal care products, as well as a range of industrial compounds (Barnes et al., 2004; Buszka et al., 2009).

Groundwater not only supplies a large portion of public water supply, but also provides water for industry and irrigation, baseflow support to surface water (Clara et al., 2004; Tran et al., 2013; Tran et al. 2015). The interaction between groundwater and surface could consider being another important pathway for various micropollutants. As some treated discharge from industrial premises and sewage treatment is going to surface water directly, certain untreated micropollutants may then infiltrate to groundwater.



**Fig. 2.1** Contamination pathways for micropollutants into the aquatic environment (adapted from "Review of risk from potential emerging contaminants in UK groundwater" by Stuart et al., 2012, *Science of the Total Environment*, 416, 1-21. )

In general, pharmaceuticals are biologically active compounds and they are intended not to be easily biodegradable but are water soluble. A wide range of detected pharmaceutical products in the aquatic environment include: veterinary and human antibiotics (e.g. ciprofloxacin, erythromycin, lincomycin, sulfamethoxazole, and tetracycline); prescription drugs (e.g. codeine, salbutamol, and carbamazepine); nonprescription drugs (e.g. acetaminophen, ibuprofen, and salicylic acid) and iodinated X-ray contrast media (e.g. iopromide and iopamidol) (Barnes et al., 2008; Miller and Meek, 2006; Nikolaou et al., 2007; Pérez and Barceló, 2007; Ternes and Hirsch, 2000; Vulliet and Cren-Olivé, 2011; Watkinson et al., 2009; Stuart et al., 2012). They primarily entered the environment through human excretion, disposal of unused products and agricultural use (Poynton and Vulpe, 2009). Personal care products (such as bacteriocide, antifungal agents, polycyclic musks and UV filters/sunscreen etc.) are commonly transmitted into the environment via wastewater treatment discharges. The possible fates of personal care products and their metabolites are as follows: a) mineralization to  $CO_2$  and water; b) retention of the solids portion (sludge/biosolids) if the compound entering the plant or the product of biologically mediated transformation is lipophilic; c) release to the receiving water either as the original compound or as a degradation product (Jiang et al., 2013). For instance, triclosan is widely used in household products and its metabolite methyl triclosan are considered to be persistent (Stuart et al., 2012).

Hormone steroids are a group of endocrine disruptors that are synthesized from cholesterol and have in common a cyclopentan-*o*-perhydrophenanthrene ring (Ying et al., 2002). Some types of hormone steroids (estradiol, estrone and estriol etc.) are essential to human for maintaining the health of the reproductive tissues, skin and brain. Some synthetic steroids (ethynylestradiol and mestranol) are used as contraceptives. Hormone steroids have been detected in wastewater treatment plant effluent and surface water, receiving from domestic discharge and animal waste disposal.

As pesticides post threat to the environment, some compounds have been banned or replaced other substitutes (Stuart et al., 2012; Gavrilescu et al., 2015). Current studies have been focused on pesticide metabolites that are often being detected in water sources and wastewater effluents at high concentrations and also biologically active and toxic (Lapworth and Gooddy, 2006; Clausen et al., 2007).

Industrial chemicals have been found to bioaccumulate and have potential endocrine disrupting properties. The ingredients of the production of alkyl phenol ethoxylates (APEs) are octyl- and nonyl-phenol. APEs are used in the manufacture of surfactants and also are used as pesticide adjuvants, therefore these can be found in groundwater as a result of agricultural activity (Stuart et al., 2012).

The concentration levels of selected micropollutants are listed in Table 2.2. PPCPs such as non-steroid anti-inflammatory drugs and antibiotics have frequently been found in raw sewage at concentrations reaching mg/L level (Kolpin et al., 2002; Ashton et al., 2004). Ibuprofen is the most abundant PPCPs found in raw sewage followed by the gemfibrozil, naproxen, salicylic acid and ketoprofen, possibly because of their excessive usage. About 160 different pharmaceuticals and their

metabolites have been identified in domestic wastewater treatment plants, surface water and groundwater in Northern Europe (Kallenborn et al., 2008). Because of high usage possibly triggered by health effect at cold climates and the readily access to medical treatment, high concentrations of antibiotics, analgesic and non-steroidal anti-inflammatory contaminants (e.g. ibuprofen, naproxen, ketoprofen, diclofenac salicylic acid and acetaminophen) have been detected in raw sewage and wastewater treatment plant effluent in Europe and North America (Pal et al., 2010). For example, concentration of near 20, 000 ng/L of ibuprofen has been detected in hospital wastewater (Gómez et al., 2006). The concentration of acetaminophen was 36,950 ng/L in hospital effluents of Taiwan and 16,020 ng/L in that of Spain, while its concentrations were less than 100 ng/L in wastewater treatment plant effluents and rivers (Lin et al., 2008; Gómez et al., 2006; Pedrouzo et al., 2007; Bueno et al., 2007; Kim et al., 2007).

The concentrations of estriol, estrone,  $17\beta$ -estradiol and  $17\alpha$ -ethinylestradiol (EE2) exceed the reported lowest predicted no-effect concentration (PNEC) values in freshwaters of most countries (Pal et al., 2010). Labadie et al. (2007) reported that estrogens (such as estrone) could transport through riverbed sediments and penetrate into the underlying groundwater. The concentration level of these contaminants in Asia and Australia is relatively low (Kallenborn et al., 2008). Excessive concentrations of estrogenic contaminants have also been detected in hospital effluent, wastewater treatment plant effluents and surface waters in USA and Europe. This could mainly be attributed to the animal husbandry and excessive usage of contraceptives. It could also indicate insufficient removal of hormone compounds in wastewater treatment plants.

Surface water recharge using the treated effluent has been the primary cause for the presence of micropollutants in surface water bodies (Luo et al., 2014a; Kasprzyk-Hordern et al., 2008). It is notable that ibuprofen, carbamazepine, triclosan, and pentachlorophenol were detected at alarmingly high levels with concentrations of several  $\mu$ g/L. Micropollutants are subjected to dilution, sorption onto suspended solids and sediments, photolysis and biodegradation after discharge to the surface water. Occurrence level of micropollutants in surface water depends on the

abundance and the persistency of the compound. Ibuprofen is a ready biodegradable compound, but is the most abundant in raw sewage also detected in surface water at high concentration. Triclosan and pentachlorophenol are persistent compounds, present in raw sewage at high levels, and also rich in surface water. In contrast, steroid hormones were detected at low concentrations in surface water either due to low level of raw sewage and efficient removal during treatment.

In comparison to the surface water, ground water contamination is found to be low. Landfill leachate and artificial recharge using treated wastewater have been a significant cause for micropollutants contamination in ground water. Micropollutant polarity is a vital factor for their occurrence in ground water. Polar compounds have less affinity for subsoil and are likely to infiltrate through soil and contaminate the ground water. Most pharmaceutical compounds are polar (e.g. carbamazepine, diclofenac and primidone), and under recharge conditions, can leach through the subsoil and contaminate the groundwater (Heberer 2002b). Additionally, compounds with a high molecular weight and a high log  $K_{ow}$  of >5 are easily sorbed to sediments and can be primarily removed by coagulation. Hence, such compounds are unlikely to be present in surface water (Vieno et al., 2007). On the other hand, compounds with low log  $K_{ow}$  of <2.5 have low sorption, and are thus likely to be present in surface water (Mompelat et al., 2009).

Micropollutants		Concentra	ntion (ng/L)		Reference
	WWTP effluent	Surface water	Ground water	Raw sewage	
PPCPs					
Salicylic acid	<2098	<302	6.5	340-8000	Pal et al., 2010; Heberer 2002a; Yu et al., 2013
Metronidazole	0.055	-	-	-	Gavrilescu et al., 2015
Ketoprofen	20-1620	3.4-329	<80	80-5700	Pal et al., 2010; Yu et al., 2013
Acetaminophen	1.8-220	4.1-777	<5.0-4689	1520- 182853	Gavrilescu et al., 2015; Tran et al., 2014
Naproxen	1-5100	1-610	1.2-263	8000	Pal et al., 2010; Yu et al., 2013; Luo et al., 2014
Primidone	110-200	55-635	-	-	Díaz-Cruz & Barceló, 2008
Ibuprofen	20-48240	<5044	<200	1000- 56500	Kolpin et al., 2002; Ashton et al., 2004; Heberer 2002b; Santos et al., 2007; Andreozzi et al., 2007; Yu et al., 2013; Joss et al., 2005
Diclofenac	8.8-5450	1.1-568	<380	86-1000	Pal et al., 2010; Ashton et al., 2004; Heberer 2002b; Andreozzi et al., 2007; Yu et al., 2013; Joss et al., 2005; Kasprzyk-Hordern et al., 2008
Carbamazepine	73-2100	<1075	<10.4	1000-2000	Pal et al., 2010; Heberer 2002a; Andreozzi et al., 2007; Joss et al., 2005;

 Table 2.2 Concentration of selected micropollutants in aquatic environment

					Kasprzyk-Hordern et al., 2008; Standley et al., 2008			
Gemifibrozil	<4000	1.8-790	<340	1090-8500	Pal et al., 2010; Kolpin et al., 2002; Heberer 2002b; Yu et al., 2013;			
Triclosan	12-9300	35-2300	2-118	180-4400	Stasinakis & Gatidou, 2011; Kolpin et al., 2002; Yu et al., 2013; Peng et al., 2008; Gómez et al., 2007			
Steroid hormones								
Estriol	0.4-30	5-19	149-1661	0.5-10	Pal et al., 2010; Laganà et al., 2004; Cargouet 2004			
Estrone	1-196	1-65	0.7-79	1-160	Pal et al., 2010; Yu et al., 2013; Kuch & Ballschmiter, 2001; Laganà et al., 2004; Vanderford et al., 2003; Servos et al., 2006			
17α-Ethinylestradiol	1-17	0.1-831	-	1-15	Pal et al., 2010			
17β-Estradiol	1-43	5-21.4	-	1-15	Pal et al., 2010; Kuch & Ballschmiter, 2001; Laganà et al., 2004; Servos et al., 2006; Cargouet 2004			
Pesticides								
Pentachlorophenol	-	2000	-	-	Kolpin et al., 2004			
Industrial chemicals								
Bisphenol A	4.8-800	0.5-140	5-2550	60-600	Kolpin et al., 2002; Yu et al., 2013			
4-tert-Octylphenol	<1000	-	-	80-3900	Kolpin et al., 2002;			
4-n-Nonylphenol	<4400	0.1-7300	-	220-870	Kolpin et al., 2002; Yu et al., 2013; Shao et al., 2005; Jonkers et al., 2009			

#### 2.2.3 Removal of micropollutant in wastewater treatment

Activated sludge treatment technology is most commonly used biological treatment process for conventional wastewater treatment plants (WWTPs). However, activated sludge process is not specifically designed to remove micropollutants. Therefore, the removal of micropollutants in conventional WWTPs is inefficient and often inadequate (Luo et al., 2014a). The micropollutant removal depends on certain operating parameters including redox conditions, nitrification/denitrification capacity, solid retention time, hydraulic retention time, temperature and mixed liquor pH value (Tadkaew et al., 2010; Hai et al., 2011a; Hai et al., 2011b; Clara et al., 2005; Joss et al., 2004; Czajka and Londry, 2006; Miège et al., 2008). In addition, compound specific parameters, such as polarity, molecular structure and biodegradability, also have significant influence on micropollutant removal during biological treatment (Tadkaew et al., 2011; Hai et al., 2011c).

Fig.2.2 shows the main mechanisms for micropollutant removal in conventional wastewater treatment as follows: (a) biological transformation, (b) sorption, (c) volatilization and (d) abiotic degradation. Sorption and volatilization consist of a transfer of the micropollutant from one compartment (water) to another (solid or gas) while degradation leads to the transformation of the micropollutant. Complete mineralization produces water,  $CO_2$  and minerals.



Fig. 2.2 Main removal mechanisms of micropollutants in conventional WWTPs

Biological transformation is the main removal mechanism for many hydrophilic organic micropollutants in conventional wastewater treatment. Micropollutant concentrations in wastewater are usually too low (ng/L to  $\mu$ g/L) to support the growth of microorganisms or to induce the corresponding enzymes and/or cofactors for their biodegradation. Under this situation, there is probably no acclimatization / adaptation occurs at such low concentrations. Thus, biological transformation of micropollutants generally requires the presence of other growth substrates (e.g. carbon and energy sources) (Tran et al., 2013). Biotransformation of trace contaminants can be separated in two main processes: metabolic reactions on mixed substrate or co-metabolic reactions.

In metabolic reactions of mixed substrates, microorganisms use organic micropollutants and other organic compounds as a growth substrate. These substrates are used as energy (catabolism) and/or carbon source (anabolism) for their cell development (Margot 2015). Catabolic reactions lead to transformation of the pollutant to smaller molecules, ultimately until their complete bio-mineralization, i.e., their conversion to water, carbon dioxide and other minerals (Benner et al., 2013). Many bacterial strains are able to utilize and mineralize specific pollutants as the sole energy source, meaning that metabolic pathways exist for these substances. The degradation of these pollutants at very low concentrations requires the presence of other substrates that will sustain the growth of cells. However, high concentrations of easily biodegradable substrates in wastewater can repress the expression of these specific catabolic pathways. This preferential substrate selection may thus reduce micropollutant degradation until all the readily degradable substrates are consumed (Benner et al., 2013).

During co-metabolic reactions, micropollutants are not used as a growth substrate but are biologically transformed, by side reactions catalysed by unspecific enzymes (e.g. mono-oxygenases or di-oxygenases, nacetyltransferases, hydrolases) or cofactors produced during the microbial conversion of the growth substrate (Margot 2015). Therefore, co-metabolism can be defined as "the transformation of a non-growth substrate in the obligate presence of a growth substrate or another transformable compound" (Fischer and Majewsky, 2014). Co-metabolism often leads to the formation of transformation products (TPs), but these TPs may possibly be used as growth substrates for other microorganisms (Benner et al., 2013; Tran et al., 2013). Although co-metabolic transformations require the presence of a growth substrate, if present at high concentrations the substrate can reduce the transformation of some micropollutants by competitive inhibition, i.e., competition between the growth and the co-metabolic substrate to the non-specific enzyme active site (Plósz et al., 2012; Plósz et al., 2010).

Biological transformation of micropollutants in WWTPs depends on the sludge concentration, their biodegradability in the sludge, and the hydraulic retention time within the reactor. Bioavailability of the pollutant is a prerequisite for bio-transformation. The soluble fraction is considered as being available but the bioavailability of the sorbed fraction is assumed to be much lower (Pomiès et al., 2013). As the sorbed fraction is in equilibrium with the dissolved one, desorption occurs during the degradation of the soluble fraction. Thus, part of the sorbed fraction can also be degraded when no sequestered in the sludge.

Sorption onto sludge or particulate matter can be considered as an important removal mechanism for hydrophobic or positively charged micropollutants, especially when they are poorly biodegradable. Adsorption onto biological sludge can be differentiated into two main processes (Joss et al., 2006): (a) Hydrophobic *interactions* between pollutants and suspended solids or sludge components, such as extracellular polymeric substances (EPS) or the lipophilic cell membrane of microorganisms. (b) *Electrostatic interactions* between positive charged groups of the pollutant and the mainly negative charged surfaces of microorganisms or effluent organic matter. Other phenomena such as active/passive cells uptake (absorption by microorganisms), cationic exchanges, cationic bridges, surface complexation and hydrogen bridges may also play an essential role in sorption mechanisms (Pomiès et al., 2013). Adsorption is a complex process dependent on the physico-chemical properties of the pollutant (such as charge, hydrophobicity) and the properties of the sludge (such as surface charge, specific surface area, EPS content, oxidation degree of the organic matter, mineral content). Different adsorption capacities are thus observed among different sludge (primary or secondary, flocs or biofilms) (Barret et

al., 2010a; Mailler et al., 2013). Electrostatic interactions are influenced by the pH of the wastewater as slight variation in the pH can lead to either protonation (positively charged or neutral) or deprotonation (neutral or negatively charged) of compounds containing functional moieties with a pKa around 6-9.

Micropollutants not only sorb to particulate matter, but also onto colloidal particles, which are considered as part of the "dissolved" phase (Pomiès et al., 2013). Sorption onto dissolved or colloidal matters increases the solubility of hydrophobic substances, such as persistent organic pollutants, polycyclic aromatic hydrocarbons or heavy metals (Barret et al., 2010a; Barret et al., 2010b; Katsoyiannis and Samara, 2007). This means that the presence of dissolved or colloidal matters or dissolved organic carbon (DOC) in wastewater can significantly affect the partitioning of these pollutants between the "dissolved" and the "particulate" phases, limiting their removal by adsorption onto the sludge and therefore facilitating their discharge into the environment together with the treated effluent. Reversible sorption is usually process composed of two reactions which occur simultaneously: adsorption and desorption. Sorption equilibrium is reached when the rate of both reactions is equal (Joss et al., 2006a). The sorption kinetics of various pollutants onto secondary sludge, including hydrophobic PAHs and hydrophilic substances such as polar pharmaceuticals and pesticides, are reported to be fast, with sorption equilibrium reached in less than 0.5 to 2 h (Barret et al., 2010b; Ternes et al., 2004; Wick et al., 2011). The sorption equilibrium on colloids is reached even faster (<5 min) (Maskaoui et al., 2007). Due to longer hydraulic retention time (HRT) in biological treatments, equilibrium can be assumed for solid-liquid partitioning in WWTPs.

Volatilization of micropollutants can occur during wastewater treatment, occurring as surface volatilization but more significantly by stripping during aeration. The transfer of the pollutant from water to air depends on the volatility of the compound (Henry's law constant) and the operation conditions of the process (aeration, agitation, temperature and atmospheric pressure) (Pomiès et al., 2013). Stripping should not be considered as an option for water treatment if the gas flow is not treated afterwards, otherwise the WWTP could cause atmospheric pollution.

Organic micropollutants can potentially be degraded during wastewater treatment by abiotic reactions, such as photolysis, hydrolysis or reaction with other chemicals. Direct *photolysis* occurs when a photon is absorbed by a compound, leading to bond cleavage to form a new compound. Pollutants can be also degraded by indirect photolysis, due to the production, during sun irradiation of dissolved organic matter,  $NO_2^-/NO_3^-$  or  $HCO_3^-/CO_3^{2-}$ , of transient excited species (reactive oxygen, radicals) which can react with the pollutants (Wang and Lin, 2014). In conventional treatments, photolysis by natural sunlight is very restricted due to the low surface-tovolume ratio available for sunlight irradiation and the high turbidity of the wastewater, which strongly limits the penetration of light into the water. Phototransformation is not expected to be a significant degradation mechanism in conventional systems. Photolysis can play a significant role in wastewater treatment with open water lagoons for compounds having aromatic rings, heteroatoms, and other functional chromophore groups that can either absorb solar radiation or react with photogenerated transient species (Verlicchi and Zambello, 2014). Hydrolysis is the result of the cleavage of chemical bonds by substitution of an atom or group of atoms in an organic compound by a water molecule (Schwarzenbach et al., 2003). But not all micropollutants can be hydrolyzed. Rates of hydrolysis in water are strongly dependent on the pH and the temperature (Mabey and Mill, 1978). Rates usually increase rapidly with the temperature, and hydrolysis at high pH (basecatalyzed) is often faster than acid-catalyzed or neutral hydrolysis for many compounds (Mitchell et al., 2014). Hydrolysis half-lives  $(t_{1/2})$  of micropollutants at neutral pH and 25°C vary from few seconds (e.g., tert-butyl chloride) to thousands of years (e.g., trichloromethane) (Schwarzenbach et al., 2003). Pollutants with very fast hydrolysis rates are expected to be completely transformed in sewers before reaching the WWTP. On the other hand, compounds with  $t_{1/2} > 7$  days will not be significantly hydrolyzed (< 10%) during wastewater treatment (HRT < 24h). In domestic wastewater (pH 6.5-8 and 10-25°C), hydrolysis rates are relatively slow for most micropollutants ( $t_{1/2} > 7$  d) compared to biodegradation or sorption (Schwarzenbach et al., 2003). Thus, hydrolysis can be considered as a negligible removal mechanism in WWTPs.

The removal of micropollutants is also like to the treatment process conditions including solid retention time, hydraulic retention time, redox conditions, temperature and mixed liquor pH value. The higher the SRT, the less the amount of micropollutants to be adsorbed onto the sludge, as high SRT leads to longer sludge age for microorganism to biodegrade micropollutants. But high SRT is not necessary for better removal of micropollutants (Luo et al., 2014a). Longer HRT allows micropollutants that have slow/intermediate kinetics to experience more effective biodegradation or sorption. Redox conditions may affect the wastewater or sludge characteristics and the microbial biodiversity (Göbel et al., 2007), resulting in inefficient biodegradation of some micropollutants. The acidity or alkalinity of wastewater can have influence on the physiology of microorganisms and the solubility of micropollutants in wastewater (Cirja et al., 2008).

# 2.3 Background of the attached growth systems

## 2.3.1 Attached growth systems

As a promising and effective technology, the attached-growth process provides not only excellent performance in removing organic matters and nutrients, but also has a potential to eliminate micropollutants in recent years (Li et al., 2011; Falås et al. 2013; Luo et al. 2014a). Unlike suspended growth processes, the attached-growth technology possesses certain advantages due to the fixed biofilm that grows on media. The media can be either fixed or moving freely in attached-growth reactors. With the development of wastewater treatment technology, different configurations of attached-growth systems were invented and have been in use commercially, such as trickling filters, rotating biological contactors (RBCs), fixed media submerged biofilters, fluidized bed reactors, and moving-bed biofilm reactors etc.

The primary and well-known form of attached-growth system is trickling filter (TF), which was introduced in 1890 (Bitton, 2005). The TF was the main alternative to activated sludge for the secondary treatment of settled domestic wastewater for BOD and NH<sub>3</sub> reduction. The BOD removal by TF is between 65% and 90%, while the nitrification rate is from 75% to 95%. TFs can be operated as single-stage or two-stage in either single pass mode or recirculation mode (Guo et al., 2012), based upon

their hydraulic or organic loading rate and the media provided to support bacterial growth. There are several factors are important in the function of TF, particularly the filter media (Lekang and Kleppe, 2000). With respect to supporting media for TFs, the important characteristics to be considered are the specific surface and the percent of void space. The surface area is related directly to the available active biological population, while the void space is significant in conveying both waste and the oxygen required for stabilization. There are several materials can be used to trickling filters as supporting media: crushed limestone, large gravel, rock, ceramic material, treated wood, hard coal and plastic etc.

Rotating biological contactors (RBCs) are widely used in wastewater treatment for secondary treatment since they were market in the 1960s. The RBCs allow obtaining high performance in the removal of dissolved carbon and ammonia at the expense of less energy consumption than to use AS systems. In addition, another advantage of RBCs stems from the fact that the interfacial area generated is very high and practically independent of the speed of rotation (Patwardhan, 2003). The performance of RBCs depends on parameters like rotational speed, organic and hydraulic rates, HRT, and media material etc., while the RBC media further affects biofilm characteristics. Studies indicated that the biofilms on the media of RBCs were structurally heterogeneous, consisting of cell clusters and voids with spatial microbial distribution of nitrifiers and heterotrophs (Rodgers and Zhan, 2003). In full-scale disc RBCs, it is reported that the thickness of biofilm ranged from 0.5 to 4.5 mm (Cortez et al., 2008). The control of biofilm thickness is very important to avoid clogging or material fatigue stresses in wastewater treatment (Griffin and Findlay, 2002). The materials of media that used in RBCs are producing from Styrofoam, polycarbonate sheet, high-density polyethylene and other non-corrosive durable materials (Cortez et al., 2008; Guo et al., 2012).

Fluidized-bed reactor (FBR) was the most volumetric efficient biological reactor available in mid-70s because of the high biomass concentration covered on media surface (Harremoes and Henze, 2002). The level of biomass can be up to 1,000 times higher than the level found in conventional AS systems (Guo et al., 2012). The FBR technology can be used in industrial wastewater treatment, post-nitrification of

drinking water and municipal wastewater. In a FBR process, the biofilm carriers are usually small particles, such as granular activated carbon, sand, anion and cation exchange resins (Rodgers and Zhan, 2003). However, the FBRs are not widely utilized in wastewater treatment due to certain drawbacks, including complicated inlet and outlet design, high energy requirement and the difficulty in control the thickness of attached-growth biofilm.

Submerged fixed-bed biofilm reactor (SFBBR) works on the basis of biofilm attached to submerged monolithic plastic supports with no sludge recirculation, so the issues related to clogging of support media and control problem of biofilm growth can be effectively minimized. In this case, the media properties, such as durability, specific surface area density and void space percentage, are the main concerns for SFBBRs operation. Chapanova et al. (2007) reported that greater surface density permitted a larger biomass per unit volume, while greater void space allowed for a higher oxygen and mass transfer to the biofilm and reduced the clogging risks. Hence, those factors should be taken into consideration in order to improve the treatment efficiency of SFBBRs. If designed properly, the SFBBRs could consistently provide BOD removal  $\geq 95\%$ , COD removal  $\geq 80\%$  and TKN (Total Kjeldahl Nitrogen)  $\geq 90\%$  in municipal wastewater treatment (Schlegel and Koeser, 2007).

Moving-bed biofilm reactor is an effective and affordable attached-growth biological process and has been attracted increasing attention over the last decade in wastewater treatment. The basic concept of MBBR is to provide additional biomass attached on media, without increasing the suspended mixed liquid concentration in the reactor. The MBBR technology has showed success in organic matter and nutrient removal, designed to meet a wide range of effluent quality standards. More recently, MBBR has proven its potential in treating micropollutants in wastewater (Li et al., 2011; Falås et al. 2013; Luo et al. 2014a). The types of media can be used in a MBBR are polyethylene media, activated carbon, reticulated polyurethane, polymer foam pads, and nonwoven media etc.

Table 2.3 presents recent research data concerning the performance of different attached-growth systems. As can be seen from table 2.3, the attached-growth processes are widely used in wastewater treatment, which are not only applied for municipal wastewater, but also expended to treat some specific effluent, such as oilfield produced wastewater and coal gasification wastewater. In addition, the treatment efficiencies are varied even for the same type of attached-growth system, when different media is employed. For instance, Chu and Wang (2011) examined two different media in a MBBR, showing that the TOC removal was 90% for polyurethane foam while 72% for biodegradable polymer.

					Org	anic	Nutr	ient	Reference
Attached- growth system	Support media	Influent	HRT	SS removal efficiency	Load	Removal efficiency	Load	Removal efficiency	
BAF	Expended clay	Sedimentation effluent	2.44 h	65±5%	3.7-8 kg COD/m3/d	35±5%	0.78±0.12 kg NH4-N /m <sup>3</sup> /d	91±2%	Farabegoli et al., 2009
	Ceramic or			93%		88%	<1.21	73%	
BAF	Zeolite or	Synthetic wastewater	0.5-2.5 h	90%	< 6  kg COD/m <sup>3</sup> /d	85%	< 1.2 Kg	85%	Qiu et al., 2010
	Carbonate media			90%	COD/III /u	87%	$NH_3-N/m^3/d$	83%	2010
BAF	Lightweight ceramisite	Synthetic wastewater	2.5-5 h	-	234-780 mg/L COD	92%	58.6 mg/L NH4-N	> 62%	Liu et al., 2010
		Quality and star	1.51		116-245	0.40/	32 mg/L	01.50/	Feng et al.,
BAF	Grain siag	Grain slag Synthetic wastewater	1-3 11		mg/L COD	84%	NH3-N	91.5%	2012
Biofilm filter	Non-woven	Pond water	3 h	91-96%	-	Up to 90% COD	-	Up to 63.9%	Chang et al., 2010
Moving bed sequencing batch reactor	Polyurethane foam cubes	Low C/N wastewater	-	-	-	70%	-	100%	Lim et al., 2011
MDDD	Polyurethane foam	Synthetic westewater			200-401	90% TOC	49.8±8.5	42.6%	Chu and
MBBR Bio	Biodegradable polymer	gradable lymer	-	-	mg/L COD	72% TOC	mg TN/L	60.1%	Wang, 2011

 Table 2.3 Performance of different attached growth systems

MBBR	Ceramic	Oilfield produced water	36-10 h	-	343-365 mg/L COD	73%	41-48 mg/L	79-86%	Dong et al. 2011
MBBR	Polyethylene media	Coal gasification wastewater	48-32 h	-	1712-2340 mg/L COD	81%	NH3-N 182-259 mg/L NH3-N	93%	Li et al., 2011
Fixed bed sequencing batch reactor	Polypropylene carriers	Synthetic wastewater	4 h, 6 h, 8 h	-	0.5-1.5 kg COD/m³/d	95-96%	0.56-2.23 kg NH4-N /m³/d	70-88% TN 76-90% TP	Rahimi et al., 2011
Sequencing batch biofilter granular reactor	Wheel shaped plastic elements	Primary effluent	300 min	Up to 90%	2.5 kg COD/m³/d	80-90%	28-47 mg/L NH4-N	80-100%	Di laconi et al., 2010
Aerobic submerged	Lava rock	Grit tank effluent	3.96-8.31 h	-	0.45-3 kg COD/m <sup>3</sup> /d	80%	0.06±0.02 kg NH4-N/m³/d	20-90%	Morgan- Sagastume
filter					9.4 kg COD/m <sup>3</sup> /d	54%			and Noyola, 2008
			2.2 h		4.3±0.5 kg	87%	0.51 kg/m <sup>3</sup> /d N	84.5±1.3% N	
Fluidized bed bioreactor	Lava rock	Lava rock Municipal wastewater	2.3 11	86%	COD/m <sup>3</sup> /d		0.06 kg/m <sup>3</sup> /d P	55.0±8.0% P	Andalib et al., 2010
			9 h		528±88 mg/L COD	63±7.4%	8.6±2.5 mg/L PO4-P	7.3±13.6% PO4-P	

HBR	Polyurethane	Domestic wastewater	2.5 h	-	10.5 gCOD/m²/d	67±7.5%	0.23 g/m <sup>2</sup> /d	86.6±4% NH4-N	Tawfik and Klapwijk, 2010
Moving bed MBR	Nonwoven carriers	Synthetic wastewater	12h	-	400 mg/L COD	95.6%	20 mg/L NH <sub>4</sub> - N	91.8% NH4- N	Yang et al., 2009

#### 2.3.2 Media used in attached growth systems

When engineering biofilm to be utilized for wastewater treatment, it is of great importance to pay more attention to media development. The characteristics of the support media have significant impact on the attached biofilm structure in the reactor, the operational mode and the process effectiveness (Guo et al. 2012b). Therefore, the ideal media not only should provide a large surface area to maximize microbial attachment and growth, and sufficient void space for air diffusion as well as allow sloughed microbial biofilm passing through, but also should not be toxic to microorganisms and be stable both chemically and mechanically (Grady and Lim, 1980).

The design of biofilm carrier is important due to requirements for good mass transfer and nutrients to microorganisms. The key parameters of the biofilm carriers are its shape and the percentage of the tank filled with it (Robescu et al., 2009). For the effective growth of biofilm and its performance in a reactor we need to take special care while we design the specific surface area of the carrier and the filling fraction of the carrier in the reactor (Odegaard et al., 2000). The specific surface area of the carrier reflects the amount of surface area available for biofilm development per unit volume of the carrier on a bulk volume basis. The reactor specific surface area equals the specific surface area of the carrier multiplied by the fraction of the total reactor volume that the carrier occupies (bulk volume basis) (Weiss et al., 2005). The attachment of microorganism to the surface and the subsequent growth of the biofilm community depend upon the surface of the biofilm carriers that are rougher, more hydrophobic, and coated with surface-conditioning films (Vayenas, 2011). Sponge has been considered as a reasonable attached growth media because it can act as a mobile carrier for active biomass resulting in improved organic and nutrient removal as well as reduces fouling of the membrane by reducing the cake layers formed on the surface of the membrane and retain microorganisms by incorporating a hybrid growth system (Guo et al., 2009; Ngo et al., 2006).

Table 2.4 summarized the amount of biomass content that attached-growth on various types of support media.

		Struct	ural characte	eristics				
Media type		Media size	Specific Density surface area		The amount of attached growth biomass	Application	References	
Powdered	Bentonite	29.72 µm	2.48 g/cm <sup>3</sup>	118.94 m²/g	570-1180 mg MLVSS/L	Hybrid biological reactor treating	Lee et al 2002	
minerals	Clinoptilolite	27.55 µm	2.32 g/cm <sup>3</sup>	323.15 m²/g	4020-4620 mg MLVSS/L	synthetic wastewater	Lee et al., 2002	
polyuretl	nane forms	10×10×10 mm	28-30 kg/m <sup>3</sup>	-	2.58–3.34 g/L	Hybrid biological reactor treating sewage	Wang et al., 2000	
Inner tube	Inner tube of used tyres		1.925±0.21 g/cm <sup>3</sup>	-	5400–6700 mg/reactor Moving biofilm-aerobic- sequencing batch reactor trea synthetic wastewater		Sirianuntapiboon and Yommee, 2006	
Polyethyle	ene granule	1-3 mm	0.9 g/cm <sup>3</sup>	-	< 2 g protein/L reactor	Hybrid membrane bioreactor treating tannery wastewater	Artiga et al., 2005	
Kaldnes K3 b	iofilm carriers	Diameter = 25 mm	0.95 g/cm <sup>3</sup>	500 m <sup>2</sup> /m <sup>3</sup>	2000 mg/L	Hybrid membrane bioreactor treating municipal wastewater	Liu et al., 2010	
Kaldnes K3 b	iofilm carriers	Diameter = 25 mm	0.95 g/cm <sup>3</sup>	500 m <sup>2</sup> /m <sup>3</sup>	0.03- 0.12 g-VSS/g- COD	Hybrid membrane bioreactor treating industrial wastewater	Artiga et al., 2008	
Kaldnes K3 biofilm carriers		Diameter = 25 mm	0.95 g/cm <sup>3</sup>	500 m <sup>2</sup> /m <sup>3</sup>	2.4 – 12.6 g/m <sup>2</sup>	Integrated fixed-film activated sludge treating municipal	Regmi et al., 2011	

 Table 2.4 Biomass content on various attached growth media

					wastewater	
Polyether foam cubes	-	-	-	< 0.85 g/g	Moving bed biofilm reactor treating synthetic wastewater	Marques et al., 2008
Polyurethane foam		84-117 kg/m <sup>3</sup>	$620 \ m^2/m^3$	< 2 g-VSS/g	Sequencing batch reactor treating synthetic wastewater	Moe and Irvine, 2000
Polyurethane sponge	-	-	-	19-24.6 g VSS/L sponge volume	Combination of UASB and DHS reactor treating sewage	Tandukar et al., 2006
Sponge strips	2.5×2.5×50 cm	-	-	< 26 g-VSS/L sponge volume	Combination of UASB and DHS reactor treating municipal wastewater	Tandukar et al., 2005
Polyurethane snonge	12×12×12 mm	_	_	22.000-26.000 mg/L	Moving bed biofilm reactor treating caprolactam wastewater	Chae et al., 2004
i oly al colland sponge	15×15×15 mm			,	Moving bed biofilm reactor treating synthetic wastewater	Chae et al., 2008
Biological powder activated carbon (BGAC)	-	-	-	3 g/L	-	Xing et al., 2008
Polyethylene bio-carrier	Diameter = 7 mm	0.97-0.98 g/cm <sup>3</sup>	900 m <sup>2</sup> /m <sup>3</sup>		Anaerobic-aerobic moving bed biofilm reactor treating landfill leachate	Chen et al., 2008
Full-scale disc		-		Thickness of $0.5 - 4.5$ mm	Rotating biological contractors treating municipal wastewater	Cortez et al., 2008
Polyvinyl chloride	Diameter	1004.2	30.7 m <sup>2</sup> /kg	400-1700 mg/L	suspended carrier biofilm reactor	Wang et al., 2005

cylindrical particles	= 2.5 mm	kg/m <sup>3</sup>			treating synthetic wastewater		
Nonwoven geotextiles baffles		-		1.4 – 3.4 g/baffle	Treating wastewater from a combined sewer	Korkut et al., 2006	
					system		
Polyurethane carrier	Diameter = 8-10 mm	0.3-0.5 kg/L	900 m <sup>2</sup> /m <sup>3</sup>	0.067 ± 0.006 g/sponge	Moving bed biofilm reactor treating synthetic wastewater with low C/N ratio	Chu and Wang, 2011	
	2×2×2 mm	-	$455\pm7$ m <sup>2</sup> /m <sup>3</sup>	49±9 g	49±9 g		
	3×3×3 mm	-	$451\pm7$ m <sup>2</sup> /m <sup>3</sup>	35±5 g	Moving bed sequencing batch	Lim et al., 2011	
Polyuretnane cubes	4×4×4 mm	-	438±7 m <sup>2</sup> /m <sup>3</sup>	33±4 g	wasterwater		
	5×5×5 mm	-	412±6 m <sup>2</sup> /m <sup>3</sup>	30±3 g			
Kaldnes K1 biofilm carrier	Diameter = 10 mm	0.95 g/cm <sup>3</sup>	$500 \text{ m}^2/\text{m}^3$	3234-6749 mg TS/L	Moving bed biofilm reactor treating municipal wastewater	Xiao and Ganczarczyk, 2005	
Hollow cylinder biofilm	Diameter	0.96-0.98	1200	0.7±0.2 g/L (AMBR)	Attached-growth MBR treating	Hu et al., 2012	
carrier	= 10 mm	g/cm <sup>3</sup>	m <sup>2</sup> /m <sup>3</sup>	0.8±0.2 g/L (AMBRb)	synthetic wastewater		

Plastic media is the most popular support carrier currently used in attached-growth system, the material of plastic media includes polycarbonate (PC), polyethylene (PE), polypropylene (pp), polytetrafluoroethylene (PTFE), polyvinyl chloride (PVC), acrylonitrile butadiene styrene (ABS), nylon (Ny) and tufnol (Tu) etc. The most widely reported plastic media for use at full-scale wastewater treatment facilities are Kaldnes media, while the PE media can be employed for simultaneous nitrification and denitrification (SND) (Guo et al., 2012b). Stephenson et al. (2013) tested 8 different plastic media in order to determine the relation between the media properties and the development of nitrifying biofilms in mixed cultures for wastewater treatment. Their result indicated that nitrifiers are better adapted to adhere to low-energy surfaces, or possibly the inability of low-energy surfaces to support greater biomass associated with the rapid growth of heterotrophic biofilm (Table 2.5). These findings have significant implications for media selection in wastewater treatment.

**Table 2.5** Data of 8 different plastic media (adapted from "Media surface properties and the development of nitrifying biofilms in mixed cultures for wastewater treatment" by Stephenson et al., 2013, *Process Safety and Environmental Protection*, 91, 321-324)

Parameter	Materia	al						
	PC	ABS	PTFE	РР	Tu	Ny	PE	PVC
R <sub>a</sub> (nm)	6	34	162	25	62	107	603	75
R <sub>max</sub> (nm)	0.28	0.75	2.02	1.85	1.96	2.75	6.58	2.10
<b>SA (nm<sup>2</sup>)</b>	5639	5891	5839	5719	5735	5799	6322	5737
Surface adhesion force (nN)	11.7	nd	8.0	10.5	12.5	40.0	21.0	23.5
Dry Biomass (g/m <sup>2</sup> )								
Week 8	46.8	56.4	44.7	33.1	40.7	56.2	36.5	17.3
Week 10	35.2	51.9	24.7	42.2	42.2	50.5	40.6	22.0
Nitrification rate (g/m <sup>2</sup> d <sup>-1</sup> )								
Week 8	0.24	0.02	1.52	0.19	0.26	0.00	0.04	0.01
Week 10	0.39	0.36	0.32	0.17	0.49	0.07	0.15	0.06

 $R_{a}$ : average roughness;  $R_{max}$ : maximum range of the profile; SA: surface area; Nd: no data

According to Chu and Wang (2011), synthetic support media consisting of polyurethane foam are particularly suitable for microbial attachment and growth due to the presence of high specific surface area, huge void volume, and long service life. In addition, Ngo et al. (2008) demonstrated that reticulated polyurethane foam not only is ideal media for active biomass, but also reduce cake layers formed on the membrane surface and retain microorganisms by incorporating in a SSMBR. Apartfrom effective removal for nitrogen and phosphorus, the requirement for removing micropollutants has become the interest of the future research and development. Luo et al. (2014a) investigated the effectiveness of a sponge-based MBBR for removing five groups of micropollutants. They found that the MBBR achieved varying removals for the selected micropollutants due to their diverse physicochemical properties and the acclimatized sponge improved the removal of some less hydrophobic (logD < 2.5) compounds.

Other than the media material mentioned above, biological activated carbon (BAC) and nonwoven (NW) fabric are also used in attached-growth system as carrier material. Generally, BAC can act as an absorbent and show strong affinity for attaching organic substance, while NW offers a large surface-to-volume ratio for biological attachment. However, NW may be a more attractive option due to its low capital cost (Roy et al., 1998; Turbak, 1993).

# 2.4 Moving bed biofilm reactors (MBBRs)

The MBBR was developed in order to adopt the best features of the AS process and those of the biofilm process without being restrained by their drawbacks (Ødegaard et al., 1994). The idea was basically achieved by allowing the biomass grows on small media that freely move inside a reactor. The reactor can be used for aerobic, anoxic or anaerobic processes (Ødegaard et al., 1994).

A MBBR consist of a tank equipped with an outlet sieve to retain the media, the media itself, and a means of aeration or mixing. Aeration is by coarse bubble, using stainless steel laterals. Within the reactor, the media, effluent and air are completely mixed resulting in very efficient contact between the biofilm and substrates within

the liquid. One of the important features of this process is that biofilm thickness is controlled by the movement of the carriers so the oxygen diffusion through the biofilm is encouraged. Detached biofilm is suspended with the reactor and leaves the reactor with the effluent. Mechanical mixing is required in anoxic and anaerobic reactors, usually provided by slow speed submersible mixers (Ødegaard et al., 2004).

The MBBR is an effective, affordable attached growth biological process and has been attracted increasing attention over the last decade in wastewater treatment and reuse field. The merits of MBBR systems are compact space requirement, low head loss, no filter-bed channeling, no need for periodic backwashing and sludge recycle, and large surface for colonization and high specific biomass activity (Guo et al, 2010; Lee et al, 2006; Ødegaard et al., 2006). Additionally, MBBR may use almost any reactor shape or may choose various operating loading rates in a given reactor volume, due to the fact that the choice of carrier filling fraction is subjected to preference (Ødegaard et al., 2004).

## 2.4.1 Media material

Inside the Kaldnes MBBR, polyethylene carriers with various shape and size can be used to offer the treating flexibility depending on wastewater characteristics, pretreatment, discharge standards and available reactor volume (Ødegaard et al., 1994, 1999; Rasmussen, 2011). Other than polyethylene, a wide range of biocarriers, including polyurethane (PU), granular activated carbon, sand and diatomaceous earth, have been utilized in MBBR system (Chu and Wang, 2011). In general, polymer carriers have low density and excellent processability, and expansion can be obtained easily as the water circulates. However, the poor hydrophilicity and biocompatibility of plastic carriers often cause some deficiencies in the rate and amount of biofilm culturing, and the adhesion extent of biofilm (Dong et al., 2011). In addition, biomass buildup and reactor head-loss may occur due to rapid clogging and jamming of the plastic carriers below the upper grid, including the upper retention grid to break (Dupla et al., 2006). In terms of inorganic carriers, such as limestone, zeolite, activated carbon, graystone, slag, and coke, they have good mechanical strength and biocompatibility. However, relatively high density of inorganic carriers can increase the energy requirement for expansion. Ideally, PU is

the perfect growth medium with high porosity for microorganism immobilization, good mechanical strength and low cost (Chea et al., 2008; Chu and Wang et al., 2011; Kim et al., 2009).

Because the properties of media have effects on the ability to form biofilms, the quality of biomass and the efficiency of treatment (Dong et al, 2011), many efforts have been put on the development and application of novel carriers recently. Chen et al. (2007) used tube chip shaped biocarriers, which consist of organic polymer mixed with nano-sized inorganic ingredients material, in a MBBR combined with Fentoncoagulation pretreatment to treat pesticide wastewater. The results supported that the MBBR has excellent advantages such as flexibility, easy operation and strong resistance against loading impact, due to high biomass attached on the surface of the biocarriers and high biofilm activity. Delnavaz et al. (2010) investigated a MBBR filled with light expended clay aggregate (LECA) for a toxic and hard biodegradable aniline removal using artificial neural network (ANN) model. The results showed that up to maximum of 90% removal efficiencies were gained for COD of 2000 mg/L after 3 days. The PU foam (sponge) was applied as an active mobile carrier in aerobic moving bed bioreactor by Guo et al. (2010) for treating a high strength synthetic wastewater. The results showed outstanding ammonium (100% at filtration flux of 10 and 15 L/m<sup>2</sup>h) and phosphorus (> 91% at all fluxes range) removal with optimum pH range of 6 - 7. Chu and Wang (2011) presented a comparison between inert PU foam and biodegradable polyer polycaprolactone (PCL) particles as carriers in MBBR system for the removal of organics and nitrogen from wastewater with low C/N ratio. Their results demonstrated that total organic carbon and ammonium removal efficiency were 95% and 65% in the reactor filled with PU carriers, compared with 72% and 56% in the one filled with PCL carriers at an hydraulic detention time of 14 h. The good performance of the reactor with PU media was because the fact that numerous microorganisms were entrapped on the pores of the PU carriers so that enhanced the nitrifiers to inhabit. On the other hand, reactor with PCL media showed good behavior in terms of TN removal as the biodegradable polymer was an effective substrate for denitrification. Dong et al. (2011) utilized both unmodified and sepiolite-modified suspended ceramic carriers to feed two MBBRs with a filling fraction of 50% to treat oilfield produced water. The results indicated that the suspended ceramic carrier was an excellent MBBR carrier. Particularly, the modification of ceramic biocarrier with sepiolite produced better outcomes in removal efficiencies of chemical oxygen demand, ammonia nitrogen and polycyclic aromatic hydrocarbons. The feasibility of using bioplastic-based carrier for removal of three selected xenobiotic from synthetic wastewater was evaluated in Accinelli et al. (2012) study. Their result suggested that the concept behind the MBBR technology can be extended to biodegradable carriers inoculated with bioremediation microorganisms.

## 2.4.2 Media geometry and size

In MBBR, the geometry and sizing of specially designed biofilm medium have been considered to optimize performance. Ødegaard et al. (2000) investigated how carrier size and shape influenced the performance of MBBR system. Three different types of carriers (Kaldnes, AWT and ANOX) were used and the MBBR was operated at relatively high organic loads  $(10 - 120 \text{ g COD/m}^2\text{d} \text{ and } 5 - 45 \text{ s SCOD/m}^2\text{d})$ . The results demonstrated that once the effective surface area had been established, the difference of shape and size of the carriers had minor effect on removal efficiency. However, smaller carriers required less bioreactor volume than larger ones, in terms of given organic effective surface area load. Levstek and Plazl (2009) studied the influence of carrier type on nitrification in moving bed biofilm process. Two types of carriers were utilized: a cylindrical high-density polyethylene ring shaped carrier (K1) and a spherical polyvinyl alcohol (PAV) gel bead shaped carrier (PVA-gel). Their results supported the conclusion that the effective surface area was the key factor for MBBR design. Therefore, a 9.7 % of PVE-gel filling performed equally well as a 37% of K1 filling in a bioreactor as far as nitrification is concerned. Chai et al. (2013) investigated treatment performance of winery wastewater in two MBBRs (R9 and R30) with low-density polyethylene carriers differing in size, shape, structure, and specific surface area. The results implied that the performance of the anaerobic MBBR was enhanced by an increase in the specific area of the carrier used. In conclusion, the shape and size of the MBBR carriers can affect the specific surface area on which grows biofilm, resulting in influencing the treatment performance.

## 2.4.3 Media filling ratio

In most MBBR systems, treatment performance highly depends on the surface area available for biofilm growth, which accordingly related to filling ratios of the carrier elements. Regards to recommended filling fraction (carrier volume versus total bioreactor volume), it varies in different literature. Ødegaard et al. (1994, 2004) recommended that a filling fraction could be 67% (but maximum 70%) for cylindrical plastic carriers to enable smooth carrier suspension movement inside MBBR. Andreottola et al. (2000) occupied the bioreactor tank volume with 70% of elements for treating municipal wastewater. The average efficiency for totCOD removal was 76% and average ammonium removal efficiency was 92%. Canziani et al. (2006) used MBBR for enhancing denitrification in treatment of leachate from Italian landfills ranges from 0.5 to as high as 3 g/L. The K1 plastic carriers were filled with 37.5% of the denitrification tank volume. Both Di Trapani et al. (2008a) and Mannina and Viviani (2009) investigated the organic matter and nutrient removal performance of hybrid MBBR at 35% and 66% filling ratio and a very high efficiency was obtained from the results. In Falletti and Conte (2007) pilot-scale studies, it was reported that 80% of ammonium was nitrified with 43% of tank volume occupied by carriers while about 86% of ammonium was removed with 60% of tank volume occupied by carriers. Guo et al. (2010) applied 20% volume of sponge in a MBBR coupled with submerged membrane bioreactor. The results showed outstanding ammonium and phosphorus removal efficiency, which were 100% at filtration flux of 10 and 15 L/m<sup>2</sup>h and >91% at all fluxes range respectively. Levstek and Plazl (2009) used 37% filling fraction for K1 and 9.6% for PVA-gel carriers for evaluation of nitrification potential in MBBR system. Feng et al. (2012) investigated the effects of packing rates (29%, 30%, and 40%) of PU foam on removal of organics and nitrogen in MBBR. Their results indicated that filling ratio of PU foam carriers had little influence on COD removal but affected the ammonium removal efficiency, presumably due to the different relative abundances of nitrifying bacteria. Consequently, filling ratios are subject to specific surface area of a particular carrier element, which may considerably vary due to its physical characteristics.

## 2.4.4 Other impact factor

The optimal range of temperature for microorganism activity is between 25 to 35°C. The MBBR usually operate around  $20 - 25^{\circ}$ C. The temperature in a MBBR often relate to feedwater and the ambient temperature, while the temperature of media depend upon the material. The temperature of surface layer of media could be different from that of the core of the media (Loupasaki and Diamadopoulos, 2012). The change of temperature has effects on the system performance, such as nitrification rate. Chu and Chen (2002) conducted experiment on the impact of temperature on nitrification rate and the results indicated that the effect of temperature on nitrification was less significant than modeling prediction. In Zhang et al. (2013) study, the dependence of nitrification kinetics on temperature in MBBR systems treating low ammonia polluted raw water was investigated. The impacts of temperature on NH<sub>4</sub><sup>+</sup>-N and NO<sub>2</sub><sup>-</sup>-N oxidation kinetics could be described by the temperature coefficients of 1.099 and 1.098, respectively, after exclusions of the inhibitory impacts of temperatures below 5.0 °C and the variations in nitrifying community structure. They believed that these temperature coefficients should be considered when designing and operating such MBBRs.

### 2.4.5 Micropollutant removal in MBBR processes

In spite of the extensive investigations on the MBBR treatment of traditional contaminants (COD and nutrients), little attempt has been made to evaluate the micropollutant removal in MBBRs. Nevertheless, the results from some bench-scale studies demonstrated that MBBR technology is a promising treatment method for eliminating micropollutants from wastewater.

Luo et al. (2014a) investigated the removal of micropollutants using polyurethane sponge as attached-growth carrier. Batch experiments demonstrated that micropollutants could adsorb to non-acclimatized sponge cubes to varying extents. Acclimatized sponge showed significantly enhanced removal of some less hydrophobic compounds (logD < 2.5), such as ibuprofen, acetaminophen, naproxen, and estriol, as compared with non-acclimatized sponge. The results for bench-scale sponge-based MBBR system elucidated compound-specific variation in removal,
ranging from 25.9% (carbamazepine) to 96.8% (b-Estradiol 17-acetate) on average. In the MBBR system, biodegradation served as a major removal pathway for most compounds. However, sorption to sludge phase was also a notable removal mechanism of some persistent micropollutants. Particularly, carbamazepine, ketoprofen and pentachlorophenol were found at high concentrations (7.87, 6.05 and 5.55 µg/g, respectively) on suspended biosolids. As a whole, the effectiveness of MBBR for micropollutant removal was comparable with those of activated sludge processes and MBRs. Dvorák et al. (2014) used a full-scale MBBR to remove aniline, cyanides and diphenylguanidine from industrial wastewater. Long-term (5-years) MBBR operation has demonstrated that, following initial stabilisation and implementation of additional pretreatment, the system is capable of treating such hardly biodegradable industrial wastewater with high removal efficiency, with mean cyanide removal efficiency ranging from 75% to 99%. Aniline removal efficiency also reached more than 85%, while diphenylguanidine, phenylurea and N,Ndiphenylurea removal was almost quantitative. The results given in this study indicate that MBBR technology is promising option for treatment of poorly biodegradable and toxic industrial wastewater, furthermore with very high removal efficiency.

### 2.5 Attached growth based membrane hybrid system

Membrane technology has been rapidly advanced around the world and membrane bioreactors (MBRs) are being increasingly applied for municipal and industrial wastewater treatment. The membrane separation technique provides a solution to non-settling sludge, therefore, it can be used to replaced secondary clarifier and to obtain high effluent quality and compactness of treatment plants (Moore et al., 2006). Despite the benefits of MBR, the wide implementation of MBR technology still faces some challenges, such as issues related to membrane fouling. The sludge characteristics in MBR are one of the vital factors to membrane fouling. In terms of attached-growth biofilm media, they can not only reduce the suspended solid in the reactor, but also can affect the characteristics of the sludge, thereby mitigating membrane fouling (Hu et al., 2012). Hence, membrane hybrid systems that combine

attached-growth processes with membrane filtration are able to decrease membrane fouling to a great extent (Guo et al., 2012b).

#### 2.5.1 Membrane fouling

Membrane fouling remains the main concern in the application of MBRs, resulting in reduced filtration performance, shortened life of membrane and higher operating cost (Rodríguez-Hernández et al., 2013). Fouling of membranes is caused by complex physical and chemical interactions between the various fouling constituents in the feed and between these constituents and the membrane surface. Mass transport can lead to the attachment, accumulation, or adsorption of materials onto membrane surfaces and/or within membrane pores (Guo et al., 2012a). For a given membrane, fouling is directly related to sludge characteristics (Meng et al., 2009). However, the fouling behavior in MBRs is complicated, considering the complex nature of the activated sludge. The use of support media shows a potential in controlling and minimizing membrane fouling, because the media can be effectively remove organic impurities and turbidity.

Previous work done by Basu and Huck (2005) demonstrated the impact of high density polyethylene media on an integrated biofilter-membrane system treating difficult-to-treat surface water. They found that fouling rate in a biofilter-membrane process with 40% fill fractions of high density polyethylene media was at least two times slower than the same system without support media. Guo et al. (2009) conducted an experiment to examine the feasibility of using nonwoven fabric material as bedding material in a biofilter-submerged membrane absorption hybrid system. Their results showed that the system could eliminate dissolved organic pollutants ranging from 90 to 4200 Daltons and EBCT applied to the system was the main factor governing the membrane fouling during high rate operation. Also, other researchers have used mechanical cleaning of membranes by introducing granular material into the submerged membrane reactor to assist with fouling control (Siembida et al., 2010; Johir et al., 2011). The granular medium resulted in the enhancement of scouring of the membrane surface. These results indicate that introducing support materials has a potentially significant role in controlling fouling. Hu et al. (2012) investigated the effect of media on membrane fouling in 3 reactors and the results showed that membrane fouling was considerably mitigated by the carriers' biochemical effects on sludge characteristics rather than by the carriers' physical effects. Alresheedi and Basu (2013) examined the impact of high-density polyethylene (HDPE) media on a submerged hollow fiber PVDF ultrafiltration membrane system filtering synthetic water with different organic and inorganic matter content using a factorial design of experiments. The addition of support media into the membrane reactor resulted in a 35–50% reduction in membrane fouling compared to the control system with no media. Deng et al. (2014) also compared membrane fouling in a sponge-submerged bioreactor and a conventional membrane bioreactor based on sludge properties. They found that sponge addition could mitigate membrane fouling significantly by preventing pore blocking and reducing cake layer formation.

#### 2.5.2 Enhanced hybrid membrane system with attached growth media

In hybrid MBRs application, the attached-growth media play a significant role in enhancement of system performance by mitigating membrane fouling and retaining membrane permeability. In addition, the characteristics of the attached-growth media enable a large amount of biomass remain in the bioreactor, attributing to the high interaction between attached-growth biofilm and suspended solid. Among all the media, BAC, plastic media and sponge have been commonly used to improve the performance of SMBR.

Table 2.6 listed some information regarding the performance of enhanced membrane hybrid systems by adding biofilm support media.

Table 2.6 Performance of	f en	hanced	mem	brane	hy	brid	systems
--------------------------	------	--------	-----	-------	----	------	---------

	Attached-growth media material	Hybrid system	Raw Water	Treatment Performance	References
	BPAC (5 g/L)	SMBR (MF with pore size of 0.1 µm)	Synthetic biologically treated effluent	<ul> <li>The average TOC removal efficiency was up to 86% and the TOC removal efficiency was maintained over 85% even after 55-day operation.</li> <li>The hybrid system was able to remove effectively the small- and large-molecular weight organic matters from 270 to 36,270 Da.</li> <li>The TMP development during 55 days was only 49 kPa, as the PAC replacement enhanced both biological activity and adsorption which helped reducing membrane fouling.</li> </ul>	Guo et al., 2006
2-40	BPAC (1.5 and 3 g/L)	SMBR (MF with pore size of 0.4 µm)	Tannery industrial wastewater	<ul> <li>COD removal stability appeared to increase as PAC concentration increased.</li> <li>The fouling rate decreased with an increasing PAC concentration and showed complete reversibility both in presence and in absence of PAC.</li> <li>PAC dosing enhances typical MBR system stability in terms of effluent quality, allowing a better control of the operational criticisms related to the fouling rate, fouling reversibility and membrane life cycle.</li> </ul>	Munz et al., 2007
	PAC (2 g/L)	MBR (MF with pore size of 0.22 µm)	Micro-polluted lake water	<ul> <li>Adding PAC helped to improve the quality of the treated water, with effective removal of ammonia nitrogen above 80%.</li> <li>The addition of PAC enhanced the microbial activity in the bioreactor, contributed to the increase of the removal efficiency of organic pollutants</li> <li>The experiment proved that PAC addition relieves membrane fouling</li> </ul>	Sagbo et al., 2008
	PAC (2 g/L)	$\begin{array}{l} MBR \mbox{ (submerged} \\ 30 \ \mu m \ nylon \ mesh \\ filter \ with \ 0.05 \ m^2 \\ filtration \ area) \end{array}$	Distillery effluent	<ul> <li>PAC enhances filtration performance in terms of increasing the critical flux and extending the operation period without mesh cleaning.</li> <li>For a given OLR, the COD removal is higher with PAC addition.</li> </ul>	Satyawali and Balakrishnan, 2009

PAC (1.5 g/L)	SMBR (PVC UF with pore size of 0.01 µm)	Synthetic polluted raw water	<ul> <li>PAC supplementation results in adsorption of low molecular weight components (&lt;6000 Da) but PAC dosage has a major impact on the extent of adsorption.</li> <li>Due to the pre-treatment of BAC, the membrane fouling of sMBR in the hybrid process was substantially mitigated; and less frequent membrane cleaning was required.</li> <li>The results confirmed the synergetic effects between the BAC and the subsequent sMBR.</li> </ul>	Tian et al., 2009
PAC (750 mg/L)	SMBR (PVDF flat sheet membrane with 0.4 m <sup>2</sup> filtration area)	municipal secondary effluent	<ul> <li>Approximately 63%TOC, 95% NH4<sup>+</sup>-N and 98% turbidity in secondary effluent were removed by the PAC-MBR process.</li> <li>The addition of PAC significantly increased organic removal and responsible for the largest fraction of organic removal.</li> <li>Membrane fouling analysis showed the enhanced membrane performance in terms of sustainable operational time and filtration resistances by PAC addition.</li> </ul>	Lin et al., 2011
Kaldnes K1	MBBR-MBR (tubular zirconium ceramic with pore size of 50 nm)	old landfill leachate	<ul> <li>Compared to conventional nitrification, the process tested during this experimentation (ammonia oxidation stopped to nitrite) allows up to 25% savings in oxygen demand and up to 40% in COD demand for denitrification.</li> <li>When DO concentration in the MBR was kept in the range 0.2–0.5 mg L-1, 90% oxidation of ammonia to nitrite was achieved, with stable inhibition of nitrite-oxidizing bacteria even at sludge retention time higher than 45 days.</li> </ul>	Canziani et al., 2005
Polyurethane sponge	SMBR (PVDF with pore size of 0.2 µm)	Synthetic wastewater	<ul> <li>Sponge addition could mitigate membrane fouling significantly by preventing pore blocking and reducing cake layer formation.</li> <li>High NH<sub>4</sub>-N removal could be attributed to the enhanced population of ammonium oxidation bacteria on the acclimatised sponge.</li> <li>As sponge could provide the anoxic condition around the surface of the sponge and the anaerobic condition inside the sponge, the system achieved high removal efficiency of PO<sub>4</sub>-P.</li> </ul>	Deng et al., 2014

#### 2.5.3 Micropollutant removal in attached-growth based hybrid membrane system

The membrane bioreactor (MBR) has been considered as a reliable and effective alternative to conventional activated sludge (AS) processes in terms of pollutants removal. In addition to excellent removal of traditional contaminants (organic matters and nutrients), MBRs are also able to considerably eliminate a broad range of micropollutants, some of which are impervious to the conventional activate sludge treatment. This is due to the facts that 1) MBRs are able to retain sludge that can adsorb micropollutants (Abegglen et al., 2009); 2) The membrane surface can also reject the release of micropollutants (Kimure et al., 2003; Kimure et al., 2004; Radjenović et al., 2008); and 3) The prolonged SRT in MBRs can enhance the removal efficiencies of some micropollutants (Kimure et al., 2007; Luo et al., 2014b). The superiority of MBRs over conventional activated sludge treatment in terms of micropollutant removal has been well documented in literature. Hai et al. (2011) showed that an exceptionally high removal ( $68 \pm 10$  %) of carbamazepine was achieved by MBR under near-anoxic condition. Radjenović et al. (2009) reported that MBR technology could significantly improve the elimination of some pharmaceuticals by 30 - 65%. Chen et al. (2008) found that a submerged MBR was able to achieve much more stable removal of bisphenol A at varied volumetric loadings, as compared to a conventional activated sludge reactor.

In spite of the advantages, the implementation of MBRs has long been limited by a high whole-life cost, due to the substantial energy consumption and constant maintenance. Moreover, the membrane fouling is another remarkable obstacle to the widespread application of MBR technology. Strategies to prevent fouling before its occurrence are of great significance and many efforts have been made to achieve the goal (Fang et al., 2006; Guo et al., 2004; Liao et al., 2004; Yang et al., 2006;) by 1) improving the anti-fouling properties of the membrane, 2) operating the MBR under specific non-or-little-fouling conditions, 3) pretreating the biomass suspension to limit its fouling propensity (Le-Clech et al. 2006).

On the other hand, moving bed biofilm reactor (MBBR) is a compact system with high concentrations of attached-growth biomass but significantly low suspended solids production. By integrating MBBR with a MBR, the considerably low solid concentrations contribute to reduced fouling potential. Furthermore, readily biodegradable chemical oxygen demand (COD) can be removed in the MBBR and hence biological activity on the membrane surface will be minimized (Pervissian et al., 2011). Accordingly, the fouling tendency of a MBR could be alleviated by using MBBR as a pretreatment. Among all suspended media used in MBBR, the superiority of polyurethane foam (sponge) has been proven in terms of removal of both conventional and unconventional pollutants (Chu and Wang, 2011; Luo et al., 2014b).

Luo et al. (2015) compared a moving bed biofilm reactor-submerged membrane bioreactor (MBBR-MBR) hybrid system and a conventional membrane bioreactor (CMBR) in terms of micropollutant removal efficiency and membrane fouling propensity. The results show that the MBBR-MBR hybrid system could effectively remove most of the selected micropollutants. By contrast, the CMBR system showed lower removal of some micropollutants including ketoprofen, carbamazepine, primidone, bisphenol A and estriol. Mass balance calculations suggest that biological degradation was the primary removal mechanism in the MBBR-MBR system. During operation, the MBBR-MBR system exhibited significantly slower fouling development as compared to the CMBR system, which could be ascribed to the wide disparity between the SMP levels of the two systems. It is evident that adding a MBBR process prior to MBR treatment can not only enhance micropollutant elimination but also mitigate membrane fouling.

## **2.6 Conclusion**

Moving bed biofilm reactor (MBBR) is a compact system with high concentrations of attached growth biomass but significantly low suspended solids production. In MBBR, attached growth media moves freely in the aeration reactor and enables better oxygen transfer with high nitrification rate. Moreover, enhanced process stability can be achieved as most biomass is attached onto the media rather than being washed away. Compared to activated sludge (AS) processes, the operation of MBBR requires less energy consumption and low maintenance cost. Thus, MBBR delivers a flexible, cost-effective, and easy-to-operate mean to address current wastewater issues and the expandability to meet future loads or more stringent discharge requirements. In addition, the MBBR is able to remove a board spectrum of micropollutants as it is more likely to have different redox conditions within biofilm.

By integrating with membrane bioreactor (MBR), MBBR also offers a promising alternative to activated sludge based technology. It has been reported that high mixed liquid suspended solid (MLSS) concentration caused high suspension viscosity in membrane unit. It might be expected to have a profound influence on membrane performance due to its effect on both the dynamic cake layer thickness and the viscosity. Considering the relatively low suspended solid concentration in MBBR, this unique characteristic contributes to low viscosity of suspended solid and reduced fouling potential in membrane unit. In this context, the combination of MBBR and membrane filtration could sufficiently reduce MLSS content in the whole process, resulting in a different composition and properties of the membrane feed solution.

Although there are limited amount of studies on the performance of MBBR-MBR hybrid system in previous research, the results from the available reports illustrate a potential for improved membrane fouling control and mitigation due to a decreased suspended solid environment in the process, regardless system configurations and the choices of biocarriers. In addition, the performance of MBBR-MBR was comparable and competitive with that of the activated sludge (AS)-MBR in terms of removal of COD and ammonium. However, no effort has been made to examine the effectiveness of micropollutants removal by using a MBBR-MBR system so far. Therefore, micropollutant removal by MBBR-MBR is a strategy showing excellent possibilities and is likely to be the subject of considerable future research.



University of Technology, Sydney

Faculty of Engineering and Information Technology

# Chapter 3

# **Experimental Investigation**

#### 3.1 Introduction

This chapter describes the materials, experimental setups and analytical methods used for the MBBR system, the MBBR-SMBR hybrid system and the SSMBR system. A continuous bench-scale MBBR was set up for a long-term assessment of micropollutant removal at HRT of 24 h, 18 h, 12 h, and 6 h. Additionally, an MBBR-MBR system was investigated in terms of its micropollutant removal efficiency and membrane fouling at HRT of 24 h, 18 h, 12 h, and 6 h. Furthermore, a SSMBR was also studied in order to compare the treatment efficiency to the MBBR-MBR system.

#### 3.2 Materials

#### 3.2.1 Synthetic wastewater

In order to maintain constant feed pollutant concentration, a synthetic wastewater (simulating medium strength municipal wastewater primary effluent) spiked with micropollutants was used in this study. The chemical composition of synthetic wastewater is shown in Table 3.1. The stock solution was stored in a refrigerator at 4 °C. The synthetic wastewater was produced by diluting the stock solution with tap water on a daily basis and was fed continuously and evenly to the treatment system to avoid any fluctuation in the feed concentration and provide a sufficient source of biodegradable organic pollutants such as glucose, ammonium sulphate and potassium dihydrogen orthophosphate. The synthetic wastewater contains chemical oxygen demand (COD) of 320–360 mg/L, total organic carbon (TOC) of 100–120 mg/L, NH<sub>4</sub>-N of 13–16 mg/L, NO<sub>2</sub>-N of 0–0.02 mg/L, NO<sub>3</sub>-N of 0.4–1.1 mg/L and PO<sub>4</sub>-P of 3.0–3.5 mg/L. NaHCO<sub>3</sub> or H<sub>2</sub>SO<sub>4</sub> was used to adjust the pH to a constant value of 7.

#### 3.2.2 Selected micropollutants

A set of 22 micropollutants that have been frequently detected in municipal wastewater were selected for investigation (Table 3.2, Nguyen et al., 2012) to represent five important groups of micropollutants, namely pharmaceuticals and personal care products (PPCPs), pesticides, hormones and industrial chemicals. A concentrated stock solution containing 100 mg/L of each micropollutant was

prepared in pure methanol and kept in a freezer. The stock solution was added to obtain an initial concentration of 5  $\mu$ g/L for each compound in the feed wastewater.

Compound	Chemical formula	Molecular weight (g/mol)	Concentration (mg/L)
Organics and nutrients			
Glucose	$C_6H_{12}O_6$	180.0	280
Ammonium sulfate	$(NH_4)_2SO_4$	132.1	142
Potassium phosphate	KH <sub>2</sub> PO <sub>4</sub>	136.1	26
Trace nutrients			
Calcium chloride	$CaCl_2 \cdot 2H_2O$	147.0	0.368
Magnesium sulfate	MgSO <sub>4</sub> ·7H <sub>2</sub> O	246.5	5.07
Manganese chloride	$MnCl_2 \cdot 4H_2O$	197.9	0.275
Zinc sulfate	$ZnSO_4 \cdot 7H_2O$	287.5	0.44
Ferric chloride anhydrous	FeCl <sub>3</sub>	162.2	1.45
Cupric sulfate	$CuSO_4 \cdot 5H_2O$	249.7	0.391
Cobalt chloride	CoCl <sub>2</sub> ·6H <sub>2</sub> O	237.9	0.42
Sodium molybdate dihydrate	Na <sub>2</sub> MoO <sub>4</sub> ·2H <sub>2</sub> O	242.0	1.26
Yeast extract			30

Table 3.1 Composition and concentration of the stock solution

### 3.2.3 Sponge (polyurethane foam)

The biofilm carriers were made of reticulated sponge S28/80R (Joyce Foam Products), composed of skeletal strands and containing a homogenous structure of evenly sized air cells. The sponge has a density of 28–30 kg/m<sup>3</sup> with 80 cells per 25 mm. The sponge was cut into small sponge cubes ( $2 \text{ cm} \times 2 \text{ cm} \times 2 \text{ cm}$ , Figure 3.1). Before the experiments, the sponge was acclimatized using activated sludge fed with synthetic wastewater without the addition of micropollutants.



Figure 3.1 The attached-growth carriers (sponge cubes) used in this study

Category	Compound	CAS number	Molecular weight (g/mol)	Log K <sub>OW</sub> <sup>a</sup>	Log D (pH 7) a	Dissociation constant <sup>a</sup> ( pKa)	Water solubility (mg/L) <sup>b</sup>	Vapor pressure <sup>a</sup> (mm Hg), at 25°C	Chemical structure
Pharmaceuticals	Ibuprofen (C <sub>13</sub> H <sub>18</sub> O <sub>2)</sub>	15687-27-1	206.28	3.50 ± 0.23	0.94	4.41 ± 0.10	21	1.39E-4	CH <sub>3</sub> H <sub>3</sub> C OH
	Acetaminophen (C <sub>8</sub> H <sub>9</sub> NO <sub>2)</sub>	103-90-2	151.16	0.48 ± 0.21	0.47	$9.86 \pm 0.13$ $1.72 \pm 0.50$	14000	1.43E-6	HO CH3
	Naproxen (C <sub>14</sub> H <sub>14</sub> O <sub>3)</sub>	22204-53-1	230.26	2.88 ± 0.24	0.73	$4.84 \pm 0.30$	16	3.01E-7	H <sub>3</sub> C O OH
	Ketoprofen (C <sub>16</sub> H <sub>14</sub> O <sub>3)</sub>	22071-15-4	254.28	2.91 ± 0.33	0.19	4.23 ± 0.10	16	3.32E-8	О СН3 СООН

Т	Fable 3.2 Phy	vsicochemical p	properties of	f the selected	trace of	rgar	nics			
	Catagomi	Compound	CAS	Malaaular	LogV	а	LagD	Discoviation	Watar	

Diclofenac (C <sub>14</sub> H <sub>11</sub> Cl <sub>2</sub> NO <sub>2)</sub>	15307-86-5	296.15	$4.55 \pm 0.57$	1.77	$4.18 \pm 0.10$ -2.26 ± 0.50	2.4	1.59E-7	COOH NH Cl
Primidone (C <sub>12</sub> H <sub>14</sub> N <sub>2</sub> O <sub>2)</sub>	125-33-7	218.25	$0.83 \pm 0.50$	0.83	$12.26 \pm 0.40$ -1.07 $\pm 0.40$	500	6.08E-11	H <sub>3</sub> C NH
Carbamazepine (C <sub>15</sub> H <sub>12</sub> N <sub>2</sub> O)	298-46-4	236.27	$1.89 \pm 0.59$	1.89	$13.94 \pm 0.20$ -0.49 ± 0.20	18	5.78E-7	O NH2
Salicylic acid (C <sub>7</sub> H <sub>6</sub> O <sub>3)</sub>	69-72-7	138.12	$2.01 \pm 0.25$	-1.13	3.01 ± 0.10	2240	4.45E-5	ОН
Metronidazole (C <sub>6</sub> H <sub>9</sub> N <sub>3</sub> O <sub>3)</sub>	443-48-1	171.15	-0.14 ± 0.30	-0.14	$14.44 \pm 0.10$ $2.58 \pm 0.34$	9500	2.67E-7	HO O

	Gemifibrozil (C <sub>15</sub> H <sub>22</sub> O <sub>3)</sub>	25812-30-0	250.33	$4.30 \pm 0.32$	2.07	4.75	19	6.13E-7	CH <sub>3</sub> CH <sub>3</sub> CH <sub>3</sub> CH <sub>3</sub>
	Triclosan (C <sub>12</sub> H <sub>7</sub> Cl <sub>3</sub> O <sub>2)</sub>	3380-34-5	289.54	$5.34\pm0.79$	5.28	$7.80 \pm 0.35$	10	3.36E-5	CI OH
Pesticides	Fenoprop (C <sub>9</sub> H <sub>7</sub> Cl <sub>3</sub> O <sub>3)</sub>	93-72-1	269.51	$3.45 \pm 0.37$	- 0.13	2.93	71	2.13E-6	CI O CI CH <sub>3</sub>
	Pentachloro- phenol (C <sub>6</sub> HCl <sub>5</sub> O)	87-86-5	266.34	$5.12 \pm 0.36$	2.58	$4.68 \pm 0.33$	14	3.49E-4	$Cl \rightarrow Cl \rightarrow Cl$
Surfactants and industrial chemicals	4-tert- butylphenol (C <sub>10</sub> H <sub>14</sub> O)	98-54-4	150.22	$3.39 \pm 0.21$	3.40	10.13 ± 0.13	580	0.0361	HO CH <sub>3</sub> CH <sub>3</sub> CH <sub>3</sub>

	4-tert- octylphenol (C <sub>14</sub> H <sub>22</sub> O)	140-66-9	206.32	5.18 ± 0.20 5	.18	$10.15 \pm 0.15$	5	1.98E-3	HO CH <sub>3</sub> H <sub>3</sub> C CH <sub>3</sub> CH <sub>3</sub>
	4-n-nonylphenol (C <sub>15</sub> H <sub>24</sub> O)	104-40-5	220.35	6.14 ± 0.19 6.	.14	10.15	6.35	8.53E-5	HO CH
	Bisphenol A (C <sub>15</sub> H <sub>16</sub> O <sub>2)</sub>	80-05-7	228.29	3.64 ± 0.23 3.	.64	$10.29 \pm 0.10$	120	5.34E-7	HO CH <sub>3</sub> OH
Steroid hormones	Estrone (C <sub>18</sub> H <sub>22</sub> O <sub>2)</sub>	53-16-7	270.37	3.62 ± 0.37 3.	.62	$10.25 \pm 0.40$	677	1.54E-8	HO HO HO HO HO HO HO HO HO HO HO HO HO H
	17-β-estradiol (C <sub>18</sub> H <sub>24</sub> O <sub>2</sub> )	50-28-2	272.38	$4.15 \pm 0.26$ 4	.15	10.27	3.9	9.82E-9	ОН



<sup>a</sup> Source: SciFinder database <u>https://scifinder.cas.org/scifinder/view/scifinder/scifinderExplore.jsf</u>

<sup>b</sup> Source: <u>http://chem.sis.nlm.nih.gov/chemidplus/</u>

na: data not available

# 3.3 Experimental setup and operation protocol

## 3.3.1 MBBR system

A bench-scale MBBR system with a working volume of 40 L was used in the study (Figure 3.2). The reactor was filled with 20 % (determined from the batch experiments) of acclimatized sponge cubes. The HRT were 24 h, 18 h, 12 h, and 6 h, respectively. Accordingly, the reactor had a flow rate of 27.8 mL/min, 37.0 mL/min, 55.6 mL/min, and 111.1 mL/min. To avoid excessive detachment of the biosolids within the sponge cubes, the aeration of the MBBR was adjusted constantly to achieve gentle circulation of the sponge cubes. The DO concentration of the MBBR was controlled in the range between 5.5–6.5 mg/L. Before the experiment with addition of micropollutants, the MBBR system was acclimatized to the synthetic wastewater (without addition of micropollutants) for 20 days until TOC, TN, and PO<sub>4</sub>-P removal became stable. After the acclimatization stage, micropollutant-bearing wastewater was continuously introduced to the MBBR and the investigation of micropollutant removal was carried out over a period of 100 days for each HRT operation condition.





Figure 3.2 On-site photo and schematic diagram of the MBBR

# 3.3.2 MBBR-MBR system

The MBBR-MBR system (Figure 3.3) consisted of the MBBR and an MBR (working volume: 10 L). The membrane used for the MBR unit was a polyvinylidene fluoride (PVDF) hollow fiber module with a pore size of 0.2  $\mu$ m and surface area of 0.2 m<sup>2</sup>. The MBR unit was fed with MBBR effluent through a buffer tank. The MBR permeate flow was controlled by a suction pump in order to obtain a constant flux of 8.83 L/m<sup>2</sup>·h.





Figure 3.3 On-site photo and schematic diagram of the MBBR-MBR system

A pressure gauge was used to measure the transmembrane pressure (TMP). A soaker hose air diffuser was mounted at the bottom of the MBR unit to provide aeration (6 L/min). The MBR unit was operated in a continuous mode without backwash, relaxation or cleaning and the operation was terminated when the trans-membrane pressure (TMP) exceeded 35 kPa. Samples for the analyses of soluble microbial products (SMP) and extracellular polymeric substance (EPS), sludge hydrophobicity, zeta potential and particle distribution were collected at TMPs of 7, 10, 15, 18, 21, 24, 27, 30 and 35 kPa.

#### 3.3.3 SSMBR system

The SSMBR system (Figure 3.4) was a sponge-submerged membrane bioreactor. The reactor had a working volume of 10 L and was filled with sponge cubes (10%  $v_{sponge}/v_{reactor}$ ). The SSMBR used identical hydrophilic polyvinylidene fluoride (PVDF) hollow fibre microfiltration (MF) membrane modules with a pore size of 0.2  $\mu$ m and surface area of 0.2 m<sup>2</sup>. Polyurethane sponge cubes (S28/80R, Joyce Foam Products; dimension of 1 cm × 1 cm × 1 cm) were used as biofilm carriers. As no sludge withdrawal was performed except for removing sludge from carrier and mixed liquor for measurement, the SRT could be considered infinite. The SSMBR was operated at a constant flow rate of 27.8 mL/min, resulting in a HRT of 6 h. Accordingly, the permeate flux of was 8.34 L/(m<sup>2</sup>·h). The SSMBR was operated in a

continuous mode without backwash, relaxation or cleaning and the operation was terminated when the trans-membrane pressure (TMP) exceeded 35 kPa.



Figure 3.4 On-site photo of the SSMBR system

#### 3.3.4 Operation protocol

Every five days, influent and effluent aqueous samples were collected for COD, TOC, NH<sub>4</sub>-N, NO<sub>2</sub>-N, NO<sub>3</sub>-N and PO<sub>4</sub>-P analysis. In addition, the attached biomass on sponge cubes and suspended MLSS and MLVSS were also measure in both MBBR and MBR unit, as well as the SSMBR system.

Every ten days, 300 mL of influent and effluent aqueous samples were collected in duplicate for micropollutant analysis. Effluent samples were centrifuged at 3000 rpm for 30 min to improve filterability. The influent and centrifuged effluent samples were filtered with 1µm glass microfiber filter paper (47 mm DIA, Filtech) and acidified to pH 2 with 4 M HCl for subsequent solid phase extraction (SPE). Figure 3.5 showed the SPE process for GC-MS analysis of micropollutants. Oasis® HLB 6 cc cartridges (containing polymeric reversed-phase sorbent) were used for the SPE process (Figure 3.6). After SPE, the cartridges were kept in a freezer and sent to the

University of Wollongong for gas chromatography-mass spectrometry (GC-MS) analysis within one month.



Figure 3.5 Schematic diagram of SPE process for GC-MS analysis of micropollutants

To assess the extent of sorption of micropollutants on biosoilds (in suspension and on the sponge cubes), mixed liquor and sponge samples were withdrawn on Day 1, Day 50 and Day 100. The sludge within the sponge cubes was collected by squeezing the cubes. Subsequently, the micropollutants were extracted from sludge using a solvent extraction method previously described by Wijekoon et al. (2013, Figure 3.3). The sludge sample was initially centrifuged. The obtained pellet was freeze-dried for 4 h in an Alpha 1-2 LDplus Freeze Dryer (Christ GmbH, Germany). The dried sludge was ground to powder and 0.5 g of the dried sludge powder was transferred to a glass test tube. 5 mL of methanol was added to the test tube, followed by thorough mixing in a vortex mixer (VM1, Ratek, Australia) for 3 min and ultrasonic extraction for 10 min at 40 °C. The sample was then centrifuged at  $3270 \times g$  for 10 min (Alleegra X-12R, Beckman Coulter, USA) and the supernatant was stored in a glass beaker for subsequent analysis. Dichloromethane (5 mL) and methanol (5 mL) were added to the remained sludge. The above mentioned process of mixing, ultrasonic extraction and centrifugation was repeated. The supernatants from both steps were then mixed in a beaker. Milli-Q water was added into the beaker to fill up a solution volume of 50 mL. The residual methanol and dichloromethane were purged using nitrogen gas. Finally, Milli-Q water was filled in to obtain an aqueous sample of 500 mL.



Figure 3.6 Schematic diagram of solvent extraction process of sludge samples

# 3.4 Analytical methods

# 3.4.1 Organic matter, nutrients, pH and DO

TOC of the influent and effluent was measured using a TOC analyzer (Analytikjena Multi N/C 2000, Figure 3.5). The analysis of COD was carried out according to Standard Methods (APHA, 1998). NH<sub>4</sub>-N, NO<sub>2</sub>-N, NO<sub>3</sub>-N and PO<sub>4</sub>-P were measured by spectrophotometric method using Spectroquant Cell Test (NOVA 60, Merck, Figure 3.5). The pH and DO of the reactor were measured everyday using pH

meter (Hach Company, model no. HQ40d) and DO meter (Horiba Ltd. Japan, model no. OM -51E), respectively (Figure 3.7).



Figure 3.7 The analytical instruments used in this study

### 3.4.2 MLSS and MLVSS

The measurement of biosolids (monitored as mixed liquor suspended solids, MLSS) and biomass (monitored as mixed liquor volatile suspended solids, MLVSS) concentrations was conducted based on the method described in Standard Methods (APHA, 1998). A well-mixed sample was first filtered with 1µm glass microfiber filter paper (47 mm DIA, Filtech). The retained residue on the filter was dried in an oven at 105 °C for 2 h. The increment in weight of the filter paper indicates the total SS in the sample. Subsequently, the filter paper was ignited in a furnace at 550 °C for 20 min. The weight reduced during ignition represents the volatile SS in the sample.

The attached-growth biosolids was obtained by hand squeezing the sponge cubes and rinsing the squeezed cubes with Mill-Q water.

# 3.4.3 Micropollutants

Micropollutant concentrations in the aqueous samples were determined using an analytical method previously reported by Hai et al. (2011c). This method consists of a solid phase extraction procedure (SPE) followed by gas chromatography and quantitative determination by mass spectrometry with electron ionisation. TrOC concentrations in liquid samples (500 mL each) were extracted using 6 mL 200 mg Oasis HLB cartridges (Waters, Milford, MA, USA). First, the cartridges were preconditioned with 7 mL dichloromethane (DCM) and methanol (MeOH) mixture (1:1 v/v), 7 mL methanol followed by 7 mL reagent water (synthetic feed wastewater excluding TrOCs). The samples were acidified to pH 2-3 (4 M H2SO4) and loaded onto the cartridges at a flow rate of 1-5 mL/min. Then, the cartridges were rinsed with 20 mL Milli-Q water ( $6 \times 7mL$ ) and dried in a stream of nitrogen for 30 min.



Figure 3.8 Procedure used for SPE for analysing micropollutants

# 3.4.4 Fouling resistance

Fouling resistance of the MBR was determined after the MBBR-SMBR experiment by applying the resistance-in-series model described by Choo and Lee (2006):

$$\mathbf{J} = \Delta \mathbf{P} / \boldsymbol{\mu} \mathbf{R}_{\mathrm{T}} \tag{1}$$

$$R_{\rm T} = R_{\rm M} + R_{\rm C} + R_{\rm P} \tag{2}$$

Where J is the permeation flux;  $\Delta P$  is the transmembrane pressure;  $\mu$  is the dynamic viscosity of the permeate;  $R_T$  is total resistance;  $R_M$  is the intrinsic membrane resistance;  $R_C$  is the cake layer resistance; and  $R_P$  is the pore blocking resistance. To analyze each membrane resistance, a membrane cleaning process was carried out as follows:

1. To determine cake layer resistance: gently shake the membrane in a tank filled with distilled water until the cake layer on the membrane surface was completely removed. The decrease in membrane resistance before and after the physical cleaning represented the cake layer resistance.

2. To obtain pore blocking resistance: firstly clean the membrane with 0.5% citric acid for 6 hours to remove inorganic scaling deposit; then apply 6 hours' sodium hydroxide (0.4%) wash to eliminate organic substance; lastly immerse the membrane in 0.8 % sodium hypochlorite for 6 hours to destroy microorganisms. The pore blocking was determined by calculating the difference between the membrane resistances before and after the chemical cleaning.

The samples for SMP and EPS analysis were prepared as follows (Figure 3.7). A 30 mL of mixed liquor sample was drawn from the SMBR and immediately centrifuged at 3,000 rpm for 30 min. The supernatant and sludge pellet were collected separately. The supernatant was centrifuged again at 3000 rpm for 30 min, followed by a filtration process through 0.45  $\mu$ m of Whatman 934-AH glass fiber filter (to ensure the particles and bound EPS were removed). The filtered sample was kept in the refrigerator for subsequent SMP analysis. The previously obtained sludge pellet was re-suspended in 30 mL of phosphorus buffer solution. Cation exchange resin was added to the solution and a centrifuge process was then performed at 900 rpm for 2h to extract EPS from the pellet. The centrifuged EPS sample was filtered using 1.2  $\mu$ m Whatman 934-AH glass fiber filter (in order to separate particles) and then stored in the refrigerator. The SMP and EPS samples were analysed for proteins (SMP<sub>P</sub> and

 $EPS_P$ ) and polysaccharides (SMP<sub>C</sub> and  $EPS_C$ ) concentrations using Anthrone-sulfuric acid and modified Lowry method (Sigma, Australia) method, respectively.



Figure 3.9 Schematic diagram of processes for SMP and EPS extraction

Several physical properties of mixed liquor in the MBR and SSMBR were analysed, including relative hydrophobicity (RH), zeta potential, and particle size distribution. RH is determined by analyzing the distribution of sludge between mixed liquor and hydrocarbon. 50 mL mixed liquor and 50 mL n-hexane were added in a separatory funnel (Figure 3.8) and shaken by hand vigorously for 30 min. The mixture was then allowed to settle for 30 minutes to achieve complete separation of the two solvents.

The concentration of suspended solids (MLSS<sub>e</sub>) in the aqueous phase was measured. RH was calculated by RH (%) =  $(1-MLSS_e/MLSS_i) \times 100\%$ , where MLSS<sub>i</sub> is the initial concentration of suspended solids in the mixed liquor. The surface charge of sludge flocs (represented by Zeta potential) was measured by Zetasizer Nano ZS (Malvern Instruments, UK). Olympus System Microscope Model BX41 (Olympus, Japan, Figure 3.8) was used to obtain the images of sludge flocs for the subsequent examination of microorganisms and analysis of particle size distribution (using Image-Pro Plus software).



Figure 3.10 Relative hydrophobicity test (A) and microscopic analysis (B)



University of Technology, Sydney

Faculty of Engineering and Information Technology

# Chapter 4

# Micropollutant Removal in a Moving Bed Biofilm Reactor (MBBR) at Different HRTs

#### 4.1 Introduction

The ubiquitous occurrence of micropollutants and their metabolites in the aquatic environment has posed threats to living organisms to a great extent. However, effective micropollutants removal normally requires longer hydraulic retention time (HRT) when using biological treatment systems. As an ideal and low-cost material for attached-growth microorganisms, polyurethane sponge has exhibited high potential to eliminate micropollutants. In this study, a sponge-based moving bed biofilm reactor (MBBR) was investigated at four different HRTs (24, 18, 12, 6 h), to better understand the effect of HRT on micropollutants removal. Four groups of frequently detected micropollutants in wastewater (total 22 compounds) were selected, namely pharmaceuticals and personal care products (PPCPs), pesticides, hormones and industrial chemicals. The MBBR showed stable and effective removals of TOC (92.6% - 95.8%), COD (93.0% - 96.1%) and NH4-H (73.6%-95.6%) at all HRTs while improving PO<sub>4</sub>-P removal at HRT of 18 h. In terms of micropollutants removal, S-MBBR achieved comparable removal compared to other biological treatment such as activated sludge processes and membrane bioreactor. Although the micropollutants were subjected to biodegradation and sorption, the results indicated compound-specific variation in removal at all HRTs, ranging from 10.7% (carbamazepine) to 98.4% (ibuprofen). Among the selected micropollutants, most of them were biodegradable excluding carbamazepine, fenoprop and metronidazole. In addition, the micropollutants removal could remain constantly high even at lower HRTs with more consistent removal efficiency over the experimental period (except for carbamazepine, fenoprop, 17a-ethinylestradiol and 4-tertoctylphenol). Particularly, at HRT of 18 h, the removal of Diclofenac was significantly improved by more than 30% and the removals of ketoprofen, gemifibrozil, acetaminiphen, bisphenol A, and pentachlorophenol were also better. Overall, HRT of 18 h was the optimum HRT for biological degradation of the micropollutants.

#### 4.2 Organic and nutrient removal

Table 4.1 summarizes the removal efficiencies of DOC, COD, NH<sub>4</sub>-N, PO<sub>4</sub>-P and total nitrogen (TN) in MBBR at four HRTs (HRT of 24, 18, 12, and 6 h). The MBBR

was able to achieve effective removal of DOC (>94%) and NH<sub>4</sub>-N (>82%) at all HRTs. However, unstable TN (45.2-72.3%) reduction and

PO<sub>4</sub>-P (26.4-49.9%) elimination were observed throughout the experimental period. It was noteworthy that the MBBR showed the highest performance efficiency for removing DOC, COD, NH4-N, PO4-P and TN at HRT of 18 h, which were 96.1±0.4%, 97.4±0.8%, 91.1±1.6%, 49.9±7.2%, and 72.3±6.9%, respectively. This could be explained by the food to microorganisms (F/M) ratio in the MBBR unit, calculated based on COD and MLVSS. The F/M ratio were 0.72, 0.91, 1.83, and 3.06 g COD/ g MVLSS·d at HRT of 24, 18, 12, 6 h, respectively. The amount of food present in the system was sufficient to maintain microorganism growth when the F/M ratio was 0.91 g COD/ g MLVSS·d in the MBBR unit, which is in good agreement with the study of Pozo et al. (2012, 1.05 g COD/ g VSS·d). Villamar et al. (2009) also pointed out that the sludge showed good settler conditions when F/M ratio range between 0.63-1.26 g COD/ g VSS·d. In addition, higher NH<sub>4</sub>-N removal at HRT of 18 h could be attributed to the increased population of ammonium oxidation bacteria in the MBBR unit. Moreover, the use of phosphate for biomass growth and the phosphorus uptake by phosphate accumulating organisms (PAOs) could contribute to the high removal of PO<sub>4</sub>-P at HRT of 18 h.

**Table 4.1** Results for organic matters and nutrients removal by MBBR at HRT of 24 h, 18 h, 12 h, and 6 h

LIDT	Removal efficiency (%)										
IIIXI	TOC	COD	NH4-N	TN	PO <sub>4</sub> -P						
24	94.7±0.5	94.7±0.7	81.9±6.4	45.2±9.3	45.4±21.5						
18	96.1±0.4	97.5±0.8	91.1±1.6	72.3±6.9	49.9±7.2						
12	95.2±0.5	95.2±1.1	85.1±4.5	57.5±4.3	42.8±6.2						
6	94.0±0.4	94.1±0.5	82.0±2.1	46.9±3.2	26.4±6.1						



**Figure 4.1** TOC removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h).



**Figure 4.2** COD removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h).


**Figure 4.3** NH<sub>4</sub>-N removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h).



**Figure 4.4** TN removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h).



**Figure 4.5** PO<sub>4</sub>-P removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h).



**Figure 4.6** Summarized removal efficiencies in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h).



Figure 4.7 TOC removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h).



**Figure 4.8** COD removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h).



Figure 4.9 NH<sub>4</sub>-N removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h).



Figure 4.10 TN removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h).



Figure 4.11 PO<sub>4</sub>-P removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 18 h).



Figure 4.12 Summarized removal efficiencies in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 24 h).



**Figure 4.13** TOC removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h).



**Figure 4.14** COD removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h).



Figure 4.15 NH<sub>4</sub>-N removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h).



**Figure 4.16** TN removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h).



**Figure 4.17** PO<sub>4</sub>-P removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h).



**Figure 4.18** Summarized removal efficiencies in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 12 h).



Figure 4.19 TOC removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h).



**Figure 4.20** COD removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h).



Figure 4.21 NH<sub>4</sub>-N removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h).



**Figure 4.22** TN removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h).



**Figure 4.23** PO<sub>4</sub>-P removal in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h).



Figure 4.24 Summarized removal efficiencies in the MBBR (aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT: 6 h).

The MLSS and MLVSS concentrations in the MBBR units at HRT of 24 h, 18 h, 12 h, and 6 h were showed in Table 2. In the MBBR unit, the suspended MLSS  $(0.13\pm0.05 \text{ g/L}$  at HRT of 24 h,  $0.09\pm0.02 \text{ g/L}$  at HRT of 18 h,  $0.11\pm0.02 \text{ g/L}$  at HRT of 12 h, and  $0.13\pm0.02 \text{ g/L}$  at HRT of 6 h) and MLVSS  $(0.11\pm0.06 \text{ g/L}$  at HRT of 24 h,  $0.08\pm0.01 \text{ g/L}$  at HRT of 18 h,  $0.10\pm0.01 \text{ g/L}$  at HRT of 24 h,  $0.08\pm0.01 \text{ g/L}$  at HRT of 18 h,  $0.10\pm0.01 \text{ g/L}$  at HRT of 12 h, and  $0.12\pm0.01 \text{ g/L}$  at HRT of 6 h) concentrations of mixed liquid maintained very low, as the suspended solids were washed away into the subsequent MBR unit and no sludge was recycled back to the MBBR. Furthermore, a large amount of biomass was attached onto the sponge, which were  $0.45\pm0.05 \text{ g}$  MLSS/g sponge and  $0.41\pm0.06 \text{ g}$  MLVSS/g sponge at HRT of 18 h,  $0.44\pm0.04 \text{ g}$  MLSS/g sponge and  $0.40\pm0.03 \text{ g}$  MLVSS/g sponge at HRT of 12 h, as well as  $0.40\pm0.03 \text{ g}$  MLSS/g sponge and  $0.37\pm0.03 \text{ g}$  MLVSS/g sponge at HRT of 12 h.

HRT	Suspend	led	Attached		
	MLSS (g/L)	MLVSS (g/L)	SS(g/g Sponge)	VSS(g/g Sponge)	
24	0.123±0.051	0.108±0.052	0.385±0.107	0.348±0.103	
18	0.089±0.013	0.083±0.012	0.451±0.039	0.415±0.042	
12	0.105±0.009	0.096±0.010	0.435±0.035	0.400±0.032	
6	0.127±0.011	0.117±0.011	0.418±0.035	0.382±0.035	

**Table 4.2** Results for both suspended and attached biomass in MBBR at HRT of 24 h, 18 h, 12 h, and 6 h

### 4.3 Removals of selected micropollutants

The removal efficiencies of micropollutants in the MBBR at four HRTs are presented in Fig. 4.25. A significant variation in removal among each individual compound was observed over the study period. This could be related to certain factors of various micropollutants, such as hydrophobicity, microbial composition and compound structure. Among the selected micropollutants, most of them were biodegradable excluding Carbamazepine, Fenoprop and Metronidazole. In addition, the micropollutants removal could remain constantly high even at lower HRTs with more consistent removal efficiency over the experimental period (except for Carbamazepine, Fenoprop,  $17\alpha$ -Ethinylestradiol and 4-tert-Octylphenol). Particularly, at HRT of 18 h, the removal of Diclofenac was significantly improved by more than 30% and the removals of Ketoprofen, Gemifibrozil, Acetaminiphen, Bisphenol A, and Pentachlorophenol were also better. Overall, the highest removal of most micropollutants can be achieved at HRT of 18 h.

As complex synthetic substances, pharmaceuticals are classified into various groups with highly variable physico-chemical properties. As the selected pharmaceuticals (PPCPs) in this study generally displayed low hydrophobicity (log D < 2.5), biodegradation (rather than sorption) was the major removal pathway of these compounds (Luo et al., 2014a). The removal efficiency of these compounds was much more strongly affected by their intrinsic biodegradability when log D value was below 3.2 (Tadkaew et al., 2011). One possible reason is that these micropollutants have diverse molecular structure and functional groups (Wijekoon et al., 2013). Four of the investigated PPCPs were efficiently removal (> 80%), including ibuprofen (97.8±2.1% at HRT of 24 h, 98.4±1.2% at HRT of 18 h, 97.1±1.5% at HRT of 12 h, and 92.6±2.3% at HRT of 6 h), salicylic acid (96.3±0.9% at HRT of 24 h, 98.1±0.5% at HRT of 18 h, 97.3±1.6% at HRT of 12 h, and 90.0±3.9% at HRT of 6 h), primidone (81.1±13.0% at HRT of 24 h, 90.9±1.2% at HRT of 18 h, 87.2±3.5% at HRT of 12 h, and 82.5±5.0% at HRT of 6 h), and triclosan (90.9±4.9% at HRT of 24 h, 90.6±1.0% at HRT of 18 h, 82.6±4.4% at HRT of 12 h, and 80.4±4.2% at HRT of 6 h). This could be ascribed to the inclusion of strong electron donating (readily biodegradable) functional groups (e.g., - OH) in these compounds. Among the studied PPCPs, carbamazepine showed particularly low removals, which were 22.9±5.3% at HRT of 24 h, 26.2±5.7% at HRT of 18 h, 14.2±7.3% at HRT of 12 h, and 10.7±8.1% at HRT of 6 h. This low removal could be attributed to its low hydrophobicity and the occurrence of strong electron donating groups such as amide and chloride in its molecular structure (Wijekoon et al., 2013). It is noted that a significant increase in the removal of metronidazole and diclofenac, which was from 27.4±10.8% and 33.3±11.0% (HRT of 24 h) to 50.8±6.5% and 83.3±7.8% (HRT of 18h), respectively. Ketoprofen, acetaminophen, gemifibrozil, and naproxen were moderately removed (50-80%) in the MBBR at HRT of 24 h, 12 h, and 6 h, but

relatively high (80 - 90%) at HRT of 18 h. The formation of biocoenosis could be influenced by the change of HRT, resulting in the differences in removal efficiencies of micropollutants by the MBBR.

Regarding steroid hormones, high removals (> 90%) of estrone, estriol,  $\beta$ -estradiol 17-acetate, and 17 $\beta$ -estradiol were found at all HRTs, probably due to the high hydrophobicity (log D > 3.2) of these compounds (except estriol), which is in good agreement with Tadkaew et al. (2011).

For industrial chemicals, the observed removals for bisphenol A and 4-nnonylphenol were generally high (> 70%) at all HRTs, as these compounds are commonly characterized by high hydrophobicity (log D > 3.2). In this study, two pesticides displayed different removals. Fenoprop exhibited inefficient elimination (16.5 – 50.5%), whereas pentachlorophenol experienced much higher removal (71.3 – 92.8%). The poor removal of fenoprop could be attributed to its low hydrophobicity (log D = -0.13) and recalcitrance (Hai et al., 2011).



Figure 4.25 Micropollutant removals in the MBBR unit at HRT of 24 h, 18 h, 12 h, and 6 h. The error bar of each sample represents the standard deviation over the experiment period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h, 18h, 12h and 6 h).



Figure 4.26 Relationship of Log D and pharmaceutical compounds removals in the MBBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 24h, 18h, 12h and 6h)







Figure 4.28 Relationship of Log D and industrial chemicals removals in the MBBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L;

HRT = 24h, 18h, 12h and 6h)



Figure 4.29 Relationship of Log D and estrogenic hormones removals in the MBBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h, 18h, 12h and 6h)

To gain further insight into the fate of micropollutants in the MBBR unit, a mass balance of the investigated compounds was evaluated (Eq. (1)), taking into account the removal pathways of biodegradation and sorption in the MBBR unit.

$$L_{inf} = L_{s,MBBR} + L_{b,MBBR} + L_{eff}$$
(1)

where  $L_{inf}$  is the influent load of micropollutants over the experimental period (ng),  $L_{s,MBBR}$  and  $L_{b,MBBR}$  are the amounts of a compound removed via sorption (ng) and biodegradation (ng), respectively, in the MBBR unit;  $L_{eff}$  is the amount of a compound released from the system (ng).

The calculation of the sorption and biodegradation in the MBBR was carried out according to Eq. (2) and Eq. (3).

$$L_{s,MBBR} = Q \cdot MLSS_{MBBR} \cdot C_{ss} \cdot t + \Delta SS \cdot C_{sa}$$
<sup>(2)</sup>

$$L_{b,MBBR} = (L_{inf} - L_{eff}) - L_{s,MBBR}$$
(3)

where Q is the flow rate of the MBBR (L/day); MLSS<sub>MBBR</sub> is mixed liquor suspended biosolids concentration in the MBBR (g/L);  $C_{ss}$  is the concentration of a compound on the suspended biosolids (ng/g); t is the duration of the experimental period (days);  $\Delta$ SS is the increased amount of attached biosolids (g);  $C_{sa}$  is the concentration of a compound on the attached biosolids (ng/g).

Fig. 4.30-4.33 illustrates the fate of the selected compounds in the MBBR unit at HRT of 24, 18, 12, and 6 h. The results show that biodegradation in the MBBR was the main pathway for the micropollutant removal, which is consistent with outcomes from our previous study involving both MBBR-MBR and CMBR (Luo et al., 2015). Compared to biodegradation, sorption accounted for much less proportion of most micropollutant removals. This probably due to that sorption is a more rapid process than biodegradation. In addition, sorption of micropollutants to biosolids results in longer residence time in the reactor, which may lead to further removal via biodegradation (Tadkaew et al., 2011). Moreover, the acclimatized sponge in the MBBR unit remained fully occupied by biomass over time, and the reduced sorption

site resulted in the limited sorption efficiency in long term experiment (Luo et al., 2014b). As a result, although most micropollutants were effectively removed at HRT of 18 h, the decreased removals of  $17\alpha$ -ethinylestradiol, triclosan and 4-tert-octylphenol were observed. On the other hand, the proportion of sorption varied at different HRTs. It is notable that the percentage of sorption increased after HRT changed to 6 h. One reason could be the excessive nutrient in the MBBR unit leading to over growth and detachment of biofilm. The new biofilm could help adsorbing micropollutant to a certain extent.



Figure 4.30 Fate of the studied micropollutants in the MBBR system (aeration rate: 4L/min; HRT: 24 h).



Figure 4.31 Fate of the studied micropollutants in the MBBR system (aeration rate: 4L/min; HRT: 18 h).



Figure 4.32 Fate of the studied micropollutants in the MBBR system (aeration rate: 4L/min; HRT: 12 h).



Figure 4.33 Fate of the studied micropollutants in the MBBR system (aeration rate: 4L/min; HRT: 6 h).

# 4.4 Comparison between the MBBR and other techniques for micropollutant removal

Since WWTPs are not able to provide a complete barrier for micropollutant removal, establishing optimal removal strategies for micropollutants remains a challenge to environmental engineers in order to minimize their adverse effects on the environment. Table 4.3 compares the micropollutants removal in the studied MBBR, suspended-growth activated sludge processes and MBRs. Generally, the effectiveness of MBBR for micropollutant removal was comparable with those of activated sludge processes and MBRs. Moreover, the MBBR seemed to be very effective in eliminating most of the investigated micropollutants, except for carbamazepine, fenoprop and metronidazole. The presence of biofilm carriers played an important role in the biodegradation of diclofenac and biofilm carriers from full-scale nitrifying wastewater treatment plants had apparently higher removal rates per unit biomass for diclofenac in comparison to the activated sludge (Falås et al. 2012). Nevertheless, MBR systems had the ability for more efficient removal of 4-tertbutylphenol and 4-tert-butylphenol (Nguyen et al., 2013; Wijekoon et al., 2013; Tadkaew et al. 2011). The MBBR in this study was able to achieve similar or higher elimination of the estrogenic hormones (70.7–96.6%) compared with the activated sludge processes, despite Servos et al. (2005) stated that activated sludge processes tended to have higher removal of estrogenic potentials (58->99% with an average of 81%) than some attached-growth processes (0–75% with an average of 28%), such as trickling filters and rotating biological contactors.

Compounds <sup>a</sup>		This study (MBBR,	Activated sludge <sup>b</sup>	MBR <sup>c</sup>
		HRT=18h, %)	(%)	(%)
Pharmaceuticals	ACM	83.7 ±4.4	98.7-100	40–100
	CBZ	$26.2\pm5.7$	<0-63.2	0–35
	DCF	$83.3 \pm 7.8$	<0-81.4	0-87
	GFB	$86.0 \pm 3.2$	<0-92.3	90–98
	$\mathrm{IBU}^{\mathrm{d}}$	$98.4 \pm 1.2$	72–100	50-99
	KTP	$80.7\pm6.2$	10.8–100	52–92
	MET	$50.8\pm6.5$	0-64.0	36–40
	NPX	$80.0\pm3.7$	43.3–98.6	10-84
	PRM	$90.9\pm2.7$	30-50	10–91
	SA	$98.1\pm0.5$	89.6-100	93–98
	TCS	$90.6 \pm 1.0$	71.3–99.2	70–99
Steroid hormones	E1	$94.1 \pm 1.1$	74.8–90.6	96–99
	E2	$96.6\pm0.7$	92.6-100	97–99
	E2AC	$96.3\pm0.8$	-	98–99
	EE2	$70.7\pm3.6$	43.8–100	60–98
	E3	$94.9 \pm 1.8$	100	83–97
Industrial chemicals	BP	$64.8\pm3.9$	-	93–98
	BPA	$91.7 \pm 1.6$	62.5–99.6	52–98
	NP	$95.4\pm2.1$	21.7–99	87–97
	OP	$75.1 \pm 3.8$	<0–96.7	97-98
Pesticides	FNP	$50.5\pm7.6$	-	10–21
	PECP	$92.8\pm2.0$	-	61–99

**Table 4.3** Comparison of micropollutant removal efficiency (%) in the MBBR and in other biological treatment technologies

<sup>a</sup> ACM: acetaminophen; BP: 4-tert-butylphenol; BPA: bisphenol A; CBZ: carbamazepine; DCF: diclofenac; E1: estrone; E2: 17- $\beta$ -estradiol; E2AC:  $\beta$ -estradiol 17-acetate; EE2: 17- $\alpha$  ethinylestradiol; E3: estriol; FNP: fenoprop; GFB: gemfibrozil; IBP: ibuprofen; KTP: ketoprofen; MET: metronidazole; NP: 4-n-nonylphenol; NPX: naproxen; PECP: pentachlorophenol; PRM: primidone; OP: 4-tert-octylphenol; SA: salicylic acid; TCS: triclosan.

<sup>b</sup> Data from Kasprzyk-Hordern et al., 2009; Lin et al., 2009; Luo et al., 2014; Wick et al., 2009.

<sup>c</sup> Data from Nguyen et al., 2013.

<sup>d</sup> Listed in bold type are the compounds whose percentage removals are close to or above the maximum removals reported in activate sludge or/and MBR.

### 4.5 Conclusion

Overall, the removal efficiency by the MBBR is comparable with other processes (activated sludge and MBR). The MBBR appeared to be an effective process for removal of ibuprofen, metronidazole, naproxen, primidone, triclosan, estrone, 17- $\alpha$  ethinylestradiol, 4-n-nonylphenol, 4-tert-octylphenol and fenoprop. In this set of experiment, the optimal HRT was 18h, thus under this condition, further studies could be done to optimise the MBBR system in order to achieve better removal.



University of Technology, Sydney

Faculty of Engineering and Information Technology

## Chapter 5

## Removal of Micropollutants by an Moving Bed Biofiln Reactor-Membrane Bioreactor Hybrid System at Different HRTs

### 5.1 Introduction

The frequent detection of micropollutants in our aquatic environment has gained growing attention in recent years. Even though these micropollutants are at trace concentrations, they may cause long term effects, such as bioaccumulation and carcinogenicity (Soares et al., 2008, Klavarioti et al., 2009). In addition, low biodegradability and highly resistance of micropollutants make their efficient removal a great challenge in conventional wastewater treatment processes (Bu et al., 2013). Generally, carbamazepine, ibuprofen, ketoprofen, naproxen, gemfibrozil, nonylphenol and triclosan were commonly detected in treated effluent at relatively high concentrations (4.60, 55.0, 3.92, 5.09, 5.24, 7.80 and 6.88  $\mu$ g/L, respectively) (Luo et al., 2014a).

Although some wastewater treatment technologies (e.g. oxidation and advanced oxidation processes, powdered activated carbon addition, nanofiltration and reverse osmosis ) have been employed to minimise micropollutants from municipal wastewater, they are normally associated with high operation cost, formation of oxidation by-products, requirement of chemical dosage, as well as permeability to some relatively low molecular weight micropollutants (Gerrity et al., 2011; Boehler et al., 2012; Sahar et al., 2011). MBRs, as a promising biological treatment technology, have been widely used to remove micropollutants (Hai et al., 2011). Radjenovic et al. (2007) and Chen et al. (2008) compared the performance of micropollutant removal in a submerged MBR and a conventional activated sludge (CAS) treatment process (consisting of an aeration tank and a settling tank). Both studies showed that the MBR was more efficient in micropollutant removal than CAS process. Trinh et al. (2012) investigated the removal of 48 trace micropollutants by a full-scale submerged MBR, including steroidal hormones, xenoestrogens, pesticides, caffeine, pharmaceuticals and personal care products. Even though the MBR could remove most examined trace contaminants (>90%) at high filtration flux of 25 L/m<sup>2</sup>·h and hydraulic retention time (HRT) of 1 d, regular physical cleaning (every 360 seconds for a period of 60 seconds) and chemical backwashing (every 3 weeks) were performed to maintain a transmembrane pressure (TMP) less than 20 kPa. Similarly, Tadkaew et al. (2011) conducted experiment to investigate the

removal efficiency in a submerged MBR with respect to 40 micropollutants. The MBR was operated at HRT of 24 h and filtration flux of 4.3 L/m<sup>2</sup>·h, along with a relaxation cycle of 14 min suction and 1 min off. Wijekoon et al. (2013) examined the fate of 29 trace organic compounds in a bench scale external MBR with a HRT of 26 h. However, the MBR was operated at very low membrane flux (2.14  $L/m^2 \cdot h$ ) to eliminate the influence of membrane fouling on micropollutant removal. Moreover, 15 min on and 15 min off relaxation cycle was employed to minimise membrane fouling. Regardless of changes in configurations, general concerns remain during the application of MBR systems. First of all, the membrane fouling is still the major problem while employing membrane related technology. Frequent physical and chemical backwashing or relaxation is essential for alleviating membrane fouling. In addition, it requires long HRT in MBR to remove micropollutants but still has inconsistent removal of some polar and persistent hydrophilic micropollutants, e.g. carbamazepine, diclofenac, and gemfibrozil etc. Therefore, in order to effectively remove micropollutants, optimization of the performance of MBR has been the subject of recent studies and is likely to draw more attention in the future research.

Moving bed biofilm reactor (MBBR) is a fast growing wastewater treatment technique with potential to effectively remove both traditional pollutants and micropollutants (Luo et al., 2014b). Combination of MBBR with MBR in series (MBBR-MBR) is one of recent advances in improving MBR performance, due to the potential of reducing membrane fouling as well as the possibility of prompting microbial degradation of certain organic compounds (Leiknes and Ødegaard, 2007). In comparison to conventional MBR in municipal wastewater treatment, this hybrid system is able to operate at 3-4 times higher flux rate with 10-15 times higher organic loading rate (OLD) and 10-30 times shorter HRT (Guo et al., 2012). Moreover, MBBR-MBR hybrid system could lower fouling and prolong filtration duration compared to conventional MBR (Sombatsompop et al., 2006). With the objective of enhancing micropollutant removal efficiency in MBRs, the impacts of operational parameters including SRT (Weiss and Reemtsma, 2008), pH (Tadkaew et al., 2010) and temperature (Hai et al., 2011) have also been specifically investigated. A general conclusion can be drawn from these studies is that MBRs can achieve

good and stable micropollutant removal under extended SRT (> 15 days), near neutral pH (pH 6-8) and warmer temperature (20 - 35  $^{\circ}$ C).

Apart from the above-mentioned operational parameters, HRT can also potentially affect the degree of biodegradation and adsorption of micropollutants in MBRs. Generally, mixed liquid suspended solid (MLSS) concentration and sludge viscosity increases with shortened HRT, resulting in the increase in extracellular polymeric substances (EPS) and soluble microbial products (SMP), thereby affecting membrane fouling propensity (Meng et al., 2007; Huang et al., 2011; Shariati et al., 2011). With regard to micropollutants removal in MBR, prolonged HRT enhances the formation of a diverse biocoenosis, including some slow growing bacteria. Higher biodegradation efficiency for certain micropollutants can be achieved when biomass is rich in nitrifying bacteria (Kim et al., 2007; Roh et al., 2009). De Gusseme et al. (2009) reported a high removal (97.7 $\pm$ 3.7%) of 17 $\alpha$ -ethinylestradiol at HRT of 4 d in a submerged MBR, but lower removal (58.3±2.4%) at HRT of 1 d. Schröder et al., (2012) showed the performance of MBR treatment plant in removing 6 pharmaceutical compounds was better at HRT of 13 h than that of at HRT of 9 h. Nevertheless, Chen et al. (2008) reached a conclusion that the changes in HRT (HRT of 8, 6, and 4 h) had little effect on removal efficiency of Bisphenol A in a lab-scale submerged MBR, probably due to its rapid biodegradation and low potential for bioaccumulation. To date, there has been no comprehensive study regarding the impact of HRT for micropollutant removal. In our previous investigation, a comparison has been performed regarding the micropollutant removal efficiency between an MBBR-MBR hybrid system and a conventional MBR (CMBR) (Luo et al., 2015). The results indicated that MBBR as pretreatment to MBR exhibited better performance than single CMBR, and more importantly MBBR can mitigate membrane fouling of MBR to a great extent. Therefore, in this study, the performance of MBBR-MBR hybrid system was investigated at four different HRTs (24 h, 18 h, 12 h, and 6 h) to determine the optimal HRT in terms of micropollutant removal and membrane fouling control. Moreover, the membrane fouling propensity was examined based on mixed liquor characteristics, such as SMP, EPS, zeta potential, and relative hydrophobicity (RH).

### 5.2 Organic and nutrient removal

Table 5.1 summarizes the removal efficiencies of DOC, COD, NH<sub>4</sub>-N, PO<sub>4</sub>-P and total nitrogen (TN) in MBBR-MBR hybrid system at four HRTs (HRT of 24, 18, 12, and 6 h). As shown in Table 5.1, the MBBR-MBR system was able to achieve effective removal of TOC (>94%) and NH<sub>4</sub>-N (>82%) at all HRTs. However, unstable TN (46.8-68.7%) reduction and PO<sub>4</sub>-P (32.5-51.7%) elimination were observed throughout the experimental period. It was noteworthy that the MBBR-MBR showed the highest performance efficiency for removing DOC, COD, NH<sub>4</sub>-N, PO<sub>4</sub>-P and TN at HRT of 18 h, which were 96.7±0.2%, 98.1±0.6%, 92.1±1.6%, 51.7±8.3%, and 68.7±7.2%, respectively. This could be explained by the food to microorganisms (F/M) ratio in the MBBR-MBR system, calculated based on COD and MLSS. The F/M ratio were 0.31, 0.43, 0.68, and 1.28 g COD/ g MLSS·d at HRT of 24, 18, 12, 6 h, respectively. The amount of food present in the system was sufficient to maintain microorganism growth when the F/M ratio was 0.43 g COD/ g MLSS d in the MBBR-MBR system, which is in good agreement with Pozo et al. (2012). Villamar et al. (2009) also pointed out that the sludge showed good settler conditions when F/M ratio range between 0.3-0.6 g BOD<sub>5</sub>/ g VSS·d. In addition, higher NH<sub>4</sub>-N removal at HRT of 18 h could be attributed to the increased population of ammonium oxidation bacteria in the MBBR-MBR. Moreover, the use of phosphate for biomass growth and the phosphorus uptake by phosphate accumulating organisms (PAOs) could contribute to the high removal of PO<sub>4</sub>-P at HRT of 18 h.

	Removal efficiency (%)					
HKI	TOC	COD	NH4-N	TN	PO <sub>4</sub> -P	
24	95.0±0.6	95.5±0.9	86.5±3.5	47.4±9.2	34.9±3.9	
18	96.7±0.2	98.1±0.6	92.1±1.6	68.7±7.2	51.7±8.3	
12	96.0±0.4	96.1±0.8	87.5±1.2	58.1±2.8	42.7±4.3	
6	94.7±0.1	95.0±0.3	83.1±1.2	46.8±1.0	32.5±3.8	

**Table 5.1** Results for organic matters and nutrients removal by MBBR-MBR at HRT of 24 h, 18 h, 12 h, and 6 h





(MBBR conditions: aeration rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.2 COD removal in the MBBR-MBR (MBBR conditions: aeration Rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).





(MBBR conditions: aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.4 TN removal in the MBBR-MBR (MBBR conditions: aeration rate: 4 L/min; DO: 5.5–6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^{2}$ ·h; HRT = 6h; SRT =  $\infty$ ).





MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.6 Summarized removal efficiencies in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=24 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).


Figure 5.7 TOC removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.8 COD removal in the MBBR-MBR

(MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.9 NH<sub>4</sub>-N removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.10 TN removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2$ ·h; HRT = 6h; SRT =  $\infty$ ).





(MBBR conditions: aeration rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.12 Summarized removal efficiencies in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=18 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2$ ·h; HRT = 6h; SRT =  $\infty$ ).



Figure 5.13 TOC removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.14 COD removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2$ ·h; HRT = 6h; SRT =  $\infty$ ).





(MBBR conditions: aeration rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.16 TN removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2$ ·h; HRT = 6h; SRT =  $\infty$ ).





(MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.18 Summarized removal efficiencies in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=12 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2$ ·h; HRT = 6h; SRT =  $\infty$ ).





(MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.20 COD removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^{2}$ ·h; HRT = 6h; SRT =  $\infty$ ).







Figure 5.22 TN removal in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).





(MBBR conditions: aeration rate = 4 L/min; DO = 5.5–6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).



Figure 5.24 Summarized removal efficiencies in the MBBR-MBR (MBBR conditions: aeration rate = 4 L/min; DO: 5.5–6.5 mg/L; HRT=6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83  $L/m^2 \cdot h$ ; HRT = 6h; SRT =  $\infty$ ).

The MLSS and MLVSS concentrations in the MBBR-MBR hybrid system were showed in Table 5.2. In the MBBR unit, the MLSS and MLVSS concentrations were very low, as the suspended solids were washed away into the subsequent MBR unit and no sludge was recycled back to the MBBR. Regarding the MBR unit, the initial MLSS and MLVSS were similar to the values of suspended MLSS and MLVSS concentrations in MBBR tank, and all showed gradual growth during the operation, reaching 0.91 and 0.89 g/L (HRT of 24 h), 1.02 and 0.98 g/L (HRT of 18 h), 1.09 and 1.04 g/L (HRT of 12 h), and 1.05 and 0.96 g/L (HRT of 6 h), respectively.

HRT	MB	BR	MBR		
	Suspended MLSS (g/L)	Attached biosolids (g/g <sub>sponge</sub> )	Initial MLSS (g/L)	Initial MLVSS (g/L)	
	Suspended MLVSS (g/L)	Attached biomass (g/g <sub>sponge</sub> )	Final MLSS (g/L)	Final MLVSS (g/L)	
24 h	0.13±0.05	0.45±0.05	0.06	0.05	
	0.11±0.06	0.41±0.06	0.91	0.89	
18 h	0.09±0.02	0.51±0.03	0.09	0.07	
	0.08±0.01	0.47±0.03	1.02	0.98	
12 h	0.11±0.02	0.44±0.04	0.11	0.10	
	0.10±0.01	0.40±0.03	1.09	1.04	
6 h	0.13±0.02	0.40±0.03	0.15	0.11	
	0.12±0.01	0.37±0.03	1.05	0.96	

**Table 5.2** Results for both suspended and attached biomass in MBBR unit and MBR unit at HRT of 24 h, 18 h, 12 h, and 6 h

### 5.3 Removals of selected micropollutants

During the MBBR-MBR treatment, the compound-specific removal efficiencies varied significantly, ranging from 11.0 to 99.5%, without an evident correlation (correlation coefficient = 0.35) to their Log Ds. However, most compounds were eliminated to large extents (>70%). One possible reason for the high removal efficiency is that the attached growth pattern could enhance the retention of the biomass, thus promoting the enrichment of slow growing microorganisms and the formation of a diverse biocoenosis. Even MBR can prevent the washout of slowgrowing microorganisms like nitrifiers, the impact of MBR removal was minimal. This may probably due to the low MLSS concentration and the large pore size (0.2 µm; two orders of magnitude larger than the molecular sizes of micropollutants) of the MF membrane used in this study. Nevertheless, the MBR unit was able to complement the removal of a few compounds including metronidazole and carbamazepine. Both metronidazole and carbamazepine are nitrogen bearing compounds, where nitrogen is bound to the cyclic structure. The infinite SRT applied in this study could have facilitated the enhanced removal of the nitrogenous compounds mentioned above. Wijekoon et al. (2013) stated that the removal of nitrogen bearing compounds could be selectively enhanced by the nitrifying microbial consortium, but detailed study on the effect of the location of nitrogen molecules in nitrogenous compounds on their degradation by nitrfiers would be required to substantiate this hypothesis. In addition to size exclusion, membrane can intercept the release of micropollutants through charge repulsion, adsorption onto membrane surface, sorption diffusion, solute-solute interactions and fouling layer interactions (Schäfer et al. 2011). For example, Chang et al. (2003) ascribed the complete removal of estrone (feed concentrations: 2.6-154 ng/L) in a MF dead-end process to adsorption on the membrane. However, in this study, estrone (feed concentrations: 43-118 ng/L) experienced a very limited further removal in the MF process. A possible cause was that the adsorption sites of the membrane quickly saturated after the beginning of the operation, and the absence of adsorption sites impeded the retention of estrone. Additionally, biological removal might be helpful in the reduction of the compounds that were less likely to interact with particulate matter and membrane surface, such as gemfibrozil, pentachlorophenol and diclofenac. In this study, a better removal of certain micropollutants was observed at optimal HRT in MBBR unit (HRT of 18 h) in comparison to several previous studies involving both MBBR (Pozo et al., 2012) and MBR (Wijekoon et al., 2013; Tadkaew et al., 2011). A high removal (92.1±2.5%) of Bisphenol A was obtained at HRT of 18 h in this study, which was 52.8% higher than a low removal of total phenolic compound removal (39.27±8.09%) by a bench scaled MBBR previously reported by Pozo et al. (2012) at HRT of 19.2 h. Pozo et al. (2012) also indicated that HRT exerted an influence on the removal efficiency of a system, which supports the point of view in current study. Relatively higher removal of diclofenac (84.1%) could also be noticed compared to the lower removal (27%) reported by Wijekoon et al. (2013) and (17.3%) by Tadkaew et al. (2011). This could be attributed to the difference in the microbial composition of the sludge and sludge concentrations between the MBBR-MBR system and MBR system. Over time, phenolic and phthalate degrading bacteria were developed along with improved micropollutants biodegradation (Boonnorat et al., 2014).

The impact of hydrophobicity on the behaviour of micropollutants removal in the MBBR-MBR system was also investigated. Noting that most of the hydrophobic compounds presented very low solid phase concentration at all HRTs, despite the high hydrophobicity (Log D>3.2), such as 4-n-nonylphenol. In contrast, some persistent hydrophilic compounds (e.g. carbamazepine) were consistently detected at high concentrations in biosolids.



Figure 5.25 Micropollutant removals in the MBBR-MBR while MBBR unit operating at HRT of 24 h, 18 h, 12 h, and 6 h. The error bar of each sample represents the standard deviation over the experiment period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h, 18h, 12h and 6 h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT = ∞)





HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4– 7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )



**Figure 5.27** Relationship of Log D and pesticides removals in the MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR

conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L,

HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4– 7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )



**Figure 5.28** Relationship of Log D and industrial chemicals removals in the MBBR-MBR. An error bar represents the standard deviation over the experimental period

(MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L,

HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4– 7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )





7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )

To gain further insight into the fate of micropollutants during the MBBR-MBR treatment, a mass balance of the investigated compounds was evaluated (Eq. 1), taking into account the removal pathways of biodegradation and sorption in the MBBR unit, and total removal in the MBR unit.

$$L_{inf} = L_{s,MBBR} + L_{b,MBBR} + L_{MBR} + L_{eff}$$
(6.1)

where  $L_{inf}$  is the influent load of micropollutants over the experimental period (ng),  $L_{s,MBBR}$  and  $L_{b,MBBR}$  are the amount of a compound removed via sorption (ng) and biodegradation (ng), respectively, in the MBBR unit;  $L_{MBR}$  is the amount of a compound removed in the MBR (ng);  $L_{eff}$  is the amount of a compound released from the system (ng).

The calculation of the sorption (Eq. 2) and biodegradation (Eq. 3) in MBBR was carried out according to Luo et al. (2014).

$$L_{s,MBBR} = Q \cdot MLSS_{MBBR} \cdot C_{ss} \cdot T + \Delta SS \cdot C_{sa}$$
(6.2)

$$L_{b,MBBR} = (L_{inf} - L_{eff,MBBR}) - L_{s,MBBR}$$
(6.3)

where Q is the flow rate of the MBBR-SMBR system (L/day); MLSS<sub>MBBR</sub> is mixed liquor suspended biosolids concentration in the MBBR (g/L);  $C_{ss}$  is the concentration of a compound on the suspended biosolids (ng/g); T is the duration of the experimental period (days);  $\Delta$ SS is the increased amount of attached biosolids (g);  $C_{sa}$  is the concentration of a compound on the attached biosolids (ng/g);  $L_{eff, MBBR}$  is the amount of a compound released from the MBBR unit (ng).

Regarding the calculation for the removal in the MBR unit, the following equation was used:

$$L_{MBR} = L_{eff,MBBR} - L_{eff}$$
(6.4)

Fig. 5.30-5.33 illustrates the fate of the selected compounds in the MBBR-MBR system at HRT of 24, 18, 12, and 6 h. The results show that biodegradation in the MBBR was the main pathway for the micropollutant removal, which is consistent

with outcomes from our previous study involving both MBBR-MBR and CMBR (Luo et al., 2015). Compared to biodegradation, sorption accounted for much less proportion of most micropollutant removals. This probably due to that sorption is a more rapid process than biodegradation. In addition, sorption of micropollutants to biosolids results in longer residence time in the reactor, which may lead to further removal via biodegradation (Tadkaew et al., 2011). Moreover, the acclimatized sponge in the MBBR unit remained fully occupied by biomass over time, and the reduced sorption site resulted in the limited sorption efficiency in long term experiment (Luo et al., 2014b). As a result, although most micropollutants were effectively removed at HRT of 18 h, the decreased removals of 17α-ethinylestradiol, triclosan and 4-tert-octylphenol were observed. On the other hand, the proportion of sorption varied at different HRTs. It is notable that the percentage of sorption increased after HRT changed to 6 h. One reason could be the excessive nutrient in the MBBR unit leading to over growth and detachment of biofilm. The new biofilm could help adsorbing micropollutant to a certain extent. Even MBR can prevent the washout of slow-growing microorganisms like nitrifiers, the impact of MBR removal was minimal at all HRTs. This may probably due to the low MLSS concentration and the large pore size  $(0.2 \ \mu m)$ ; two orders of magnitude larger than the molecular sizes of micropollutants) of the MF membrane used in this study. Nevertheless, the MBR unit was able to complement the removal of a few compounds including metronidazole and carbamazepine. Both metronidazole and carbamazepine are nitrogen bearing compounds, where nitrogen is bound to the cyclic structure. The infinite SRT applied in this study could have facilitated the enhanced removal of the nitrogenous compounds mentioned above. Wijekoon et al. (2013) stated that the removal of nitrogen bearing compounds could be selectively enhanced by the nitrifying microbial consortium, but detailed study on the effect of the location of nitrogen molecules in nitrogenous compounds on their degradation by nitrfiers would be required to substantiate this hypothesis. Additionally, the observed complementary separation effect of the MF membrane could be attributed to the sorption of micropollutants to suspended solid particles and the retention of these solid particles by the MF membrane (Schäfer et al. 2011).



Figure 5.30 Fate of the studied micropollutants in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 24h;
MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT = ∞)



Figure 5.31 Fate of the studied micropollutants in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 18h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT = ∞)



Figure 5.32 Fate of the studied micropollutants in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L, HRT = 12h;
MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT = ∞)



**Figure 5.33** Fate of the studied micropollutants in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L, HRT = 6h;

MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )

### 5.4 Membrane fouling analysis

Fig. 5.34 depicts the time course of TMP increase in the MBR unit. The TMP development in the MBR unit was varied when the MBBR unit were operating at four HRTs. The TMP in the MBR unit reached up to 35 kPa, 36.5 kPa, 37.5 kPa and 36.5 kPa in 89 days, 74 days, 42 days and 20 days of operation, respectively, at HRT of 24 h, 18 h, 12 h, and 6 h in MBBR unit. The fouling rate were 0.39 kPa/d (HRT of 24 h), 0.49 kPa/d (HRT of 18 h), 0.89 kPa/d (HRT of 12 h), and 1.83 kPa/d (HRT of 6 h). In our previous study, it is concluded that the use of MBBR as the pre-treatment can lower the fouling propensity and improve the filtration performance of MBR by comparing the TMP profile gained from the MBBR-MBR hybrid system (HRT of 24 h) and a CMBR. According to the result from this experiment, a longer HRT (e.g. HRT of 24 h) can significantly mitigate membrane fouling when compared with a relatively short HRT (e.g. HRT of 6 h). Especially, the TMP value maintained less than 15 kPa for 60 days (HRT of 18 h) and 68 days (HRT of 24 h).



**Fig. 5.34** Comparison of TMP profiles in the MBR while MBBR was operating at HRT of 24 h, 18 h, 12 h, and 6 h (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2$ ·h; HRT = 6h; SRT =  $\infty$ )

Shirazi et al. (2010) reviewed the application of a resistance-in-series model for assessing membrane fouling using Darcy's Law. The model is as follows:

$$\mathbf{J} = \Delta \mathbf{P}_{\mathrm{T}} / (\boldsymbol{\mu} \cdot \mathbf{R}_{\mathrm{t}}) \tag{5.5}$$

$$\mathbf{R}_{\mathrm{t}} = \mathbf{R}_{\mathrm{m}} + \mathbf{R}_{\mathrm{C}} + \mathbf{R}_{\mathrm{f}} \tag{5.6}$$

Where, J is permeation flux;  $\Delta P_T$  is transmembrane pressure;  $\mu$  is viscosity of the permeate;  $R_t$  is total resistance of membrane filtration;  $R_m$  is intrinsic membrane resistance caused by membrane itself and permanent resistance;  $R_c$  is cake resistance formed by cake layer deposited over membrane surface;  $R_f$  is fouling resistance caused by pore plugging and/or solute adsorption onto the membrane pore and surface.

After the experiment, the fouled membrane was subjected to physical and chemical cleaning, in order to determine different membrane fouling resistances using the equations given above. Results of fouling resistance were showed in Table 5.3. The results suggested that cake layer formation was the main factor contributing to membrane fouling at all HRTs. However, pore blocking resistance was higher at short HRT than high HRT. As suspended MLSS concentrations in MBBR were different at each HRT, it could affect the sludge properties in subsequent MBR unit. Higher MLSS concentration may cause formation of sticky cake layer on membrane surface due to higher sludge viscosity (Deng et al., 2014). Besides, small sludge flocs could lead to more severe membrane fouling and induce higher membrane surface. Hence, severe membrane fouling and higher fouling rate (Fig. 5.34) occurred at HRT of 6 h.

	HRT=24 h		HRT=18 h		HRT=12 h		HRT=6 h	
Parameter	Resist ance (10 <sup>12</sup> / m)	Perce ntage (%)						
Clean membrane	1.33	10.5	1.33	11.0	1.33	11.0	1.33	11.0
Cake layer	9.63	76.5	8.67	71.8	8.17	70.7	7.83	69.4
Pore blocking	1.51	12.0	1.90	15.7	1.89	16.3	1.92	17.0
Irreversible fouling	0.12	1.0	0.18	1.5	0.16	1.5	0.20	1.7
Total resistance	12.59	100	12.08	100	11.55	100	11.28	100

 Table 5.3 Membrane resistances for different types of fouling.

The severe membrane fouling is often a result of organics accumulation on or in the membrane as bound EPS or SMP, and polysaccharides and proteins are considered as the major fractions of EPS and SMP (Guo et al., 2012, Meng et al., 2009). Fig. 5.35 exhibits the composition of mixed liquor's SMP and bound EPS at different designed TMPs in MBBR-MBR system at HRT of 24 h, 18 h, 12 h, and 6 h. The level of EPS were similar at the beginning of all HRTs, then gradually increased to 15.24 mg/L, 16.43 mg/L, 19.88 mg/L and 22.93 mg/L at the end of operation for HRT of 24 h, 18 h, 12 h, and 6 h, respectively. The SMP concentration varied for different HRTs but showed minor variation under the same HRT. The SMP concentration was lower at HRT of 24 h, while a significantly higher SMP concentration was observed at HRT of 6 h. One reason could be the shortened HRT led to increase in nutrient availability, which may accelerate the death of curtain bacteria releasing SMP.



Figure 5.35 Comparison of EPS and SMP values in the MBR while MBBR was operating at HRT of 24 h, 18 h, 12 h, and 6 h (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h, 18h, 12h and 6h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT = ∞)

The sludge characteristics (e.g. Zeta potential, hydrophobicity and floc size) were presented at Table 5.4. The zeta potential increased while RH decreased, when the HRT was changed from 24 h to 6 h. According to Ji et al. (2010), increase of zeta potential could neutralise the negative charge on the surface of the floc and form large aggregates, while high RH generally resulted in less interaction between the hydrophobic flocs and the hydrophilic membrane. The floc size of sludge was larger at higher HRT than that of at lower HRT.

HRT	Sludge characteristics					
	Zeta potential (mV)	RH (%)	Floc size (µm)			
24 h	-12.1 to -19.0	75.0-83.3	15-40			
18 h	-11.8 to - 19.6	73.0-80.9	15-40			
12 h	-16.1 to -20.2	70.8-78.6	10-30			
6 h	-18.1 to -20.9	68.4-75.5	10-25			

Table 5.4 Sludge characteristics in MBR unit at HRT of 24 h, 18 h, 12 h, and 6 h

### 5.5 Conclusion

The investigated MBBR-MBR demonstrated excellent performance in removing organics, nutrients as well as micropollutants. By using an MBBR as a pretreatment to an MBR, it could not only enhance overall organic and nutrient removal, but also prolong the operative time of the MBBR-MBR due to efficient fouling reduction. Therefore, the use of an MBBR as pretreatment to an MBR could be a promising solution to improve the performance of the MBR system



University of Technology, Sydney

Faculty of Engineering and Information Technology

## Chapter 6

# Evaluation of Micropollutant Removal and Fouling Reduction in a Moving Bed Biofilm Reactor-Membrane Bioreactor Hybrid System

### 6.1 Introduction

In recent years, the frequent detection of micropollutants in the aquatic environment has raised specific concerns due to their detrimental effects on aquatic organisms and human health. It has been reported that micropollutants often exhibit incomplete removal during the activated sludge process. As an alternative to the activated sludge process, the moving bed biofilm reactor (MBBR) technology has demonstrated its suitability for micropollutant removal (Luo et al., 2014).

While MBBR has become an emerging technology for eliminating micropollutants, a major concern for MBBR applications is the decrease of sludge settleability when treating high strength wastewater, which may lead to severe operational problems when clarifiers are employed for the separation of solids. To counter this problem, various hybrid systems have been developed, which involve modifications of the basic MBBR system by adding coagulants (metal salts or cationic polymers) or applying membrane filtration or floatation as the solid separation process (Leiknes et al., 2006). Among all these modifications, combining membrane technology with MBBR is an established concept with growing popularity, which may also result in better membrane performance (Yang et al., 2008; Duan et al., 2013). Several researchers have demonstrated that the MBBR-membrane hybrid systems have the potential to mitigate membrane fouling (Leiknes et al., 2006; Ivanovic and Leiknes, 2008). However, previous studies have also indicated that the MBBR-membrane filtration hybrid system could experience severe membrane fouling when large amounts of submicron colloidal particles were present in the reactor (Sun et al., 2012).

To date, there is a dearth of knowledge regarding the suitability of a MBBR-membrane filtration hybrid system for micropollutant removal. This study aimed to investigate the performance of an MBBR-membrane bioreactor (MBBR-MBR) hybrid system on micropollutant removal and the effectiveness of the MBBR as a pretreatment option for fouling mitigation in MBR. The fouling propensity was investigated based on mixed liquor characteristics, such as soluble microbial products (SMP), extracellular polymeric substances (EPS), zeta potential, and relative hydrophobicity (RH).

### 6.2 Organic and nutrient removal

The MBBR-MBR system effectively removed DOC (94.7 $\pm$ 0.5%) and NH<sub>4</sub>-N (84.9 $\pm$ 4.5%), are consistent with a previous study by Duan et al. (2013). However, unstable TN reduction (45.2 $\pm$ 8.8%) and low PO<sub>4</sub>-P elimination (34.9 $\pm$ 4.0%) were observed during the experimental period. It was noted that organic carbon and nutrients were principally removed by the MBBR, while the subsequent MBR process offered very limited further elimination. Compared to the MBBR-MBR, the CMBR was less efficient for NH<sub>4</sub>-N (56.1 $\pm$ 3.9%) and TN (21.9 $\pm$ 4.6%) removal, but showed similar DOC removal (94.7 $\pm$ 1.6%) and slightly higher PO<sub>4</sub>-P elimination (45.1 $\pm$ 6.8%) due to the biomass growth.



**Figure 6.1** TOC removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ ).



**Figure 6.2** COD removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ ).



**Figure 6.3** NH<sub>4</sub>-N removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ ).



Figure 6.4 TN removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ ).



**Figure 6.5** PO<sub>4</sub>-P removal in the MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; MBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ ).



Figure 6.6 TOC removal in the CMBR system (conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2 \cdot \text{h}$ ; HRT = 6h; SRT =  $\infty$ ).


Figure 6.7 COD removal in the CMBR system (conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2 \cdot \text{h}$ ; HRT = 6h; SRT =  $\infty$ ).



**Figure 6.8** NH<sub>4</sub>-N removal in the CMBR system (conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ ).







**Figure 6.10** PO<sub>4</sub>-P removal in the CMBR system (conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ ).

The MLSS and MLVSS concentrations in the MBBR tank were very low (0.05–0.13 and 0.04–0.11 g/L, respectively), as the suspended solids were continuously washed away or adsorbed on the sponge cubes and no sludge was recycled back to the MBBR. Regarding the MBR unit, the initial MLSS and MLVSS were 0.06 and 0.05 g/L, respectively, and both showed gradual growth during the operation, reaching 0.91 and 0.89 g/L at the end of the study. As for the CMBR, the MLSS and MLVSS increased from 2.27 and 2.05 g/L to 7.38 and 7.08 g/L, respectively, during operation period.



Figure 6.11 Variations of MLSS and MLVSS concentrations in the CMBR.



Figure 6.12 Nematodes (×10) in the CMBR.

### 6.3 Removal of selected micropollutants

During the MBBR-MBR treatment, the compound-specific removal efficiencies (shown in Fig. 6.13) varied significantly, ranging from 25.5 to 99.5%. Although a clear correlation between removal efficiencies and the effective octanol-water partition coefficients (Log D) of the selected micorpollutants could not be obtained, it was found that all hydrophobic compounds (log D>3.2) were effectively removed (>80%). One possible reason is that the attached growth pattern in the MBBR could enhance the retention of the biomass, thus promoting the enrichment of slow growing microorganisms and the formation of a diverse biocoenosis. In general, micropollutant removal by the MBBR-MBR was higher than that by the CMBR. Moreover, the CMBR was less effective for some micropollutants. Particularly, the removals of carbamazepine, ketoprofen, primidone, estriol and bisphenol A were lowered by 16.2, 30.1, 31.9, 34.5, and 39.9 % respectively during the CMBR treatment.

As complex synthetic substances, pharmaceuticals are classified into various groups with highly variable physico-chemical properties. As the selected pharmaceuticals (PPCPs) in this study generally displayed low hydrophobicity (log D < 2.5), biodegradation (rather than sorption) was the major removal pathway of these compounds (Luo et al., 2014a). The removal efficiency of these compounds was much more strongly affected by their intrinsic biodegradability when log D value was below 3.2 (Tadkaew et al., 2011). One possible reason is that these micropollutants have diverse molecular structure and functional groups (Wijekoon et al., 2013). Most of the investigated PPCPs were efficiently removal (> 80%) by the MBBR-MBR, except carbamazepine  $(25.53\pm7.83\%)$ , metroidazole  $(42.39\pm7.86\%)$ , and diclofenac  $(44.19\pm6.21\%)$ . This could be ascribed to the inclusion of strong electron donating (readily biodegradable) functional groups (e.g., - OH) in these compounds. On the other hand, the low removal of carbamazepine could be attributed to its low hydrophobicity and the occurrence of strong electron donating groups such as amide and chloride in its molecular structure (Wijekoon et al., 2013). The formation of biocoenosis could be different in various systems (namely MBBR-MBR and CMBR), resulting in the differences in removal efficiencies of micropollutants.

Pesticides are commonly less biodegradable compounds and some of them are even harmful to microorganisms due to certain toxic effects. As pesticides have been commonly considered of agricultural origin instead of urban origin, a limited amount of investigation has been performed at full scale and most documented WWTPs tended to show inadequate elimination of pesticides (Köck-Schulmeyer et al., 2013). In the both systems, two pesticides displayed different removals. Fenoprop exhibited inefficient elimination ( $25.52\pm16.68\%$  in the MBBR-MBR hybrid system and  $24.4\pm3.3\%$  in the CMBR), whereas pentachlorophenol experienced much higher removal ( $84.5\pm4.99\%$  in the MBBR-MBR hybrid system and  $80.9\pm1.7\%$  in the CMBR). The poor removal of fenoprop could be attributed to its low hydrophobicity (log D = -0.13) and recalcitrance (Hai et al., 2011).

Regarding industrial chemicals, the observed removals were generally high (>70%) in the MBBR-MBR hybrid system, especially for 4-n-nonylphenol (95.68±2.69%), as these compounds are commonly characterized by high hydrophobicity (log D>3.2). The less removals of 4-tert-butylphenol (71.38±8.87%), bisphenol A (89.83±4.03%) and 4tert-octylphenol (87±8.03%) could be due to the lower log D of these compounds (3.40, 3.64 and 5.18, respectively) in comparison with 4-n-nonylphenol (6.14). In the CMBR, the trend of removals was similar, but the efficiencies were considerably lower than those of in the MBBR-MBR.

In the case of estrogenic hormones, the removal was consistently high (>85%), which could be attributed to the high hydrophobicity (log D > 3.2) of these compounds (except estriol) as previously suggested by Tadkaew et al. (2011). According to Andersen et al. (2003), estrogenic hormones could be efficiently biodegraded under nitrifying conditions. Hence, they could be successfully eliminated in the MBBR-MBR with high nitrification efficiency. As the synthetic hormone  $17\alpha$ -ethinylestradiol was more resistant to biodegradation, it exhibited slight lower removal (76.2±3.04%) as compared with  $17\beta$ -estradiol,  $\beta$ -estradiol 17-acetate and estriol (98.17±0.83%, 97.81±1.04% and 97.24±2.91%, respectively).



Figure 6.13 Micropollutant removals in the CMBR and MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT = ∞)

6-12



**Figure 6.14** Relationship of Log D and pharmaceutical removals in the CMBR and MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )



Figure 6.15 Relationship of Log D and pesticides removals in the CMBR and MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5-6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4-7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2$ ·h; HRT = 6h; SRT =  $\infty$ )



**Figure 6.16** Relationship of Log D and industrial chemicals removals in the CMBR and MBBR-MBR. An error bar represents the standard deviation over the experimental period (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT =  $\infty$ )





To gain further insight into the fate of micropollutants during the MBBR-MBR treatment, a mass balance of the investigated compounds was evaluated (Eq. 1), taking into account the removal pathways of biodegradation and sorption in the MBBR unit, and total removal in the MBR unit.

$$L_{inf} = L_{s,MBBR} + L_{b,MBBR} + L_{MBR} + L_{eff}$$
(6.1)

where  $L_{inf}$  is the influent load of micropollutants over the experimental period (ng),  $L_{s,MBBR}$  and  $L_{b,MBBR}$  are the amount of a compound removed via sorption (ng) and biodegradation (ng), respectively, in the MBBR unit;  $L_{MBR}$  is the amount of a compound removed in the MBR (ng);  $L_{eff}$  is the amount of a compound released from the system (ng).

The calculation of the sorption (Eq. 2) and biodegradation (Eq. 3) in MBBR was carried out according to Luo et al. (2014).

$$L_{s,MBBR} = Q \cdot MLSS_{MBBR} \cdot C_{ss} \cdot T + \Delta SS \cdot C_{sa}$$
(6.2)

$$L_{b,MBBR} = (L_{inf} - L_{eff,MBBR}) - L_{s,MBBR}$$
(6.3)

where Q is the flow rate of the MBBR-SMBR system (L/day); MLSS<sub>MBBR</sub> is mixed liquor suspended biosolids concentration in the MBBR (g/L);  $C_{ss}$  is the concentration of a compound on the suspended biosolids (ng/g); T is the duration of the experimental period (days);  $\Delta$ SS is the increased amount of attached biosolids (g);  $C_{sa}$  is the concentration of a compound on the attached biosolids (ng/g);  $L_{eff, MBBR}$  is the amount of a compound released from the MBBR unit (ng).

Regarding the calculation for the removal in the MBR unit, the following equation was used:

$$L_{\rm MBR} = L_{\rm eff, MBBR} - L_{\rm eff} \tag{6.4}$$

Fig. 6.18 illustrates the fate of the selected compounds in the MBBR-MBR system. The results show that biodegradation in the MBBR accounted for the major proportion of the micropollutant removal, which are consistent with findings from previous studies involving both MBR (Wijekoon et al., 2013) and MBBR (Luo et al., 2014). Compared to biodegradation, sorption was a much less significant removal pathway for most micropollutants, except some refractory compounds (fenprop, diclofenac, carbamazepine and pentachlorophenol) and some hydrophobic compounds (4-tert-butylphenol and 4-tert-octylphenol). The impact of MBR removal was minimal in most cases. This was attributable to the low MLSS concentration and the large pore size  $(0.2 \ \mu m)$ ; two orders of magnitude larger than the molecular sizes of micropollutants) of the MF membrane used in this study. Nevertheless, the MBR unit was able to complement the removal of a few compounds including 4-tertoctylphenol (21.1%), metronidazole (14.9%), gemfibrozil (13.0%), and pentachlorophenol (13.0%). Generally, the removal in the MBR can be achieved through charge repulsion, adsorption onto membrane surface, sorption diffusion, solute-solute interactions and fouling layer interactions (Schäfer et al. 2011). In addition, biological degradation in the MBR unit might be helpful in the removal of the compounds that were less likely to interact with particulate matter and membrane surface, such as gemfibrozil and diclofenac.

The influence of hydrophobicity on the behaviour of micropollutants removal in the MBBR-MBR system was also evaluated. Although Rattier et al. (2014) indicated that higher affinity for solids may lead to less efficient micropollutant uptake and biodegradation, the biological transformation of the hydrophobic micropollutants (Log D > 3.2) in this study was significant (80.8% on average), compared to 57.5% for the less hydrophobic compounds. The results are also in good agreement with a study conducted by Tadkeaw et al. (2011) who suggested that hydrophobic micropollutants could adsorb onto the sludge phase and thus prolong their retention time in the reactor for subsequent biodegradation. As sorption values reported in Fig. 6.18 only present the residual micropollutant, 4-n-nonyl phenol exhibited the highest biodegradability (96.8%) as well as the lowest percentage of sorption in MBBR (0.9%).



Figure 6.18 Fate of the studied micropollutants in the hybrid MBBR-MBR system (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT = ∞)

6-17

#### 6.4 Membrane fouling analysis

Fig. 6.19 presents the membrane fouling profile of the MBBR-MBR and the CMBR systems, indicated by TMP. The TMP development in the MBBR-MBR was much less significant (reached up to 35 kPa in 89 days of operation) than that in the CMBR system (reached up to 38.5 kPa within 16 days). Therefore, it is evident that the use of MBBR as the pre-treatment can lower the fouling propensity and improve the filtration performance of MBR.



**Fig. 6.19** Comparison of TMP profiles between the hybrid MBBR-MBR and the CMBR (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate =  $8.83 \text{ L/m}^2 \cdot \text{h}$ ; HRT = 6h; SRT =  $\infty$ )

It has been suggested that severe membrane fouling is often a result of organics accumulation on or in the membrane as bound EPS or SMP (Guo et al., 2012; Meng et al., 2009). Fig. 6.16 compares the EPS in activated sludge and SMP in mixed liquor concentrations at different TMPs in both systems. It is clear that EPS concentrations of both systems remained increasing within the time frame, which could be a result of MLSS build-up during the operation. The EPS concentrations peaked at 16.24 mg/L (MBBR-MBR) and 19.53 mg/L (CMBR) at the end when severe membrane fouling occurred. By contrast, the SMP concentrations showed a

wide disparity between the MBBR-MBR and CMBR, with the former ranging from 4.02 to 6.32 mg/L and the latter between 21.78 and 33.04 mg/L. Since the measured concentrations of EPS were similar in the two systems, the SMP levels could be accounted for the considerably different fouling profiles of the two systems. Duan et al. (2013) also reported that the use of moving bed media could alter the composition of EPS and SMP (e.g., different O–H bonds in hydroxyl functional groups, and less polysaccharides and lipids), as well as produce a suspension low in fouling propensity.



Figure 6.20 Comparison of EPS and SMP values between the hybrid MBBR-MBR and the CMBR (MBBR conditions: aeration rate = 4.5 L/min, DO = 5.5–6.5 mg/L; HRT = 24h; CMBR conditions: aeration rate = 6 L/min, DO = 6.4–7.6 mg/L; filtration rate = 8.83 L/m<sup>2</sup>·h; HRT = 6h; SRT = ∞)

The sludge characteristics (e.g., Zeta potential, hydrophobicity and floc size) also play a role in membrane fouling. The sludge flocs in the MBR unit of the hybrid system showed increased zeta potential (-12.1 to -19.0 mV) and higher RH (75.0 to 83.3%) compared to the sludge in CMBR (Zeta potential: -18.0 to -20.8 mV; RH: 66.0 to 75.5%). According to Ji et al. (2010), increase of zeta potential could neutralise the negative charge on the surface of the floc and form large aggregates, while high RH generally resulted in less interaction between the hydrophobic flocs and the hydrophilic membrane. Sludge particle size distribution analysis confirmed that the floc size of sludge in the MBBR-MBR (15 to 40  $\mu$ m) was larger than that of CMBR sludge (5 to 30  $\mu$ m), which also supported the alleviated membrane fouling in the MBBR-MBR.

Shirazi et al. (2010) reviewed the application of a resistance-in-series model for assessing membrane fouling using Darcy's Law. The model is as follows:

$$J = \Delta P_{\rm T} / (\mu \cdot R_{\rm t}) \tag{6.4}$$

$$\mathbf{R}_{\mathrm{f}} = \mathbf{R}_{\mathrm{m}} + \mathbf{R}_{\mathrm{C}} + \mathbf{R}_{\mathrm{f}} \tag{6.5}$$

Where, J is permeation flux;  $\Delta P_T$  is transmembrane pressure;  $\mu$  is viscosity of the permeate;  $R_t$  is total resistance of membrane filtration;  $R_m$  is intrinsic membrane resistance caused by membrane itself and permanent resistance;  $R_c$  is cake resistance formed by cake layer deposited over membrane surface;  $R_f$  is fouling resistance caused by pore plugging and/or solute adsorption onto the membrane pore and surface.

After the experiment, the fouled membrane was subjected to physical and chemical cleaning, in order to determine different membrane fouling resistances using the equations given above. Table 6.1 displays the calculated membrane fouling resistances. As shown in the Table, the cake layer resistance contributed the most to the total resistance (76.5% in the MBR unit of MBBR-MBR and 68.8% in the CMBR), while irreversible fouling resistance, pore blocking resistance and membrane resistance were much less significant (1.0, 12.0 and 10.5% in the MBR unit of MBBR-MBR and 1.3, 15.3 and 14.6% in the CMBR, respectively). Deng et al. (2014) also found that cake layer formation was one of the main factors contributing

to the membrane fouling of a sponge-submerged membrane bioreactor and a conventional membrane bioreactor when treating synthetic domestic wastewater, whereas pore blocking was of much less significance.

	MBBR-SMBR		SMBR	
Parameter	Resistance $(10^{12}/m)$	Percentage (%)	Resistance $(10^{12}/m)$	Percentage (%)
Clean membrane	1.33	10.5	1.57	14.6
Cake layer	9.63	76.5	7.40	68.8
Pore blocking	1.51	12.0	1.64	15.3
Permanent fouling	0.12	1.0	0.14	1.3
Total resistance	12.59	100	10.75	100

Table 6.1 Membrane resistances for different types of fouling.



Figure 6.21 Comparison of fouled membrane and after physical and chemical cleaning

# 6.5 Conclusion

The MBBR-MBR hybrid system could effectively remove most of the studied micropollutants, while the CMBR was less effective for some compounds, such as ketoprofen, carbamazepine, primidone, bisphenol A and estriol. In the MBBR-MBR, biodegradation served as the primary pathway for micropollutant removal, while the contribution of other removal mechanisms was less significant. Regarding membrane performance, the hybrid system showed considerably lower fouling tendency than the CMBR. SMP was found to be the key contributor to the high fouling propensity in CMBR. The results proved that MBBR could be a prospective pretreatment to MBR for micropollutant removal and membrane fouling minimisation.



# University of Technology, Sydney

Faculty of Engineering and Information Technology

# Chapter 7

# **Conclusions and Recommendations**

### 7.1 Conclusions

The thesis investigated the effectiveness of a sponge-based MBBR for the removal of micropollutants at four HRTs (HRT of 24 h, 18 h, 12 h and 6 h). In addition, a hybrid system combining the MBBR with a submerged membrane bioreactor (MBBR-MBR) was also evaluated with respect to micropollutant removal and membrane fouling reduction strategy. Furthermore, a comparative study between the MBBR-MBR and a conventional MBR was also performed to examine the effectiveness of the use of an MBBR as pretreatment to an MBR.

The removal efficiency of the MBBR was comparable with other processes such as activated sludge or MBR). The MBBR appeared to be an effective process for removal of ibuprofen, metronidazole, naproxen, primidone, triclosan, estrone,  $17-\alpha$  ethinylestradiol, 4-n-nonylphenol, 4-tert-octylphenol and fenoprop. In this set of experiment, the optimal HRT was 18h, thus under this condition, further studies could be done to optimise the MBBR system in order to achieve better removal.

The investigated MBBR-MBR demonstrated excellent performance in removing organics, nutrients as well as micropollutants. During the MBBR-MBR treatment, the compound-specific removal efficiencies varied significantly, ranging from 11.0 to 99.5%. However, most compounds were eliminated to large extents (>70%). One possible reason for the high removal efficiency is that the attached growth pattern could enhance the retention of the biomass, thus promoting the enrichment of slow growing microorganisms and the formation of a diverse biocoenosis. According to the result from this experiment, a longer HRT (e.g. HRT of 24 h) can significantly mitigate membrane fouling when compared with a relatively short HRT (e.g. HRT of 6 h). By using an MBBR as a pretreatment to an MBR, it could not only enhance overall organic and nutrient removal, but also prolong the operative time of the MBBR-MBR due to efficient fouling reduction. Therefore, the use of an MBBR as pretreatment to an MBR could be a promising solution to improve the performance of the MBR system.

The MBBR-MBR hybrid system could effectively remove most of the studied micropollutants, while the CMBR was less effective for some compounds, such as ketoprofen, carbamazepine, primidone, bisphenol A and estriol. In the MBBR-MBR, biodegradation served as the primary pathway for micropollutant removal, while the contribution of other removal mechanisms was less significant. Regarding membrane performance, the hybrid system showed considerably lower fouling tendency than the CMBR. SMP was found to be the key contributor to the high fouling propensity in CMBR. The results proved that MBBR could be a prospective pretreatment to MBR for micropollutant removal and membrane fouling minimisation.

# 7.2 Recommendations for future research

For the MBBR unit, various aspects were found worthy of further research:

- As nitrifying conditions has considerable influence on micropollutant removal in the MBBR unit, the research on evaluating the correlation between the nitrification capacity of the MBBR and the removal efficiency of micropollutants should be considered.
- 2) Under different HRTs, the removal of micropollutants can be quite different. It is possibly due to the diverse microbial community and the existence of different sludge characteristics on and inside the sponge cubes. Further studies are required to examine the influence of microbial community on micropollutant removal.
- 3) As attached growth carrier (sponge in this study) plays a key role in the performance of MBBR, future research is needed to assess the effects of the size of sponge cubes, filling ratio and circulating velocity on micropollutant removal. It is also worthwhile to compare the MBBR efficiency using different attached-growth media, such as polymer carriers.

For the MBR unit, some aspects were found worthy of further research:

- 4) In this study, MF membrane was used in the submerged MBR unit, but its impact on micropollutants removal was minimal. In further study, other types of membrane can be used to compare the effectiveness of removing micropollutants and membrane fouling behaviours.
- The optimisation of the operation conditions of the integrated MBBR-MBR to achieve simultaneous micropollutant removal and fouling control needs further investigation.

For other factors, the following issues were found worthy of further research:

6) Wastewater characteristics can significantly affect the micropollutants removal. As only synthetic wastewater was used for this study, further evaluation using various wastewaters (e.g., municipal wastewater, industrial wastewater and hospital effluents) are necessary to elucidate the effectiveness of the MBBR in removing micropollutants. 7) Some compounds (e.g. carbamazepine, metronidazole, fenoprop) were poorly treated under all the examined conditions in this study. It may be interesting to develop removal strategies coupling MBBR with other treatment process(es) for such compounds.

## References

- Accinilli, C., Sacca, M. L., Mencarelli, M., Vicari, A. 2012. Application of bioplastic moving bed biofilm carriers for the removal of synthetic pollutants from wastewater. Bioresource Technology, 120, 180 – 186.
- Albertson, O. E. 2000. Aerobic Fixed-Growth Reactors. Water Environment Federation, Alexandria.
- Alresheedi, M. T. and Basu, O. D. 2013. Support media impacts on humic acid, cellulose, and kaolin clay in reducing fouling in a submerged hollow fiber membrane system. Journal of Membrane Science, 450, 282-290.
- Andreottola, G., Foladori, P., Ragazzi, M., Tatano, F. 2000. Experimental comparison between MBBR and activated sludge system for the treatment of municipal wastewater. Water Science and Technology, 41, 375 - 382.
- Baran, N., Lepiller, M., & Mouvet, C. 2008. Agricultural diffuse pollution in a chalk aquifer (Trois Fontaines, France): Influence of pesticide properties and hydrodynamic constraints. Journal of Hydrology, 358(1), 56-69.
- Barnes, K. K., Christenson, S. C., Kolpin, D. W., Focazio, M. J., Furlong, E. T., Zaugg, S. D., ... & Barber, L. B. 2004. Pharmaceuticals and other organic waste water contaminants within a leachate plume downgradient of a municipal landfill. Groundwater Monitoring & Remediation, 24(2), 119-126.
- Barwal, A. and Chaudhary, R. 2014. To study the performance of biocarriers in moving bed biofilm reactor (MBBR) technology and kinetics of biofilm for retrofitting the existing aerobic treatment systems: a review. Rev. Environ. Sci. Biotechnol., 13, 285–299.
- Basu, O. D., Huck, P. M. 2005. Impact of support media in an integrated biofiltersubmerged membrane system. Water Research, 39, 4220-4228.
- Benner, J., Helbling, D.E., Kohler, H.-P.E., Wittebol, J., Kaiser, E., Prasse, C., Ternes, T.A., Albers, C.N., Aamand, J., Horemans, B., Springael, D., Walravens, E., Boon, N., 2013. Is biological treatment a viable alternative for micropollutant removal in drinking water treatment processes? Water Research 47 (16), 5955-5976.
- Bitton, G. 2005. Processes based on attached microbial growth. Wastewater microbiology, 3<sup>rd</sup> ed., John Wiley & Sons, Inc., Hoboken, New Jersey, 291-306.
- Buszka, P. M., Yeskis, D. J., Kolpin, D. W., Furlong, E. T., Zaugg, S. D., & Meyer, M. T. 2009. Waste-indicator and pharmaceutical compounds in landfillleachate-affected ground water near Elkhart, Indiana, 2000–2002. Bulletin of environmental contamination and toxicology, 82(6), 653-659.
- Chae, K. J., Kim, S. M., Park, H. D., Yim, S. H., Kim, I. S. 2008. Development of pseudoamphoteric sponge media using polyalkylene oxide-modified

polydimethylsiloxane (PDMS) for rapid start-up of wastewater treatment plant. Chemosphere, 71, 961-968.

- Chai, S., Guo, J, Chai, Y., Cai, J., Gao, L. 2013. Anaerobic treatment of winery wastewater in moving bed biofilm reactors. Desalination and Water Treatment, DOI: 10.1080/19943994. 2013. 792008.
- Chapanova, G., Jank, M., Schlegel, S., Koeser, H. 2007. Effect of temperature and salinity on the wastewater treatment performance of aerobic submerged fixedbed biofilm reactors. Water Science & Technology, 55, 159-164.
- Chen, S., Sun, D., Chung, J. 2007. Treatment of pesticide wastewater by moving-bed biofilm reactor combined with Fenton-coagulation pretreatment. Journal of Hazardous Materials, 144, 577 584.
- Chen, S., Sun, D., Chung, J. S. 2008. Simultaneous removal of COD and ammonium from landfill leachate using an anaerobic-aerobic moving-bed biofilm reactor system. Waste Management, 28, 339-346.
- Chu, L. and Wang, J. 2011. Comparison of polyurethane foam and biodegradable polymer as carriers in moving bed biofilm reactor for treating wastewater with a low C/N ratio. Chemsphere, 83, 63-68.
- Clara, M., Strenn, B. and Kreuzinger, N. 2004. Carbamazepine as a possible anthropogenic marker in the aquatic environment: investigations on the behavior of carbamazepine in wastewater treatment and during groundwater infiltration. Water Res 38:947–954
- Clausen, L., Arildskov, N. P., Larsen, F., Aamand, J., & Albrechtsen, H. J. 2007. Degradation of the herbicide dichlobenil and its metabolite BAM in soils and subsurface sediments. Journal of contaminant hydrology, 89(3), 157-173.
- Copithorn, R. R. 2010. Biofilm Reactor. Water Environment Federation, Alexandria.
- Cortez, S., Teixeira, P., Oliveira, R., Mota, M. 2008. Rotating biological contactors: a review on main factors affecting performance. Reviews in Environmental Science and Biotechnology, 7, 155-172.
- Czajka, C. P., & Londry, K. L. 2006. Anaerobic biotransformation of estrogens. Science of the total environment, 367(2), 932-941.
- Delnavaz, M., Ayati, B., Ganjidoust, H. 2010. Prediction of moving bed biofilm reactor (MBBR) performance for the treatment of aniline using artificial neural networks (ANN). Water Science and Technology, 179, 769-775.
- Deng, L., Guo, W.S., Ngo, H. H., Zhang, J., Liang, S., Xia, S., Zhang, Z., Li, J. 2014. A comparison study on membrane fouling in a sponge-submerged membrane bioreactor and a conventional membrane bioreactor.
- Di Trapani, D., Christensson, M., Torregrossa, ., Viviani, G., Ødegaard, H. 2013. Performance of a hybrid activated sludge/biofilm process for wastewater

treatment in a cold climate region: Influence of operating conditions. Biochemical Engineering Journal, 77, 214-219.

- Di Trapani, D., Mannina, G., Torregrossa, M., Viviani, G. 2008. Hybrid moving bed biofilm reactors: a pilot plant experiment. Water Science and Technology, 57, 1539 1545.
- Dong, Z., Lu, M., Huang, W., Xu, X. 2011. Treatment of oilfiled wastewater in moving bed biofilm reactors using a novel suspended ceramic biocarrier. Journal of Hazardous Materials, 196, 123-130.
- Duan, L., Jiang, W., Song, Y., Xia, S., & Hermanowicz, S. W. 2013. The characteristics of extracellular polymeric substances and soluble microbial products in moving bed biofilm reactor-membrane bioreactor. Bioresource technology, 148, 436-442.
- Dupla, M., Comeau, Y., Parent, S., Villemur, R., Jolicoeur, M. 2006. Design optimization of a self-cleaning moving-bed bioreactor foe seawater denitrification. Water Research, 40, 249-258.
- Dvořák, L., Lederer, T., Jirků, V., Masák, J., & Novák, L. 2014. Removal of aniline, cyanides and diphenylguanidine from industrial wastewater using a full-scale moving bed biofilm reactor. Process Biochemistry, 49(1), 102-109.
- Elmitwalli, T. A., Oahn, K. L. T., Zeeman, G. Lettinga, G. 2002a.Treatment of domestic sewage in a two-step anaerobic filter/anaerobic hybrid system at low temperature. Water Research, 36, 2225-2232.
- Elmitwalli, T. A., Sklyar, V., Zeeman, G. Lettinga, G. 2002b. Low temperature pretreatment of domestic sewage in an anaerobic hybrid or an anaerobic filter reactor. Bioresource Technology, 82, 233-239.
- Falås, P., Baillon-Dhumez, A., Andersen, H. R., Ledin, A., & la Cour Jansen, J. 2012. Suspended biofilm carrier and activated sludge removal of acidic pharmaceuticals. Water research, 46(4), 1167-1175.
- Falås, P., Longrée, P., la Cour jansen, J., Siegrist, H., Hollender, J., Joss, A. 2013. Micropollutants removal by attached and suspended growth in a hybrid biofilm-activated sludge process. Water Research, 47, 4498-4506.
- Falletti, L. and Conte, L. 2007. Upgrading of activated sludge wastewater treatment plants with hybrid moving-bed biofilm reactors. Industrial & engineering chemistry research, 46, 6656 6660.
- Farabegoli, G., Chiavola, A., Rolle, E. 2009. The biological aerated filter (BAF) as alternative treatment for domestic sewage: optimization of plant performance. Journal of Hazard Materials, 171, 1126-1132.
- Feng, Q., Wang, Y., Wang, T., Zheng, H., Chu, L., Zhang, C., Chen, H., Kong, X., Xing, X. 2012. Effects of packing rates of cubic-shaped polyurethane foam

carriers on the microbial community and the removal of organics and nitrogen in moving bed biofilm reactors. Bioresource Technology, 117, 201 - 207.

- Feng, Y. Yu, Y., Qiu, Y., Zhang, J., Gao, L. 2012. The characteristics and application of grain-slag media in a biological aerated filter (BAF). Journal of Industrial and Engineering Chemistry, 18, 1051-1057.
- Fischer, K., Majewsky, M., 2014. Cometabolic degradation of organic wastewater micropollutants by activated sludge and sludge-inherent microorganisms. Applied Microbiology and Biotechnology 98 (15), 6583-6597.
- Gavrilescu, M., Demnerová, K., Aamand, J., Agathos, S., & Fava, F. 2015. Emerging pollutants in the environment: present and future challenges in biomonitoring, ecological risks and bioremediation. New biotechnology, 32(1), 147-156.
- Gavrilescu, M., Demnerová, K., Aamand, J., Agathos, S., & Fava, F. 2015. Emerging pollutants in the environment: present and future challenges in biomonitoring, ecological risks and bioremediation. New biotechnology, 32(1), 147-156.
- Gómez, M. J., Petrović, M., Fernández-Alba, A. R., & Barceló, D. 2006. Determination of pharmaceuticals of various therapeutic classes by solid-phase extraction and liquid chromatography-tandem mass spectrometry analysis in hospital effluent wastewaters. Journal of Chromatography A, 1114(2), 224-233.
- Grady, C. P.L., Jr., and Lim, H. C. 1980. Biological Wastewater Treatment: Theory and Applications. Marcel Dekker, Inc., New York.
- Griffin, P. and Findlay, G. E. 2002. Process and engineering improvements to rotating biological contractor design. Water Science and Technology, 41, 137 144.
- Guo, W. S., Ngo, H. H., Li, J. 2012a. A mini-review on membrane fouling. Bioresource Technology, 122, 27-34.
- Guo, W. S., Ngo, H. H., Vigneswaran, S., Xing, W., Goteti, P. 2008. A novel sponge submerged membrane bioreactor (SSMBR) for wastewater treatment and reuse. Seperation Science and Technology, 43, 273-285.
- Guo, W. S., Ngo, H., Vigneswaran, S. 2012b. Enhancement of membrane processes with attached growth media. In: Zhang, Y. et al. (Eds.), Membrane Technology and Environmental Applications. American Society of Civil Engineers, New York, pp. 603–634.
- Guo, W. S., Ngo. H. H., Dharmawan, F., Palmer, C. G. 2010. Roles of polyurethane foam in aerobic moving and fixed bed bioreactors. Bioresource Technology, 101, 1435-1439.
- Guo, W. S., Vigneswaran, S., Ngo, H. H., Van Nguyen, T. B., Ben Aim, R. 2006. Influence of bioreactor on a long-term operation of a submerged membrane adsorption hybrid system. Desalination, 191, 92-99.

- Guo, W. S., Xing, W., Ngo, H. H., Hu, A. Y. J., Zhang, R. 2009. Enhancement of organics removal by an integrated nonwoven media biofilter-submerged membrane adsorption hybrid system. Journal of Applied Membrane Science & Technology, 9, 1-8.
- Hai, F. I., Li, X., Price, W. E., & Nghiem, L. D. 2011a. Removal of carbamazepine and sulfamethoxazole by MBR under anoxic and aerobic conditions. Bioresource technology, 102(22), 10386-10390.
- Hai, F. I., Tessmer, K., Nguyen, L. N., Kang, J., Price, W. E., & Nghiem, L. D. 2011b. Removal of micropollutants by membrane bioreactor under temperature variation. Journal of membrane science, 383(1), 144-151.
- Hai, F. I., Tadkaew, N., McDonald, J. A., Khan, S. J., & Nghiem, L. D. 2011c. Is halogen content the most important factor in the removal of halogenated trace organics by MBR treatment?. Bioresource technology, 102(10), 6299-6303.
- Harremoes, P., and Henze, M. 2002. Wastewater Treatment: Biological and Chemical Processes. Springer, Heidelberg.
- Hassani, A. H., Borghei, S. M., Samadyar, H., & Ghanbari, B. 2014. Utilization of moving bed biofilm reactor for industrial wastewater treatment containing ethylene glycol: kinetic and performance study. Environmental technology, 35(4), 499-507.
- Hillebrand, O., Nödler, K., Licha, T., Sauter, M., Geyer, T., 2012. Caffeine as an indicator for the quantification of untreated wastewater in karst systems. Water Res. 46, 395–402.
- Horan, H. J. and Lowe, M. (2007). Full-scale trials of recycled glass as tertiary filter medium for wastewater treatment. Water Research, 41, 253-259.
- Hu, J., Ren, H., Xu, K., Geng, J., Ding, L., Yan, X., Li, K. 2012. Effect of carriers on sludge characteristics and mitigation of membrane fouling in attached growth membrane bioreactor. Bioresource Technology, 122, 35-41.
- Itonaga, T., Kimura, K., Watanabe, Y., 2004. Influence of suspension viscosity and colloidal particles on permeability of membrane used in membrane bioreactor (MBR). Water Sci. Technol., 50, 301-309.
- Ivanovic, I. and Leiknes, T. O. 2012. The biofilm membrane bioreactor (BF-MBR) a review. Desalination and Water Treatment, 37, 288-295.
- Ivanovic, I. and Leiknes, T., 2008. Impact of aeration rates on particles colloidal fraction in the biofilm membrane bioreactor (BF-MBR). Desalination, 231, 182-190.
- Ivanovic, I., Leiknes, T., Ødegaard, H., 2006. Influence of loading rates on production and characteristics of retentate from a biofilm membrane bioreactor (BF-MBR). Desalination, 199, 490-492.

- Jiang, J. Q., Zhou, Z., & Sharma, V. K. 2013. Occurrence, transportation, monitoring and treatment of emerging micro-pollutants in waste water—a review from global views. Microchemical Journal, 110, 292-300.
- Johir, M. A. H., Aryal, R., Vigneswaran, S., Kandasamy, J., Grasmick, A. 2011. Influence of supporting media in suspension on membrane fouling reduction in submerged membrane bioreactor (SMBR). Journal of Membrane Science, 374, 121-128.
- Joss, A., Andersen, H., Ternes, T., Richle, P. R., & Siegrist, H. 2004. Removal of estrogens in municipal wastewater treatment under aerobic and anaerobic conditions: consequences for plant optimization. Environmental science & technology, 38(11), 3047-3055.
- Joss, A., Carballa, M., Kreuzinger, N., Siegrist, H., Zabczynski, S., 2006. Wastewater Treatment. In: Ternes TA, Joss A, editors. Human Pharmaceuticals, Hormones and Fragrances: The challenge of micropollutants in urban water management. IWA Publishing, London.
- Kallenborn, R., Fick, J., Lindberg, R., Moe, M., Nielsen, K. M., Tysklind, M., & Vasskog, T. 2008. Pharmaceutical residues in Northern European environments: consequences and perspectives. In Pharmaceuticals in the Environment (pp. 61-74). Springer Berlin Heidelberg.
- Kermani, M., Bina, B., Movahedian, H., Amin, M. M., & Nikaein, M. 2008. Application of moving bed biofilm process for biological organics and nutrients removal from municipal wastewater. American Journal of Environmental Sciences, 4(6), 675.
- Kim, D., Kim, K. Y., Ryu, H. D., Min, K. K., Lee, S. I. 2009. Long term operation of pilot-scale biological nutrient removal process in treating municipal wastewater. Bioresouce Technology, 100, 3180 - 3184.
- Kim, H., Gellner, J. M., Boltz, J. P., Freudenberg, R. G., Gunsch, C. K., Schuler, A. J. 2010. Effects of integrated fixed film activated sludge media on activated sludge settling in biological nutrient removal systems. Water Research, 44, 1553-1561.
- Kim, S. D., Cho, J., Kim, I. S., Vanderford, B. J., & Snyder, S. A. 2007. Occurrence and removal of pharmaceuticals and endocrine disruptors in South Korean surface, drinking, and waste waters. Water research, 41(5), 1013-1021.
- Korkut, E. N., Martin, J. P., ASCE, M., Yaman, C. 2006. Wastewater treatment with biomass attached to porous geotextile baffles. Journal of Environmental Engineering, 132, 284-288.
- Kuroda, K., Murakami, M., Oguma, K., Muramatsu, Y., Takada, H., Takizawa, S., 2012. Assessment of groundwater pollution in Tokyo using PPCPs as sewage markers. Environ. Sci. Technol. 46, 1455–1464.

- Lapworth, D. J., & Gooddy, D. C. 2006. Source and persistence of pesticides in a semi-confined chalk aquifer of southeast England. Environmental Pollution, 144(3), 1031-1044.
- Lee, H. S., Park, S. J., Yoon, T. II. 2002. Wastewater treatment in a hybrid biological reactor using powdered minerals: effects of organic loading rates on COD removal and nitrification. Process Biochemistry, 38, 81-88.
- Leiknes, T., Bolt, H., Engmann, M. Ødegaard, H., 2006. Assessment of membrane reactor design in the performance of a hybrid biofilm membrane bioreactor (BF-MBR). Desalination, 199, 328-330.
- Lekang, O. and Kleppe, H. 2000. Efficiency of nitrification in trickling filters using different filter media. Aquacultral Engineering, 21, 181-199.
- Levstek, K. and Plazl, I. 2009. Influence of carrier type on nitrification in the moving-bed biofilm process. Water Science and Technology, 59, 875-882.
- Li, H., Han, H., Du, M., Wang, W. 2011. Removal of phenols, thiocyanate and ammonium from coal gasification wastewater using moving bed bioreactor. Bioresource Technology,102, 4667-4673.
- Li, X., Hai, F. I., Nghiem, L. D. 2011. Simultaneous activated carbon adsorption within a membrane bioreactor for an enhanced micropollutants removal. Bioresource Technology, 102, 5319 5324.
- Lim, J. W., Lim, P. E., Seng, C. E., & Adnan, R. 2013. Simultaneous 4-chlorophenol and nitrogen removal in moving bed sequencing batch reactors packed with polyurethane foam cubes of various sizes. Bioresource technology, 129, 485-494.
- Lim, J. W., Seng, C. E., Lim, P. E., Ng, S., L., Sujari, A. N. A. 2011. Nitrogen removal in moving bed sequencing batch reactor using polyurethane foam cubes of various sizes as carries materials. Bioresource Technology, 102, 9876-9883.
- Lin, H., Wang, F., Ding, L., Hong, H., Chen, J., Lu, X. 2011. Enhanced performance of a submerged membrane bioreactor with powdered activated carbon addition for municipal secondary effluent treatment. Journal of Hazardous Materials, 192, 1509-1514.
- Lin, H., Zhang, M., Wang, F., Meng, F., Liao, B., Hong, H., Chen, J., Gao, W., 2014. A critical review of extracellular polymeric substances (EPSs) in membrane bioreactors: Characteristics, roles in membrane fouling and control strategies. Journal of Membrane Science, 460, 110-125.
- Liu, Q., Wang, X. C., Liu, Q., Yuan, H., Du, Y. 2010. Performance of a hybrid membrane bioreactor in municipal wastewater treatment. Desalination, 258, 143-147.

- Liu, Y. and Fang, H. H. P. 2003. Influences of extracellular polymeric substances (EPS) on flocculation, settling and dewatering of activated sludge. Critical Reviews in Environmental Science and TechnologyCritical Reviews in Environmental Science and Technology, 33, 237-273.
- Liu, Y., Du, F., Yuan, L., Zeng, H., Kong, S. 2010. Production of lightweight ceramisite from iron ore tailings and its performance investigation in a biological aerated filter (BAF) reactor. Journal of Hazard Materials, 178, 999-1006.
- Loupasaki, E. and Diamadopoulos, E. 2012. Attached growth systems for wastewater treatment in small and rural communities: a review. Journal of Chemical Technology and Biotechnology, 88, 190-204.
- Luo, Y., Guo, W.S., Ngo, H. H., Nghiem, L. C., Hai, F. I., Kang, J. Xia, S., Zhang, Z., Price, W. E. 2014a. Removal and fate of micropollutants in a sponge-based moving bed bioreactor. Bioresource Technology, 159, 311-319.
- Luo, Y., Guo, W.S., Ngo, H. H., Nghiem, L. D., Hai, F. I., Zhang, J., Liang, S., Wang, X. C. 2014b. A review on the occurrence of micropulltants in the aquatic environment and their fate and removal during wastewater treatment. Science of the Total Environment, 473-474, 619-641.
- Luo, Y., Jiang, Q., Ngo, H. H., Nghiem, L. D., Hai, F. I., Price, W. E., ... & Guo, W. 2015. Evaluation of micropollutant removal and fouling reduction in a hybrid moving bed biofilm reactor-membrane bioreactor system. Bioresource technology, 191, 355-359.
- Mannina, G. and Viviani, G. 2009. Hybrid moving bed biofilm reactors: an effective solution for upgrading a large wastewater treatment plant. Water Science and Technology, 60, 1103 1116.
- Mara, D. and Horan, N. 2003. The Handbook of Water and Wastewater Microbiology. Academic Press, UK.
- Margot, J., 2015. Micropollutant removal from municipal wastewater From conventional treatments to advanced biological processes. PhD thesis N° 6505. EPFL, Lausanne.
- Marques, J. J., Souza, R. R., Souza, C. S., Rocha, I. C. C. 2008. Attached biomass growth and substrate utilization rate in a moving bed biofilm reactor. Brazilian Journal of Chemical Engineering, 25, 665-670.
- Martínez Bueno, M. J., Agüera, A., Gómez, M. J., Hernando, M. D., García-Reyes, J. F., & Fernández-Alba, A. R. 2007. Application of liquid chromatography/quadrupole-linear ion trap mass spectrometry and time-of-flight mass spectrometry to the determination of pharmaceuticals and related contaminants in wastewater. Analytical Chemistry, 79(24), 9372-9384.

- McQuarrie, J. P., and Boltz, J. P. 2011. Moving bed biofilm reactor technology: process applications, design, and performance. Water Environment Research, 83(6), 560-575.
- Meng, F., Chae, S., Drews, A., Kraume, M., Shin, H., Yang, F. 2009. Recent advances in membrane bioreactors (MBRs): membrane fouling and membrane material. Water Research, 43, 1489-1512.
- Meng, F., Zhang, H., Yang, F., Zhang, S., Li, Y., Zhang, X., 2006. Identification of activated sludge properties affecting membrane fouling in submerged membrane bioreactors. Separation and Purification Technology, 51, 95-103.
- Miège, C., Choubert, J. M., Ribeiro, L., Eusebe, M., & Coquery, M. 2008. Removal efficiency of pharmaceuticals and personal care products with varying wastewater treatment processes and operating conditions--conception of a database and first results. Water Science & Technology, 57(1).
- Miller, K. J., & Meek, J. 2006. Helena Valley Ground Water: Pharmaceuticals, Personal Care Products, Endocrine Disruptors (PPCPs), and Microbial Indicators of Fecal Contamination. Helena, MT, USA: Montana Department of Environmental Quality.
- Moe, W. M. and Irvine, R. L. 2000. Polyurethane foam medium for biofiltration II: operation and performance. Journal of environmental engineering, 126, 826-832.
- Mompelat, S., Le Bot, B., & Thomas, O. 2009. Occurrence and fate of pharmaceutical products and by-products, from resource to drinking water. Environment international, 35(5), 803-814.
- Moore, R., Quarmby, J., Stephenson, T. 2001. The effects of media size on the performance of biological aerated filters. Water Research, 10, 2514-2522.
- Moore, T. T., Mahajan, R., Vu, D. Q., Koros, W. J. 2006. Hybrid membrane materials comprising organic polymers with rigid dispersed phases. AIChE Journal, 50, 311-321.
- Morgan-Sagastume, J. M. and Noyola, A. (2008). Evaluation of an aerobic submerged filter packed with volcanic scoria. Bioresource Technology, 99, 2528-2536.
- Munz, G., Gori, R., Mori, G., Lubello, C. 2007. Powdered activated carbon and membrane bioreactors (MBR-PAC) for tannery wastewater treatment: long term effect on biological and filtration process performance. Desalination, 207, 349-360.
- Nakada, N., Kiri, K., Shinohara, H., Harada, A., Kuroda, K., Takizawa, S., Takada, H., 2008. Evaluation of pharmaceuticals and personal care products as watersoluble molecular markers of sewage. Environ. Sci. Technol. 42, 6347–6353.

- Ngo, H. H., Guo, W.S., Xing, W. 2008. Evaluation of a novel sponge-submerged membrane bioreactor (SSMBR) for sustainable water reclamation. Bioresource Technology, 99, 2429-2435.
- Nguyen, L.N., Hai, F.I., Kang, J., Price, W.E., Nghiem, L.D., 2012. Removal of trace organic contaminants by a membrane bioreactor-granular activated carbon (MBR-GAC) system. Bioresour. Technol., 113, 169–173.
- Nikolaou, A., Meric, S., & Fatta, D. 2007. Occurrence patterns of pharmaceuticals in water and wastewater environments. Analytical and bioanalytical chemistry, 387(4), 1225-1234.
- Ødegaard, H. 2006. Innovations in wastewater treatment:-the moving bed biofilm process. Water Science and Technology, 53(9), 17-33.
- Ødegaard, H., Rusten, B., Siljudanled, J. 1999. The development of the moving bed biofilm process – from idea to commercial product. Europe Water Manegement, 2, 36-43.
- Ødegaard, H., Rusten, B., Wessman, F. 2004. State of the art in Europe of the moving bed biofilm reactor (MBBR) process. WEFTEC.
- Ødegaard, H., Rusten, B., Westrum, T. 1994. A new moving bed biofilm reactor application and results. Water Science and Technology, 29, 157-165.
- Ødegaard, H.,Gisvold, G., Strickland, J. 2000. The influence of carries size and shape in the moving bed biofilm process. Water Science and Technology, 41, 383-391.
- Patel, A. Zhu, J., Nakhla, G. 2006. Simultaneous carbon, nitrogen and phosphorous removal from municipal wastewater in a circulating fluidized bed bioreactor. Chemosphere, 65, 1103-1112.
- Patwardhan, A. W. 2003. Rotating biological contactors: A review. Industrial & Engineering Chemistry Research, 42, 2035-2051.
- Pedrouzo, M., Reverte, S., Borrull, F., Pocurull, E., & Marce, R. M. 2007. Pharmaceutical determination in surface and wastewaters using high performance liquid chromatography - (electrospray) - mass spectrometry. Journal of separation science, 30(3), 297-303.
- Pérez, S., & Barceló, D. 2007. Fate and occurrence of X-ray contrast media in the environment. Analytical and bioanalytical chemistry, 387(4), 1235-1246.
- Pervissian, A., Parker, W. J., Legge, R. L., 2012. Combined MBBR MF for industrial wastewater treatment. Environmental Progress & Sustainable Energy, 31, 288-295.

- Phattaranawik, J., & Leiknes, T. 2010. Study of Hybrid Vertical Anaerobic Sludge -Aerobic Biofilm Membrane Bioreactor for Wastewater Treatment. Water Environment Research, 82(3), 273-280.
- Plósz, B.G., Langford, K.H., Thomas, K.V., 2012. An activated sludge modeling framework for xenobiotic trace chemicals (ASM-X): Assessment of diclofenac and carbamazepine. Biotechnology and Bioengineering 109 (11), 2757-2769.
- Plósz, B.G., Leknes, H., Thomas, K.V., 2010. Impacts of competitive inhibition, parent compound formation and partitioning behavior on the removal of antibiotics in municipal wastewater treatment. Environmental Science and Technology 44 (2), 734-742.
- Pomiès, M., Choubert, J.M., Wisniewski, C., Coquery, M., 2013. Modelling of micropollutant removal in biological wastewater treatments: A review. Science of the Total Environment 443 (0), 733-748.
- Poynton, H. C., & Vulpe, C. D. 2009. Ecotoxicogenomics: Emerging Technologies for Emerging Contaminants1. JAWRA Journal of the American Water Resources Association, 45(1), 83-96.
- Qiu, L., Zhang, S., Wang, G., Du, M. 2010. Performances and nitrification properties of biological aerated filters with zeolite, ceramic particles and carbonate media. Bioresource Technology, 101, 7245-7251.
- Rahimi, Y., Torabian, A., Mehrdadi, N., Shahmoradi, B. (2011). Simultaneous nitrification-denitrification and phosphorus removal in a fixed bed sequencing batch reactor (FBSBR). Journal of Hazard Materials, 185, 852-857.
- Rasmussen, V. 2011. The Kaldnes moving bed biofilm process an innovative solution to biological wastewater treatment. Kaldnes MiljØteknologi AS, Postboks, N-3103 Norway.
- Rattier, M., Reungoat, J., Keller, J., & Gernjak, W., 2014. Removal of micropollutants during tertiary wastewater treatment by biofiltration: Role of nitrifiers and removal mechanisms. Water Research, 54, 89-99.
- Regmi, P., Thomas, W., Schafran, G., Bott, C., Rutherford, B., Waltrip, D. 2011. Nitrogen removal assessment through nitrification rates and media biofilm accumulation in a IFAS process demonstration study. Water Research, 45, 6699-6708.
- Rodgers, M. and Zhan, X. M. 2003. Moving-medium biofilm reactors. Reviews in Environmental Science and Biotechnology, 2, 213-224.
- Roy, C., Auger, R., Chénier, R. 1998. Use of non woven textile in intermittent filters. Water Science and Technology, 38, 159-166.
- Sagbo, O., Sun, Y., Hao, A., Gu, P. 2008. Effect of PAC addition on MBR process for drinking water treatment. Separation and Purification Technology, 58, 320-327.

- Satywali, Y. and Balakrishnan, M. 2009. Performance enhancement with powdered activated carbon (PAC) addition in a membrane bioreactor (MBR) treating distillery effluent. Journal of Hazardous Materials, 170, 457-465.
- Schlegel, S. and Koeser, H. 2007. Wastewater treatment with submerged fixed bed biofilm reactor systems – design rules, operating experiences and ongoing developments. Water Science & Technology, 55, 83 – 89.
- Siembida, B., Cornel, P., Krause, S., Zimmermann, B. 2010. Effect of mechanical cleaning with granular material on the permeability of submerged membranes in the MBR process. Water Research, 44, 4037-4046.
- Sirianuntapiboon, S. and Yommee, S. 2006. Application of a new type of moving bio-film in aerobic sequencing batch reactor (aerobic-SBR). Journal of Environmental Management, 78, 149-156.
- Stephenson, T., Reid, E., Avery, L. M., & Jefferson, B. 2013. Media surface properties and the development of nitrifying biofilms in mixed cultures for wastewater treatment. Process Safety and Environmental Protection, 91(4), 321-324.
- Stricker, A. E., Berrie, A., Maas, C. L. A., Fernandes, W., Lishman, L. 2009. Comparison of performance and operation of side-by-side integrated fixed-film and conventional activated sludge processes at demonstration scale. Water Environment Research, 81, 219-232.
- Stuart, M., Lapworth, D., Crane, E., & Hart, A. 2012. Review of risk from potential emerging contaminants in UK groundwater. Science of the Total Environment, 416, 1-21.
- Sun, C., Leiknes, T., Fredriksen, R. H., & Riviere, E. 2012. Comparison of membrane filtration performance between biofilm-MBR and activated sludge-MBR. Desalination and Water Treatment, 48(1-3), 285-293.
- Tadkaew, N., Hai, F. I., McDonald, J. A., Khan, S. J., & Nghiem, L. D. 2011. Removal of trace organics by MBR treatment: the role of molecular properties. Water research, 45(8), 2439-2451.
- Tadkaew, N., Sivakumar, M., Khan, S. J., McDonald, J. A., & Nghiem, L. D. 2010. Effect of mixed liquor pH on the removal of trace organic contaminants in a membrane bioreactor. Bioresource technology, 101(5), 1494-1500.
- Tandukar, M., Machdar, I., Uemura, S., Ohashi, A., Harada, H. 2006. Potential of a combination of UASB and DHS reactor as a novel sewage treatment system for developing countries: long-term evaluation. Journal of environmental engineering, 132, 166-172.
- Tandukar, M., Ohashi, A., Harada, H. 2007. Performance comparison of a pilot-scale UASB and DHS system and activated sludge process for the treatment of municipal wastewater. Water Research, 41, 2697-2705.
- Tandukar, M., Uemura, S., Machdar, I., Ohashi, A., Harada, H. 2005. A low-cost municipal sewage treatment system with a combination of UASB and the "fourth-generation" downflow hanging sponge reactor. Water Science & Technology, 52, 323-329.
- Tawfik, A. and Klapwijk, A. 2010. Polyurethane rotating disc system for posttreatment of anaerobically pre-treated sewage. Journal of Environmental Management, 91, 1183-1192.
- Ternes, T. A., & Hirsch, R. 2000. Occurrence and behavior of X-ray contrast media in sewage facilities and the aquatic environment. Environmental Science & Technology, 34(13), 2741-2748.
- Tian, J., Chen, Z., Yang, Y., Liang, H., Nan, J., Wang, Z., Li, G. 2009. Hybrid process of BAC and sMBR for polluted raw water. Bioresource Technology, 100, 6243-6249.
- Tran, N.H., Urase, T., Ngo, H.H., Hu, J., Ong, S.L., 2013. Insight into metabolic and cometabolic activities of autotrophic and heterotrophic microorganisms in the biodegradation of emerging trace organic contaminants. Bioresource Technology 146 (0), 721-731.
- Tran, N. H., Li, J., Hu, J., & Ong, S. L. 2014. Occurrence and suitability of pharmaceuticals and personal care products as molecular markers for raw wastewater contamination in surface water and groundwater. Environmental Science and Pollution Research, 21(6), 4727-4740.
- Tran, N. H., Gin, K. Y. H., & Ngo, H. H. 2015. Fecal pollution source tracking toolbox for identification, evaluation and characterization of fecal contamination in receiving urban surface waters and groundwater. Science of the Total Environment, 538, 38-57.
- Trinh, T., B. van den Akker, H. Coleman, R. Stuetz, P. Le-Clech and S. J. Khan 2011. Fate of pharmaceuticals during wastewater treatment by a membrane bioreactor. 6th IWA specialist conference on membrane technology for wastewater treatment Aachen GWF-Wasser. 152: 98-102.
- Turbak, A. F. 1993. Non-woven: Theory, Process, Performance and Testing. Atlanta: Tappi Press.
- Vieno, N. M., Härkki, H., Tuhkanen, T., & Kronberg, L. 2007. Occurrence of pharmaceuticals in river water and their elimination in a pilot-scale drinking water treatment plant. Environmental Science & Technology, 41(14), 5077-5084.
- Vulliet, E., & Cren-Olivé, C. 2011. Screening of pharmaceuticals and hormones at the regional scale, in surface and groundwaters intended to human consumption. Environmental Pollution, 159(10), 2929-2934.
- Wang, J., Shi, H., Qian, Y. 2000. Wastewater treatment in a hybrid biological reactor (HBR): effect of organic loading rates. Process Biochemistry, 36, 297-303.

- Wang, R. C., Wen, X. H., Qian, Y. 2005. Influence of carrier concentration on the performance and microbial characteristics of a suspended carrier biofilm reactor. Process Biochemistry, 40, 2922-3001.
- Watkinson, A. J., Murby, E. J., Kolpin, D. W., & Costanzo, S. D. 2009. The occurrence of antibiotics in an urban watershed: from wastewater to drinking water. Science of the total environment, 407(8), 2711-2723.
- Wijekoon, K.C., Hai, F.I., Kang, J., Price, W.E., Guo, W., Ngo, H.H., Nghiem, L.D., 2013. The fate of pharmaceuticals, steroid hormones, phytoestrogens, UVfilters and pesticides during MBR treatment. Bioresource Technology. 144, 247–254.
- Xiao, G. Y. and Ganczarczyk, J. 2005. Structural features of biomass in a hybrid MBBR reactor. Environmental Technology, 27, 289-298.
- Xing, W., Ngo, H. H., Kim, S. H., Guo, W.S., Hagare, P. 2008. Adsorption and bioadsorption of granular activated carbon (GAC) for dissolved organic (DOC) removal in wastewater. Bioresource Technology, 99, 8674-8678.
- Yang, S., Yang, F., Fu, Z., & Lei, R. 2009. Comparison between a moving bed membrane bioreactor and a conventional membrane bioreactor on organic carbon and nitrogen removal. Bioresource Technology, 100(8), 2369-2374.
- Ye, J., Mu, Y., Sun, D., Zhong, C., 2013. Treatment of leachate from a domestic waste incineration plant by MBBR-MBR combined process. Industrial Water Treatment, 33, 20-24.
- Ying, G. G., Kookana, R. S., & Ru, Y. J. 2002. Occurrence and fate of hormone steroids in the environment. Environment international, 28(6), 545-551.
- Zhang, S., Wang, Y., He, W., Wu, M., Xing, M., Yang, J., Gao, N., Pan, M. 2014. Impacts of temperature and nitrifying community on nitrification kinetics in a moving-bed biofilm reactor treating polluted raw water. Chemical Engineering Journal, 236, 242-250.
- Zhang, S., Zhang, Q., Darisaw, S., Ehie, O., & Wang, G. 2007. Simultaneous quantification of polycyclic aromatic hydrocarbons (PAHs), polychlorinated biphenyls (PCBs), and pharmaceuticals and personal care products (PPCPs) in Mississippi river water, in New Orleans, Louisiana, USA. Chemosphere, 66(6), 1057-1069.

## **Publications Related to This Research**

## Accepted

Luo Y, Jiang Q, Ngo H H, et al. Evaluation of micropollutant removal and fouling reduction in a hybrid moving bed biofilm reactor–membrane bioreactor system. Bioresource technology, 2015, 191: 355-359.

## In preparation

Jiang Q, Guo W, Ngo H H, Nghiem LD, Hai FI, Kang J, etc. Removal of Micropollutants in an Moving Bed Biofilm Reactor (MBBR) at Different HRTs.